

CARMEN A. TRUTANICH City Attorney

May 12, 2011

VIA MESSENGER Heather Crossner, Esq. Latham & Watkins 355 South Grand Avenue Los Angeles, CA 90071-1560

Re: Development Agreement btw City of Los Angeles and Loyola Marymount

Dear Heather:

Please find enclosed with this letter one (1) fully executed copy of the Development Agreement between the City of Los Angeles and Loyola Marymount University.

Please note that the Agreement needs to be recorded within 10 days of the Effective Date, which is May 12, 2011.

If you should have any questions, I can be reached at (213) 978-8177.

Sincerely, digan LAÚRA M. CADOGAN

Deputy City Attorney

LMC:mrc: (M:\RP-E\Laura Cadogan\Ltr Heather Crossner re Dev. Agmt -LMU.doc) Enclosure

DEVELOPMENT AGREEMENT by and between THE CITY OF LOS ANGELES and

.

LOYOLA MARYMOUNT UNIVERSITY

dated as of <u>May 12,</u> 2011

DEVELOPMENT AGREEMENT

TABLE OF CONTENTS

1. DEFINITIONS		
	1 1	(6 A
	1.1	"Agreement"
	1.2	"Applicable Rules"
	1.3	"Assignment Agreement"
	1.4	"CEQA"
U 18	1.5	"City"5
<i>ti</i> :	1.6	"City Council"
	1.7	"Conditions of Approval"
	1.8	"Development Agreement Act"
	1.9	"Discretionary Action"5
	1.10	"Effective Date"6
	1.11	"Fees"6
	1.12	"FEIR"
	1.13	"General Plan"
n ⁸⁴	1.14	"Impact Fees"
	1.15	"Ministerial Permits and Approvals"
- ·	1.16	"Mitigation Measures"
	1.17	"Parties"
	1.18	"Party"
	1.19	"Planning Commission"
	1.20	"Planning Director"
	1.21	"Processing Fees"
	1.22	"Project"
	1.23	"Project Approvals"
	1.24	"Property"
	1.25	"Property Owner"
	1.25	"Reserved Powers"
	1.20	"Term"
	1.27	"Transforce"
	1.20	"Transferee"
2. RE	CITAI	S OF PREMISES, PURPOSE AND INTENT.
	2.1	State Enabling Statute
	2.2	City Procedures and Actions
	63	2.2.1 City Planning Commission Action
		2.2.2 City Council Action
	2.3	Purpose of this Agreement
	ل	2.3.1 Public Benefits. 9
		2.5.1 · Lubio Donofita

i

		2.3.2 Property Owner Objectives	9
		2.3.3 Mutual Objectives	
	2.4	Applicability of the Agreement	
3. A(GREEN	MENT AND ASSURANCES	
.* 11	3.1	A smoon and A smoon on the Dart of Deservice Owners	10
	2.1	Agreement and Assurance on the Part of Property Owner	10
			10
80			
		3.1.3 Additional Obligations of Property Owner as Consideration for thi Agreement.	
	3.2	Agreement and Assurances on the Part of the City	10
	2.2	3.2.1 Entitlement to Develop	
		3.2.2 Consistency in Applicable Rules	12
		3.2.3 Changes in Applicable Rules.	12
		3.2.4 Subsequent Development Review	12
		3.2.5 Effective Development Standards	······1.3 1.4
		3.2.6 Interim Use	14
		3.2.7 Moratoria or Interim Control Ordinances	14 1 /
		3.2.8 Impact Fees	
		3.2.9 Processing Fees	14 1 <i>1</i>
		3.2.10 Timeframes and Staffing for Processing and Review.	14 1 <i>/</i> i
4. AN	INUAL	REVIEW	15
	4.1	Annual Review	15
	4.2	Pre-Determination Procedure	15
	4.3	Director's Determination	15
	4.4	Appeal By Property Owner	15
	4.5	Period To Cure Non-Compliance	15
	4.6	Failure To Cure Non-Compliance Procedure	15
	4.7	Termination Or Modification Of Agreement.	16
	4.8	Reimbursement Of Costs	-16
5. DE	FAUL	T PROVISIONS	16
	5.1	Default By Property Owner.	
		5.1.1 Default	10
		5.1.2 Notice of Default	10
÷ 11		5.1.3 Failure to Cure Default Procedures	10 17
130		5.1.4 Termination or Modification of Agreement.	17
	5.2	Default By The City.	17 17
		5.2.1 Default	·····17
		5.2.2 Notice of Default.	·····↓/ □ 17
32 10	5.3	No Monetary Damages	1/ 10

ii

6. GE	NERA	L PROVISIONS	18
	6.1	Effective Date	18
	6.2	Term	18
	6.3	Appeals To City Council	
	6.4	Enforced Delay; Extension Of Time Of Performance	
	6.5	Dispute Resolution	
		6.5.1 Dispute Resolution Proceedings	
	17	6.5.2 Arbitration	
		6.5.3 Arbitration Procedures	19
		6.5.4 Extension Of Term.	19
	6.6	Legal Action	19
	6.7	Applicable Law	20
	·6.8	Amendments	
	6.9	Assignment	20
		6.9.1 Conditions for Assignment	
		6.9.2 Liability Upon Assignment	20
	6.10	Covenants	
	6.11	Cooperation And Implementation	.21
		6.11.1 Processing	
		6.11.2 Other Governmental Permits	21
	2	6.11.3 Cooperation In The Event Of Legal Challenge	.22
	6.12	Relationship Of The Parties	
	6.13	Indemnification	
		6.13.1 Indemnification	.22
		6.13.2 Insurance	.22
	6.14	Extension of Time for All Project Approvals	.22
	6.15	Notices	.22
	6.16	Recordation	.23
	6.17	Constructive Notice And Acceptance	
	6.18	Successors And Assignees	.23
	6:19	Severability	.23
	6.20	Time Of The Essence	
<u>*</u> :	6.21	Waiver	
	6.22	No Third Party Beneficiaries	.24
	6.23	Entire Agreement	.24
	6.24	Legal Advice; Neutral Interpretation; Headings, Table Of Contents, and	
		Index	.24
	6.25	Discretion to Encumber	.24
	6.26	Expeditious Processing	.24
	6.27	Entitlement to Written Notice of Default	.24
	6.28	Counterparts	.25

DEVELOPMENT AGREEMENT

This Development Agreement ("Agreement") is executed this 12^{+1} day of May, 2011, by and between the CITY OF LOS ANGELES, a municipal corporation ("City"), and Loyola Marymount University ("LMU" or the "Property Owner"), pursuant to California Government Code Section 65864 et seq., and the implementing procedures of the City, with respect to the following:

RECITALS

WHEREAS, the City and LMU recognize that the further development of LMU's campus, located at 1 LMU Drive, in the City of Los Angeles (the "Property"), will contribute to the economy of the City of Los Angeles, the Southern California region and California generally; and

WHEREAS, LMU wishes to obtain reasonable assurances that the Property may be developed in accordance with the Project Approvals, as defined below, and the terms of this Agreement; and

WHEREAS, LMU wishes to improve the LMU campus in order to maintain the university's status as a premier educational institution and to provide enhanced academic and research opportunities, improved athletic resources, and superior facilities for cultural events; and

WHEREAS, this Agreement is necessary to assure the Property Owner that the Project will not be reduced in density, intensity or use or be subjected to new rules, regulations, ordinances or policies unless otherwise allowed by this Agreement; and

WHEREAS, LMU will implement additional public benefits above and beyond the necessary mitigation for the project, including forming, and providing meeting space for, a neighborhood advisory committee, allowing the public to use certain LMU facilities, providing meeting space for community groups, cooperating in the formation of a parking permit district and funding the cost of certain parking permits, prioritizing the relocation of their recycling and waste management area, constructing future campus buildings according to Silver LEED standards, and allowing the public access to their historic collections and exhibits.

WHEREAS, the implementation of the Project Approvals and related actions will allow further development of the Project consistent with the Project objectives;

AGREEMENT

NOW, THEREFORE, pursuant to the authority contained in the Development Agreement Act, as it applies to the City, and in consideration of the mutual promises and covenants herein contained and other valuable consideration the receipt and adequacy of which the Parties hereby acknowledge, the Parties agree as follows:

1. DEFINITIONS.

For all purposes of this Agreement, except as otherwise expressly provided or unless the context requires:

1.1 "Agreement" means this Development Agreement and all amendments and modifications thereto.

1.2 "Applicable Rules" means the rules, regulations, ordinances and officially adopted policies of the City in full force and effect as of the Effective Date of this Agreement which are generally applicable to all or some properties within the City. The Applicable Rules shall apply to the Project, Project Approvals, and any subsequent discretionary actions which are necessary for implementation of the Project. Notwithstanding the language of this Section or any other language in this Agreement, all specifications, standards and policies regarding the design and construction of public works facilities, if any, shall be those that are in effect at the time the project plans are being processed for approval and/or under construction.

1.3 "Assignment Agreement" means a written agreement between the Property Owner and a Transferee of the Property Owner, consistent with the terms of this Agreement, in which the parties agree to specific obligations of this Agreement being transferred from the Property Owner to the Transferee of the Property Owner.

1.4 "CEQA" means the California Environmental Quality Act (Cal. Public Resources Code Sections 21000 et seq.) and the State CEQA Guidelines (Cal. Code of Regs., Title 14, Sections 15000 et seq.).

1.5 "City" means the City of Los Angeles, a charter city and municipal corporation, including each and every agency, department, board, commission, authority, employee, and/or official acting under the authority of the City, including without limitation the City Council and the Planning Commission.

1.6 "City Council" means the City Council of the City and the legislative body of the City pursuant to Section 65867 of the California Government Code (Development Agreement Act).

1.7 "Conditions of Approval" means the Conditions of Approval for the Project contained in the Loyola Marymount Specific Plan, approved by the City Planning Commission at its hearings held on October 14, 2010, October 28, 2010, and November 4, 2010, and by the City Council on Feb 25, 2011, and attached hereto as Exhibit A, Conditions of Approval.

1.8 "Development Agreement Act" means Article 2.5 of Chapter 4 of Division 1 of Title 7 (Sections 65864 through 65869.5) of the California Government Code.

1.9 "Discretionary Action" means an action which requires the exercise of judgment, deliberation or a decision on the part of the City, including any board, commission or department or any officer or employee thereof, in the process of approving or disapproving a particular activity, as distinguished from an activity which merely requires the City, including any board,

commission or department or any officer or employee thereof, to determine whether there has been compliance with statutes, ordinances or regulations.

1.10 "Effective Date" is the date on which this Agreement is attested by the City Clerk of the City of Los Angeles after execution by the Property Owner and the Mayor of the City of Los Angeles pursuant to Section 6.1.

1.11 "Fees" means Impact Fees, Processing Fees and any other fees or charges imposed or collected by the City.

1.12 "FEIR" means the Final Environmental Impact Report for the Project, State Clearinghouse No. 2008051103, certified by the City in accordance with the requirements of CEQA.

1.13 "General Plan" means the General Plan of the City.

1.14 "Impact Fees" means impact fees, linkage fees, exactions, assessments or fair share charges or other similar impact fees or charges imposed on and in connection with new development by the City pursuant to rules, regulations, ordinances and policies of the City in full force and effect as of the Effective Date of this Agreement. Impact Fees do not include (i) Processing Fees or (ii) other City-wide fees or charges of general applicability, provided that such City-wide fees or charges are not imposed on impacts of new development.

1.15 "Ministerial Permits and Approvals" means the permits, approvals, plans, inspections, certificates, documents, licenses, and all other actions required to be taken by the City in order for Property Owner to implement, develop and construct the Project and the Mitigation Measures, including without limitation, building permits, foundation permits, public works permits, grading permits, stockpile permits, encroachment permits, and other similar permits and approvals which are required by the Los Angeles Municipal Code and project plans and other actions required by the Project Approvals to implement the Project and the Mitigation Measures. Ministerial Permits and Approvals shall not include any Discretionary Actions.

1.16 "Mitigation Measures" means the mitigation measures described in the FEIR and in the Mitigation Monitoring Program for the Project which is attached hereto as Exhibit B, Mitigation Monitoring Program.

1.17 "Parties" means collectively Property Owner and the City.

1.18 ""Party" means any one of Property Owner or the City.

1.19 "Planning Commission" means the City Planning Commission and the planning agency of the City pursuant to Section 65867 of the California Government Code (Development Agreement Act).

1.20 "Planning Director" means the Planning Director for the City.

1.21 "Processing Fees" means all processing fees and charges required by the City including, but not limited to, fees for land use applications, project permits, building

applications, building permits, grading permits, encroachment permits, tract or parcel maps, lot line adjustments, air right lots, street vacations and certificates of occupancy which are necessary to accomplish the intent and purpose of this Agreement. Expressly exempted from Processing Fees are all Impact Fees which may be imposed by the City on development projects pursuant to rules, regulations, ordinances and policies enacted after the Effective Date of this Agreement, except as specifically provided for in this Agreement. The amount of the Processing Fees to be applied in connection with the development of the project shall be the amount which is in effect on a City-wide basis at the time an application for the City action is made. Notwithstanding the language of this Section or any other language in this Agreement, Property Owner shall not be exempt from the payment of fees, if any, imposed on a City-wide basis as part of the City's program for storm water pollution abatement mandated by the Federal Water Pollution Control Act of 1972 and subsequent amendments thereto, unless a waiver of these fees is provided by the City in a subsequent agreement.

1.22 "Project" means the renovation, demolition, and addition of nonresidential facilities, and renovation, demolition and addition of residential facilities, as more fully described in the Exhibit C, Project Description.

1.23 "Project Approvals" means those Discretionary Actions authorizing the Project which have been approved by the City on or before the Effective Date and not rescinded or superseded by City action taken on or before the Effective Date. Project Approvals include, but are not limited to: certification of the EIR, approval of a General Plan Amendment, Zone Change, and Specific Plan, all as adopted by the City and as amended on or before the Effective Date. The Project Approvals are listed in Exhibit D, Project Approvals.

1.24 "Property" means the real property owned by Property Owner as described in Exhibit E.

1.25 "Property Owner" means Loyola Marymount University ("LMU"), or its successors and assignees as described in Section 6.9.

1.26 "Reserved Powers" means the rights and authority excepted from this Agreement's restrictions on the City's police powers and which are instead reserved to the City. The Reserved Powers include the powers to enact regulations or take future Discretionary Actions after the Effective Date of this Agreement that may be in conflict with the Applicable Rules and Project Approvals, but: (1) are necessary to protect the public health and safety, and are generally applicable on a City-wide basis (except in the event of natural disasters as found by the City Council such as floods, earthquakes and similar acts of God); (2) are amendments to Uniform Codes, as adopted by the City of Los Angeles, and/or the Los Angeles Municipal Code, as applicable, regarding the construction, engineering and design standards for private and public improvements to be constructed on the Property; or (3) are necessary to comply with state or federal laws and regulations (whether enacted previous or subsequent to the Effective Date of this Agreement) as provided in Section 3.2.3.

1.27 "Term" means the period of time for which this Agreement shall be effective in accordance with Section 6.2 hereof.

1.28 "Transferee" means individually or collectively, Property Owner's successors in interest, assignees or transferees of all or any portion of the Development Agreement Property.

2. RECITALS OF PREMISES, PURPOSE AND INTENT.

2.1 State Enabling Statute. To strengthen the public planning process, encourage private participation in comprehensive planning and reduce the economic risk of development, the Legislature of the State of California adopted the Development Agreement Act which authorizes any city to enter into binding development agreements establishing certain development rights in real property with persons having legal or equitable interests in such property. Section 65864 of the Development Agreement Act expressly provides as follows:

"The Legislature finds and declares that:

"(a) The lack of certainty in the approval of development projects can result in a waste of resources, escalate the cost of housing and other development to the consumer, and discourage investment in and a commitment to comprehensive planning which would make maximum efficient utilization of resources at the least economic cost to the public.

(b) Assurance to the applicant for a development project that upon approval of the project, the applicant may proceed with the project in accordance with existing policies, rules and regulations, and subject to conditions of approval will strengthen the public planning process, encourage private participation in comprehensive planning, and reduce the economic cost of development."

Notwithstanding the foregoing, to ensure that the City remains responsive and accountable to its residents while pursuing the benefits of development agreements contemplated by the Legislature, the City: (1) accepts restraints on its police powers contained in development agreements only to the extent and for the duration required to achieve the mutual objectives of the parties; and (2) to offset such restraints, seeks public benefits which go beyond those obtained by traditional City controls and conditions imposed on development project applications.

2.2 City Procedures and Actions.

2.2.1 City Planning Commission Action. The City Planning Commission held a duly noticed public hearing on October 14, 2010 and recommended approval of this Agreement.

2.2.2 City Council Action. The City Council on <u>Feb 25</u>, 2011, after conducting a duly-noticed public hearing, adopted Ordinance No. <u>181604</u>, to become effective on the thirty-first day after publication, or on the forty-first day after posting, approving this Agreement, found that its provisions are consistent with the City's General Plan, the Westchester-Playa Del Rey Community Plan, and the Municipal Code, and authorized the execution of this Agreement.

2.3 Purpose of this Agreement.

2.3.1 Public Benefits. This Agreement provides assurances that the public benefits identified below in section 3.1.3 will be achieved and developed in accordance with the Applicable Rules and Project Approvals and with the terms of this Agreement and subject to the City's Reserved Powers. The Project will provide local and regional public benefits to the City, including without limitation those public benefits listed in Section 3.1.3 below.

2.3.2 Property Owner Objectives. In accordance with the legislative findings set forth in the Development Agreement Act, and with full recognition of the City's policy of judicious restraints on its police powers, the Property Owner wishes to obtain reasonable assurances that the Project may be developed in accordance with the Applicable Rules and Project Approvals and with the terms of this Agreement and subject to the City's Reserved Powers. To the extent of Project development, and as provided by Section 3.1.1, Property Owner anticipates making capital expenditures or causing capital expenditures to be made in reliance upon this Agreement. In the absence of this Agreement, Property Owner would have no assurance that it can complete the Project for the uses and to the density and intensity of development set forth in this Agreement and the Project Approvals. This Agreement, therefore, is necessary to assure Property Owner that the Project will not be (1) reduced or otherwise modified in density, intensity or use from what is set forth in the Project Approvals, (2) subjected to new rules, regulations, ordinances or official policies or plans which are not adopted or approved pursuant to the City's Reserved Powers.

2.3.3 Mutual Objectives. Development of the Project in accordance with this Agreement will provide for the orderly development of the Property in accordance with the objectives set forth in the General Plan. Moreover, a development agreement for the Project will eliminate uncertainty in planning for and securing orderly development of the Property, assure installation of necessary improvements, assure attainment of maximum efficient resource utilization within the City at the least economic cost to its citizens and otherwise achieve the goals and purposes for which the Development Agreement Act was enacted. The Parties believe that such orderly development of the Project will provide public benefits, as described in Section 2.3.1, to the City through the imposition of development standards and requirements under the provisions and conditions of this Agreement, including increased tax revenues, creation and retention of jobs. Additionally, although development of the Project in accordance with this Agreement will restrain the City's land use or other relevant police powers, this Agreement provides the City with sufficient reserved powers during the Term hereof to remain responsible and accountable to its residents. In exchange for these and other benefits to City, the Property Owner will receive assurance that the Project may be developed during the Term of this Agreement in accordance with the Applicable Rules, Project Approvals and Reserved Powers, subject to the terms and conditions of this Agreement.

2.4 Applicability of the Agreement. This Agreement does not: (1) grant density or intensity in excess of that otherwise established in the Project Approvals or Applicable Rules; (2) eliminate future Discretionary Actions relating to the Project if applications requiring such Discretionary Action are initiated and submitted by the owner of the Property after the Effective Date of this Agreement; (3) guarantee that Property Owner will receive any profits from the Project; or (4) amend the City's General Plan except as specified in the Project Approvals. This

Agreement has a fixed Term. Furthermore, in any subsequent actions applicable to the Property, the City may apply such new rules, regulations and official policies as are contained in its Reserved Powers.

3. AGREEMENT AND ASSURANCES.

3.1 Agreement and Assurance on the Part of Property Owner. In consideration for the City entering into this Agreement, and as an inducement for the City to obligate itself to carry out the covenants and conditions set forth in this Agreement, and in order to effectuate the premises, purposes and intentions set forth in Section 2 of this Agreement, Property Owner hereby agrees as follows:

3.1.1 Project Development. Property Owner agrees that it will use commercially reasonable efforts, in accordance with its own business judgment and taking into account market conditions and economic considerations, to undertake any development of the Project in accordance with the terms and conditions of this Agreement and the Project Approvals. However, nothing in this Agreement shall be deemed to obligate Property Owner to initiate or complete development of the Project or any portion thereof within any period of time or at all, or deemed to prohibit Property Owner from seeking any necessary land use approvals for any different land use project on the Property.

3.1.2 Timing of Development. The parties acknowledge that Property Owner cannot at this time predict when or at what rate the Property would be developed. Such decisions depend upon numerous factors which are not all within the control of Property Owner, such as market orientation and demand, availability of funds, and interest rates. Because the California Supreme Court held in <u>Pardee Construction Co. v. City of Camarillo</u>, 37 Cal. 3d 465 (1984), that the failure of the parties therein to provide for the timing of development permitted a later adopted initiative restricting the timing of development and controlling the parties' agreement, it is the intent of Property Owner and the City to hereby acknowledge and provide for the right of Property Owner to develop the Project in such order and at such rate and times as Property Owner deems appropriate within the exercise of its sole and subjective business judgment. The City acknowledges that such a right is consistent with the intent, purpose and understanding of the Parties to this Agreement; provided, however, that this Section 3.1.2 does not in any way affect the specific timing or implementation of improvements or other requirements of development to the extent such provisions are set forth in the Project Approvals.

3.1.3 Additional Obligations of Property Owner as Consideration for this Agreement. As additional consideration for this Agreement, Property Owner shall also provide additional public benefits as set forth below:

3.1.3.1. Neighborhood Advisory Committee. Property Owner shall form the LMU Neighborhood Advisory Committee (the "Committee") and assign a member of the Department of Communication and Government Relations as a community liaison. The Committee shall be comprised of a representative from the LMU Department of Public Safety, a representative from the LMU Student Affairs division, the university's community liaison, two LMU Students, four community representatives appointed by the applicable Council Office (which shall take into consideration the community members' proximity to LMU), two

representatives appointed by the Westchester-Playa del Rey Neighborhood Council, one representative of the applicable Council Office, and one representative of the Los Angeles Police Department. The Committee shall meet at least four times per year, beginning the first year after the Effective Date, in regard to matters of public safety and student behavior in the community, and other community concerns. Meetings shall be scheduled in advance and open to the public. Notice for each meeting shall be: 1) advertised in advance in a local newspaper; 2) posted on the Loyola Marymount University website; and 3) distributed by e-mail to people who have requested such distribution as well as all Committee members and recent meeting attendees. The Council District Office that represents Loyola Marymount University and the immediately surrounding neighborhood shall also be notified in advance of all meetings. The community liaison shall also attend special meetings on reasonable notice called by a majority of the Committee. The Committee and LMU shall work together in good faith to reach a reasonable. practical resolution of matters, which addresses community concerns and legal requirements and operating and programmatic requirements of the university. LMU also shall submit the landscape plan for 80th Street, required by Section 7.1.C.1(g) of the Specific Plan, to the Council Office and the Committee for discussion at one of the Committee's public meetings prior to submittal to the Department of City Planning.

3.1.3.2. Public Use of Facilities. Property Owner agrees that the campus shall remain open to the general public during daytime hours, except during certain limited special events, which shall include but are not limited to graduation, certain athletic events, and other events which are expected to draw a significant number of visitors and vehicles which are not registered with LMU to campus. Property Owner agrees to permit the public to use certain university facilities, such as recreational fields, in accordance with Property Owner's established policies regarding priority use and procedures necessary to maintain public safety, insurance and other requirements including parking availability. Additionally, the William H. Hannon Library, including its exhibits and collections, shall be open to the public and free to access, however Property Owner may impose restrictions on nighttime hours of use and borrower fees.

3.1.3.3. Meeting Room(s) for Community Groups. Property Owner agrees to allow nonprofit community benefit organizations to hold meetings or other events at rooms or other facilities on campus waiving normal rental fees and only charging costs incurred for meetings or events outside of Property Owner's normal operations or hours as specified in Exhibit F. The annual report required as part of Section 4.2 of this Agreement shall provide examples of community use of facilities together with a schedule of the non-rental costs associated with use of Property Owner's facilities. Such meetings or other events shall be held in accordance with Property Owner's facilities necessary to maintain public safety, insurance and other requirements including parking availability. Property Owner also agrees to host the Neighborhood Advisory Committee meetings on campus at no cost to the Neighborhood Advisory Committee, as described in Section 3.1.3.1 above.

3.1.3.4. Permit Parking District. Property Owner agrees, in the event of community requests for permit parking, to support the formation of a parking permit district, and to fund the cost of two annual parking permits per household for the area shown in Exhibit G, up

to \$24,000 annually, should a parking district be established by the Department of Transportation through the City's standard procedure for establishing such districts.

3.1.3.5. Relocate Recycling/Waste Management. Property Owner agrees to relocate the campus recycling and waste management area, as detailed in the Loyola Marymount University Specific Plan, as the first project implemented under the Project.

3.1.3.6. Green Building/LEED Certification. Property Owner agrees to develop all Future Buildings to Silver LEED Certification (as defined in 2010), and to include green building elements to reduce consumption of energy and non-renewable natural resources. For purposes of the Section, "Future Buildings" shall mean all new buildings that do not physically exist on the Loyola Marymount University campus on or before the Effective Date of this Agreement, but notwithstanding the above shall not include any buildings for which building plans have been accepted into plan check with the plan check fees paid to the City's Department of Building and Safety on or before the Effective Date of this Agreement.

3.1.3.7. Historic Exhibits and Collections. Property Owner agrees to maintain its current and all future collections and exhibits on the history of the local area and other subjects of general interest to the public, and to make such exhibits or collections open to the public periodically and at reasonable times. A list of the historic collections and exhibits made open the public during the course of the year shall be included in the annual report required as part of Section 4.2 of this Agreement.

3.2 Agreement and Assurances on the Part of the City. In consideration for Property Owner entering into this Agreement, and as an inducement for Property Owner to obligate itself to carry out the covenants and conditions set forth in this Agreement, and in order to effectuate the promises, purposes and intentions set forth in Section 2 of this Agreement, the City hereby agrees as follows:

3.2.1 Entitlement to Develop. Property Owner has the vested right to develop the Project subject to the terms and conditions of this Agreement, the Applicable Rules, Project Approvals and the Reserved Powers. Property Owner's vested rights under this Agreement shall include, without limitation, the right to remodel, renovate, rehabilitate, rebuild or replace the Project or any portion thereof throughout the applicable Term for any reason, including, without limitation, in the event of damage, destruction or obsolescence of the Project or any portion thereof, subject to the Applicable Rules, Project Approvals and Reserved Powers. To the extent that all or any portion of the Project is remodeled, renovated, rehabilitated, rebuilt or replaced, Property Owner may locate that portion of the Project at any other location of the Property, subject to the requirements of the Project Approvals, the Applicable Rules, and the Reserved Powers.

3.2.2 Consistency in Applicable Rules. Based upon all information made available to the City up to or concurrently with the execution of this Agreement, the City finds and certifies that no Applicable Rules prohibit or prevent the full completion and occupancy of the Project in accordance with the uses, intensities, densities, designs and heights, permitted demolition, and other development entitlements incorporated and agreed to herein and in the Project Approvals.

3.2.3 Changes in Applicable Rules.

3.2.3.1. Nonapplication of Changes in Applicable Rules. Any change in, or addition to, the Applicable Rules, including, without limitation, any change in any applicable general or specific plan, zoning or building regulation, adopted or becoming effective after the Effective Date of this Agreement, including, without limitation, any such change by means of ordinance, City Charter amendment, initiative, referendum, resolution, motion, policy, order or moratorium, initiated or instituted for any reason whatsoever and adopted by the City, the Mayor, City Council, Planning Commission or any other Board, Commission, Department or Agency of the City, or any officer or employee thereof, or by the electorate, as the case may be, which would, absent this Agreement, otherwise be applicable to the Project and which would conflict in any way with the Applicable Rules, Project Approvals, or this Agreement, shall not be applied to the Project unless such changes represent an exercise of the City's Reserved Powers, or are otherwise agreed to in this Agreement. Notwithstanding the foregoing, Property Owner may, in its sole discretion, consent to the application to the Project of any change in the Applicable Rules.

3.2.3.2. Changes in Building and Fire Codes. Notwithstanding any provision of this Agreement to the contrary, development of the Project shall be subject to changes which may occur from time to time in the California Building Code and other uniform construction codes. In addition, development of the Project shall be subject to changes occurring from time to time in Chapters V (Public Safety and Protection) and IX (Building Regulations) of the Municipal Code regarding the construction, engineering and design standards for both public and private improvements provided that these changes are (1) necessary to the health and safety of the residents of the City, and (2) are generally applicable on a Citywide basis (except in the event of natural disasters found by the Mayor or City Council, such as floods, earthquakes and similar disasters).

3.2.3.3. Changes Mandated by Federal or State Law. This Agreement shall not preclude the application to the Project of changes in, or additions to, the Applicable Rules, including rules, regulations, ordinances and official policies, to the extent that such changes or additions are mandated to be applied to developments such as this Project by state or federal regulations, pursuant to the Reserved Powers. In the event state or federal laws or regulations prevent or preclude compliance with one or more provisions of this Agreement, such provisions shall be modified or suspended as may be necessary to comply with such state or federal laws or regulations.

3.2.4 Subsequent Development Review. The City shall not require Property Owner to obtain any approvals or permits for the development of the Project in accordance with this Agreement other than those permits or approvals which are required by the Applicable Rules, the Reserved Powers and/or the Project Approvals. However, any subsequent Discretionary Action initiated by Property Owner which substantially changes the entitlements allowed under the Project Approvals, shall be subject to the rules, regulations, ordinances and official policies of the City then in effect. The Parties agree that this Agreement does not modify, alter or change the City's obligations, discretion or powers, pursuant to CEQA and acknowledge that future Discretionary Actions may require additional environmental review pursuant to CEQA, and that this Agreement is not intended to limit such review, or the City's ability to impose additional mitigation measures as a result of such additional environmental review, consistent with the provisions of CEQA. The Parties further agree that any future action by the City with respect to the Project which would otherwise be considered discretionary, shall not be deemed ministerial solely due to the terms of this Agreement.

3.2.5 Effective Development Standards. The City agrees that it is bound to permit the uses, intensities of use and densities on this Property which are permitted by this Agreement and the Project Approvals, insofar as this Agreement and the Project Approvals so provide or as otherwise set forth in the Applicable Rules or the Reserved Powers. The City hereby agrees that it will not unreasonably withhold or unreasonably condition any Discretionary Action which must be issued by the City in order for the Project to proceed, provided that Property Owner reasonably and satisfactorily complies with all City-wide standard procedures for processing applications for Discretionary Action.

3.2.6 Interim Use. The City agrees that Property Owner may use the Property during the Term of this Agreement for any use which is otherwise permitted by the applicable zoning regulations and the General Plan in effect at the time of the interim use or pursuant to any approvals, permits, or other entitlements previously granted and in effect as of the Effective Date.

3.2.7 Moratoria or Interim Control Ordinances. In the event an ordinance, resolution, policy, or other measure is enacted, whether by action of the City, by initiative, or otherwise, which relates directly or indirectly to the Project or to the rate, amount, timing, sequencing, or phasing of the development or construction of the Project on all or any part of the Property or the implementation of the Mitigation Measures adopted in connection with approval of the Project, City agrees that such ordinance, resolution or other measure shall not apply to the Property or this Agreement, unless such changes are adopted pursuant to the Reserved Powers or other applicable provisions of this Agreement.

3.2.8 Impact Fees. Impact Fees imposed by the City with respect to the Project shall be only those Impact Fees in full force and effect as of the Effective Date, the amounts of which are subject to ongoing annual increases which shall be calculated at time of payment. This Agreement shall not limit any impact fees, linkage fees, exaction, assessments or fair share charges or other similar fees or charges imposed by other governmental entities and which the City is required to collect or assess pursuant to applicable law (e.g., school district impact fees pursuant to Government Code Section 65995).

3.2.9 Processing Fees. Property Owner shall pay all Processing Fees for Ministerial Permits and Approvals.

3.2.10 Timeframes and Staffing for Processing and Review. The City acknowledges Property Owner's position that expeditious processing of Ministerial Permits and Approvals and Discretionary Actions, if any, and any other approvals or actions required for the Project is critical to the implementation of the Project. In recognition of the importance of timely processing and review of Ministerial Permits, Approvals, and Discretionary Actions, the City agrees that upon request from Property Owner, procedures and associated fees will be identified which provide for expeditious processing and reviewing of such Ministerial Permits, Approvals and Discretionary Actions.

4. ANNUAL REVIEW.

4.1 Annual Review. During the Term of this Agreement, the City shall review annually Property Owner's good faith compliance with this Agreement. Such periodic review shall be limited in scope to good faith compliance with the provisions of this Agreement as provided in the Development Agreement Act and Property Owner shall have the burden of demonstrating such good faith compliance.

4.2 Pre-Determination Procedure. Property Owner's submission of compliance with this Agreement, in a form which the Director of Planning may reasonably establish, shall be made in writing and transmitted to the Director of Planning not later than ninety (90) days prior to the yearly anniversary of the Effective Date.

4.3 Director's Determination. On or before the yearly anniversary of the Effective Date of the Agreement, the Director of Planning shall make a determination regarding whether or not Property Owner has complied in good faith with the provisions and conditions of this Agreement. Prior to making the first annual determination of compliance, the Director of Planning shall host a public information meeting regarding compliance with this Agreement including compliance with the Project Approvals. Individuals and homeowners associations who request notification shall receive notice of this public information meeting. In subsequent years, the Director shall hold a public information meeting if there is evidence of noncompliance with the Project Approvals. The determination of compliance shall be made in writing with reasonable specificity, and a copy of the determination shall be provided to Property Owner in the manner prescribed in Section 6.15.

4.4 Appeal By Property Owner. In the event the Director of Planning makes a finding and determination of non-compliance, Property Owner shall be entitled to appeal that determination to the Planning Commission. After a public hearing on the appeal, the Planning Commission shall make written findings and determinations, on the basis of substantial evidence, whether or not Property Owner has complied in good faith with the provisions and conditions of this Agreement. In the event of a finding and determination of compliance, there shall be no appeal by any person or entity. Nothing in this Section or this Agreement shall be construed as modifying or abrogating Los Angeles City Charter Section 245 (City Council review of Commission and Board actions).

4.5 Period To Cure Non-Compliance. If, as a result of this Annual Review procedure, it is found and determined by the Planning Director or the Planning Commission, on appeal, that Property Owner has not complied in good faith with the provisions and conditions of this Agreement, the City, after denial of any appeal or, where no appeal is taken, after the expiration of the appeal period described in Section 6.3, shall submit to Property Owner, by registered or certified mail, return receipt requested, a written notice of non-compliance in the manner prescribed in Section 6.15, stating with specificity those obligations of Property Owner which have not been performed. Upon receipt of the notice of non-compliance, Property Owner shall promptly commence to cure the identified items of non-compliance at the earliest reasonable time after receipt of the notice of non-compliance at the cure of such items of non-compliance not later than sixty (60) days after receipt of the notice of non-compliance, provided that Property Owner shall continuously and diligently pursue such remedy at all times until such item of non-compliance is cured.

Failure To Cure Non-Compliance Procedure. If the Director of Planning finds 4.6 and determines that Property Owner, or its successors, transferees, and/or assignees, as the case may be, has not cured or commenced to cure an item of non-compliance pursuant to this Section, and that the City intends to terminate or modify this Agreement or those transferred or assigned rights and obligations, as the case may be, the Director of Planning shall make a report to the Planning Commission. The Director of Planning shall then set a date for a public hearing before the Planning Commission in accordance with the notice and hearing requirements of Government Code Sections 65867 and 65868. If after such public hearing, the Planning Commission finds and determines, on the basis of substantial evidence, that Property Owner, or its successors, transferees, and/or assignees, as the case may be, has not brought the Project into compliance pursuant to this Section, and that the City may terminate or modify this Agreement, or those transferred or assigned rights and obligations, as the case may be, the finding and determination shall be appealable to the City Council in accordance with Section 6.3 hereof. In the event of a finding and determination of compliance, there shall be no appeal by any person or entity. Nothing in this Section or this Agreement shall be construed as modifying or abrogating Los Angeles City Charter Section 245 (City Council's review of Commission and Council actions).

4.7 Termination Or Modification Of Agreement. The City may terminate or modify this Agreement, or those transferred or assigned rights and obligations, as the case may be, after a finding or determination of noncompliance by the City Council or, where no appeal is taken, after the expiration of the appeal periods described in Section 6.3. There shall be no modifications of this Agreement unless the City Council acts pursuant to Government Code Sections 65867.5 and 65868, irrespective of whether an appeal is taken as provided in Section 6.3.

4.8 Reimbursement Of Costs. Property Owner shall reimburse the City for its actual costs, reasonably and necessarily incurred, to accomplish the required annual review.

5. DEFAULT PROVISIONS

5.1 Default By Property Owner.

5.1.1 Default. In the event Property Owner does not perform its obligations under this Agreement in a timely manner, the City shall have all rights and remedies provided by this Agreement which shall include compelling the specific performance of the obligations of Property Owner under this Agreement, or modification or termination of this Agreement, provided that the City has first complied with the procedure in Section 5.1.2 hereof.

5.1.2 Notice of Default. City shall first submit to Property Owner a written notice of default stating with specificity those obligations which have not been performed. Upon receipt of the notice of default, Property Owner shall promptly commence to cure the identified default(s) at the earliest reasonable time after receipt of the notice of default and shall complete the cure of such default(s) not later than sixty (60) days after receipt of the notice of default, or

such longer period as is reasonably necessary to remedy such default(s), provided that Property Owner shall continuously and diligently pursue such remedy at all times until such default(s) is cured. In the case of a dispute as to whether Property Owner has cured the default, the Parties shall submit the matter to dispute resolution pursuant to Section 6.5 of this Agreement.

5.1.3 Failure to Cure Default Procedures. If after the cure period has elapsed, the Director of Planning finds and determines that Property Owner, or its successors, transferees and/or assignees, as the case may be, remains in default and that the City intends to terminate or modify this Agreement, or those transferred or assigned rights and obligations, as the case may be, the Director shall make a report to the Planning Commission and then set a public hearing before the Commission in accordance with the notice and hearing requirements of Government Code Sections 65867 and 65868. If after public hearing, the Planning Commission finds and determines, on the basis of substantial evidence, that Property Owner, or its successors, transferees and/or assigns, as the case may be, has not cured such default pursuant to this Section, and that the City intends to terminate or modify this Agreement, or those transferred or assigned rights and obligations, as the case may be, Property Owner and its successors, transferees and/or assigns, shall be entitled to appeal that finding and determination to the City Council in accordance with Section 6.3. In the event of a finding and determination that all defaults are cured, there shall be no appeal by any person or entity. Nothing in this Section or this Agreement shall be construed as modifying or abrogating Los Angeles City Charter Section 245 (City Council review of Commission and Board actions).

5.1.4 Termination or Modification of Agreement. The City may terminate or modify this Agreement, or those transferred or assigned rights and obligations, as the case may be, after such final determination of the City Council or, where no appeal is taken, after the expiration of the appeal periods described in Section 6.3. There shall be no modifications of this Agreement unless the City Council acts pursuant to Government Code Sections 65867.5 and 65868, irrespective of whether an appeal is taken as provided in Section 6.3.

5.2 Default By The City.

5.2.1 Default. In the event the City does not accept, process, or render a decision on necessary development permits, entitlements, or other land use or building approvals for use as provided in this Agreement upon compliance with the requirements thereof, or as otherwise agreed to by the Parties, or the City otherwise defaults under the provisions of this Agreement, Property Owner shall have all rights and remedies provided herein or by applicable law, which shall include compelling the specific performance of the City's obligations under this Agreement, provided that Property Owner or Transferee, as the case may be, has first complied with the procedures in Section 5.2.2. No part of this Agreement shall be deemed to abrogate or limit any immunities or defenses the City may otherwise have with respect to claims for monetary damages.

5.2.2 Notice of Default. Property Owner shall first submit to the City a written notice of default stating with specificity those obligations which have not been performed. Upon receipt of the notice of default, the City shall promptly commence to cure the identified default(s) at the earliest reasonable time after receipt of the notice of default and shall complete the cure of such default(s) not later than one hundred and twenty (120) days after receipt of the

notice of default, or such longer period as is reasonably necessary to remedy such default(s), provided that the City shall continuously and diligently pursue such remedy at all times until such default(s) is cured. In the case of a dispute as to whether the City has cured the default, the Parties shall submit the matter to dispute resolution pursuant to Section 6.5 of this Agreement.

5.3 No Monetary Damages. It is acknowledged by the Parties that neither the City nor the Property Owner would have entered into this Agreement if it were liable in monetary damages under or with respect to this Agreement or the application thereof. Therefore, the Parties agree that the Parties shall not be liable in monetary damages and the Parties covenant not to sue for or claim any monetary damages for the breach of any provision of this Agreement.

6. GENERAL PROVISIONS.

6.1 Effective Date. This Agreement shall be effective as set forth in Section 1.10 above.

6.2 Term. The Term of this Agreement shall commence on the Effective Date and shall extend for a period of 20 years after the Effective Date, unless said Term is otherwise terminated, modified or extended by circumstances set forth in this Agreement or by mutual consent of the Parties hereto. Following the expiration of this Term, this Agreement shall terminate and be of no further force and effect; provided, however, that this termination shall not affect any right or duty arising from entitlements or approvals, including the Project Approvals on the Property, approved concurrently with, or subsequent to, the Effective Date of this Agreement. The Term of this Agreement shall automatically be extended for the period of time of any actual delay resulting from any enactments pursuant to the Reserved Powers or moratoria. or from legal actions, administrative proceedings such as appeals or delays of ministerial actions, or appeals which enjoin performance under this Agreement or act to stay performance under this Agreement (other than bankruptcy or similar procedures), or for the period of time during which a lawsuit or litigation (including appeals) relating to the Project or the Project Approvals. including this Development Agreement, has been filed and is pending in a court of competent jurisdiction.

6.3 Appeals To City Council. Where an appeal by Property Owner to the City Council from a finding and/or determination of the Planning Commission is created by this Agreement, such appeal shall be taken, if at all, within twenty (20) days after the mailing of such finding and/or determination to Property Owner, or its successors, transferees, and/or assignees, as the case may be. The City Council shall act upon the finding and/or determination of the Planning Commission within eighty (80) days after such mailing, or within such additional period as may be agreed upon by the Property Owner and the City Council. The failure of the City Council to act shall not be deemed a denial or an approval of the appeal, which shall remain pending until final City Council action.

6.4 Enforced Delay; Extension Of Time Of Performance. In addition to specific provisions of this Agreement, whenever a period of time, including a reasonable period of time, is designated within which either Party hereto is required to do or complete any act, matter or thing, the time for the doing or completion thereof shall be extended by a period of time equal to the number of days during which such Party is actually prevented from, or is unreasonably

interfered with, the doing or completion of such act, matter or thing because of causes beyond the reasonable control of the Party to be excused, including: war; insurrection; riots; floods; earthquakes; fires; casualties; acts of God; litigation and administrative proceedings against the Project (not including any administrative proceedings contemplated by this Agreement in the normal course of affairs (such as the Annual Review)); any approval required by the City (not including any period of time normally expected for the processing of such approvals in the ordinary course of affairs); restrictions imposed or mandated by other governmental entities; enactment of conflicting state or federal laws or regulations; judicial decisions; the exercise of the City's Reserved Powers; or similar bases for excused performance which is not within the reasonable control of the party to be excused (financial inability excepted). This Section shall not be applicable to any proceedings with respect to bankruptcy or receivership initiated by or on behalf of Property Owner or, if not dismissed within ninety (90) days, by any third parties against Property Owner. If written notice of such delay is given to either party within thirty (30) days of the commencement of such delay, an extension of time for such cause will be granted in writing for the period of the enforced delay, or longer as may be mutually agreed upon.

6.5 Dispute Resolution.

6.5.1 Dispute Resolution Proceedings. The parties may agree to dispute resolution proceedings to fairly and expeditiously resolve disputes or questions of interpretation under this Agreement. These dispute resolution proceedings may include: (a) procedures developed by the City for expeditious interpretation of questions arising under development agreements; or (b) any other manner of dispute resolution which is mutually agreed upon by the parties.

6.5.2 Arbitration. Any dispute between the parties that is to be resolved by arbitration shall be settled and decided by arbitration conducted by an arbitrator who must be a former judge of the Los Angeles County Superior Court or Appellate Justice of the Second District Court of Appeals or the California Supreme Court. This arbitrator shall be selected by mutual agreement of the parties.

6.5.3 Arbitration Procedures. Upon appointment of the arbitrator, the matter shall be set for arbitration at a time not less than thirty (30) nor more than ninety (90) days from the effective date of the appointment of the arbitrator. The arbitration shall be conducted under the procedures set forth in Code of Civil Procedure Section 638, et seq., or under such other procedures as are agreeable to both parties, except that provisions of the California Code of Civil Procedure pertaining to discovery and the provisions of the California Evidence Code shall be applicable to such proceeding.

6.5.4 Extension Of Term. The Term of this Agreement as set forth in Section 6.2 shall automatically be extended for the period of time in which the parties are engaged in dispute resolution to the degree that such extension of the Term is reasonably required because activities which would have been completed prior to the expiration of the Term are delayed beyond the scheduled expiration of the Term as the result of such dispute resolution.

6.6 Legal Action. Subject to the limitations on remedies imposed by this Agreement, either Party may, in addition to any other rights or remedies, institute legal action in any court of

competent jurisdiction, to cure, correct, or remedy any default, enforce any covenant or agreement herein, enjoin any threatened or attempted violation, or enforce by specific performance the obligations and rights of the Parties hereto. Notwithstanding the above, the City's right to seek specific performance shall be specifically limited to compelling Property Owner to complete, demolish or make safe any particular improvement(s) on public lands which is required as a Mitigation Measure or Condition of Approval. Property Owner shall have no liability (other than the potential termination of this Agreement) if the contemplated development fails to occur.

7. .

6.7 Applicable Law. This Agreement shall be construed and enforced in accordance with the laws of the State of California.

6.8 Amendments. This Agreement may be amended from time to time by mutual consent in writing of the parties to this Agreement in accordance with Government Code Section 65868. Any amendment to this Agreement which relates to the Term, permitted uses, density or intensity of use, height, or size of buildings, provisions for reservation and dedication of land, conditions, restrictions, and requirements relating to subsequent discretionary action or any conditions or covenants relating to the use of the Property, which are not provided for under the Project Approvals or Applicable Rules, shall require notice and public hearing before the parties may execute an amendment thereto.

6.9 Assignment. The Property, as well as the rights and obligations of Property Owner under this Agreement, may be transferred or assigned in whole or in part by Property Owner without the consent of the City, except as set forth in Sections 6.9.1 and 6.9.2 below. Upon such assignment the assignor shall be released from the obligations so assigned.

6.9.1 Conditions for Assignment. No such assignment shall be valid until and unless the following occur:

6.9.1.1. Written Notice of Assignment Required. Property Owner, or any successor transferor, gives prior written notice to the City of its intention to assign or transfer any of its interests, rights or obligations under this Agreement and a complete disclosure of the identity of the assignee or transferee, including copies of the Articles of Incorporation in the case of corporations and the names of individual partners in the case of partnerships. Upon request by Property Owner, City shall provide written acknowledgement of such assignment in the form reasonably requested by Property Owner. Any failure by Property Owner or any successor transferor to provide the notice shall be curable in accordance with the provisions of Section 5.1.

6.9.1.2. Automatic Assumption of Obligations. Unless otherwise stated elsewhere in this Agreement to the contrary, a Transferee of Development Agreement Property expressly and unconditionally assumes all of the rights and obligations of this Agreement transferred or assigned by Property Owner and which are expressly set forth in the applicable Assignment Agreement.

6.9.2 Liability Upon Assignment. Unless otherwise stated elsewhere in this Agreement to the contrary, each Transferee of any portion of the Development Agreement Property shall be solely and only liable for performance of such Transferee's obligations applicable to its portion of the Development Agreement Property under this Agreement as specified in the applicable Assignment Agreement. Upon the assignment or transfer of any portion of the Development Agreement Property, the Transferee shall become solely and only liable for the performance of those assigned or transferred obligations and shall have the rights of a "Property Owner" under this Agreement; which such rights and obligations shall be set forth specifically in the Assignment Agreement, acknowledged by the transferring Property Owner, and the Transferee, as of the date of such transfer, assignment or conveyance of the applicable portion of the Development Agreement Property. The failure of any Transferee to perform the obligations assigned to it may result, at the City's option, in a declaration that this Agreement has been breached with regards to that specific Transferee, and an election to terminate this Agreement as provided for in Section 5.1 hereof, as it relates to that Transferee's holding. This partial termination is severable from the entire Agreement, and shall not affect the remaining entirety of the Agreement.

6.10 Covenants. The provisions of this Agreement shall constitute covenants which shall run with the land comprising the Property for the benefit thereof, and the burdens and benefits hereof shall bind and inure to the benefit of all assignees, transferees, and successors to the Parties hereto.

6.11 Cooperation And Implementation.

6.11.1 Processing. Upon satisfactory completion by Property Owner of all required preliminary actions and payment of appropriate Processing Fees, including the fee for processing this Agreement, the City shall commence and diligently process all required steps necessary for the implementation of this Agreement and development of the Property in accordance with the terms of this Agreement. Property Owner shall, in a timely manner, provide the City with all documents, plans, fees and other information necessary for the City to carry out its processing obligations pursuant to this Agreement.

6.11.2 Other Governmental Permits. Property Owner shall apply in a timely manner for such other permits and approvals as may be required from other governmental or quasi-governmental agencies having jurisdiction over the Project as may be required for the development of, or provision of services to, the Project. The City shall cooperate with Property Owner in its endeavors to obtain such permits and approvals and shall, from time to time at the request of Property Owner, attempt with due diligence and in good faith to enter into binding agreements with any such entity to ensure the availability of such permits and approvals, or services, provided such agreements are reasonable and not detrimental to the City. These agreements may include, but are not limited to, joint powers agreements under the provisions of the Joint Exercise of Powers Act (Government Code Section 6500, et seq.), or the provisions of other laws to create legally binding, enforceable agreements between such parties. To the extent allowed by law. Property Owner shall be a party to any such agreement, or a third party beneficiary thereof, entitled to enforce for its own benefit on behalf of the City, or in its own name, the rights of the City or Property Owner thereunder or the duties and obligations of the parties thereto. Property Owner shall reimburse the City for all costs and expenses incurred in connection with seeking and entering into any such agreement provided that Property Owner has requested such agreement. Property Owner or Transferee, as the case may be, shall defend the City in any challenge by any person or entity to any agreement, and shall reimburse the City for

any costs and expenses incurred by the City in enforcing any agreement. Any fees, assessments, or other amounts payable by the City thereunder shall be borne by Property Owner or Transferee, as the case may be, except where Property Owner or Transferee, as the case may be, has notified the City in writing, prior to the City entering into an agreement, that it does not desire for the City to execute an agreement.

6.11.3 Cooperation In The Event Of Legal Challenge. In the event of any legal action instituted by a third party or other governmental entity or official challenging the validity of any provision of this Agreement, the Parties hereby agree to affirmatively cooperate in defending said action.

6.12 Relationship Of The Parties. It is understood and agreed by the parties hereto that the contractual relationship created between the parties hereunder is that Property Owner is an independent contractor and not an agent of the City. Further, the City and Property Owner hereby renounce the existence of any form of joint venture or partnership between them and agree that nothing herein or in any document executed in connection herewith shall be construed as making the City and Property Owner joint venturers or partners.

6.13 Indemnification.

6.13.1 Indemnification. Property Owner hereby agrees to and shall indemnify, save, hold harmless and defend the City, its agents, officers, or employees in any claim, action or proceeding brought by a third party (i) challenging the validity of this Agreement, the related Specific Plan, General Plan Amendment, or Zone Change (Case No. CPC 2009-2425-GPA-ZC-SP-DA) or the related CEQA review (ENV 2008-1342-EIR) or (ii) seeking damages which may arise directly or indirectly from the negotiation, formation, execution, enforcement or termination of this Agreement or enactment of the related Specific Plan, General Plan Amendment or Zone change or certification of the related CEQA review. Nothing in this Section shall be construed to mean that Property Owner shall hold the City harmless and/or defend it from any claims arising from, or alleged to arise from, intentional misconduct or gross negligence in the performance of this Agreement. The City shall promptly notify Property Owner of any claim, action or proceeding and the City shall cooperate fully in the defense. If the City fails to promptly notify the applicant of any claim, action or proceeding, or if the City fails to cooperate fully in the defense, LMU shall not thereafter be responsible to defend, indemnify or hold harmless the City. The City may make all reasonable decisions with respect to its representation in any legal proceeding. This indemnification provision survives even if this Development Agreement is vacated or otherwise terminated.

6.13.2 Insurance. Without limiting its obligation to hold the City harmless, Property Owner shall provide and maintain at its own expense, during the Term of this Agreement, a mutually agreed-upon program of insurance concerning its operations hereunder.

6.14 Extension of Time for All Project Approvals. Project Approvals. The duration of all Project Approvals shall automatically be extended for the Term of this Agreement.

6.15 Notices. Any notice or communication required hereunder between the City or Property Owner must be in writing, and shall be given either personally or by registered or certified mail, return receipt requested. If given by registered or certified mail, the same shall be deemed to have been given and received on the first to occur of (i) actual receipt by any of the addressees designated below as the party to whom notices are to be sent, or (ii) five (5) days after a registered or certified letter containing such notice, properly addressed, with postage prepaid, is deposited in the United States mail. If personally delivered, a notice shall be deemed to have been given when delivered to the party to whom it is addressed. Any party hereto may at any time, by giving ten (10) days' written notice to the other party hereto, designate any other address in substitution of the address, or any additional address, to which such notice or communication shall be given. Such notices or communications shall be given to the parties at their addresses set forth below:

593 C

with copies to: Los Angeles City Attorney's Office <u>Real Property/Environment Division</u> Los Angeles City Attorney's Office 7th Floor, City Hall East 200 North Main Street Los Angeles, CA 90012

If to Property Owner:

Loyola Marymount University
Evelynne Scarboro
1 LMU Drive
Los Angeles, CA 90045

Latham & Watkins LLP	
Attention: George Mihlsten and Cindy	
Starrett	
with copies to Records Department	
355 South Grand Avenue	
Los Angeles, CA 90071	3

with copies to

6.16 Recordation. As provided in Government Code Section 65868.5, this Agreement shall be recorded with the Registrar-Recorder of the County of Los Angeles within ten (10) days following the effective date. Property Owner shall provide the City Clerk with the fees for such recording prior to or at the time of such recording should City Clerk record the Agreement.

6.17 Constructive Notice And Acceptance. Every person who now or hereafter owns or acquires any right, title, interest in or to any portion of the Property, is and shall be conclusively deemed to have consented and agreed to every provision contained herein, whether or not any reference to this Agreement is contained in the instrument by which such person acquired an interest in the Property.

6.18 Successors And Assignees. The provisions of this Agreement shall be binding upon and shall inure to the benefit of the Parties, any subsequent owner of all or any portion of the Property and their respective successors and assignees.

6.19 Severability. If any provisions, conditions, or covenants of this Agreement, or the application thereof to any circumstances of either Party, shall be held invalid or unenforceable, the remainder of this Agreement or the application of such provision, condition, or covenant to

persons or circumstances other than those as to whom or which it is held invalid or unenforceable shall not be affected thereby and shall be valid and enforceable to the fullest extent permitted by law.

6.20 Time Of The Essence. Time is of the essence for each provision of this Agreement of which time is an element.

6.21 Waiver. No waiver of any provision of this Agreement shall be effective unless in writing and signed by a duly authorized representative of the Party against whom enforcement of a waiver is sought and refers expressly to this Section. No waiver of any right or remedy with respect to any occurrence or event shall be deemed a waiver of any right or remedy with respect to any other occurrence or event.

6.22 No Third Party Beneficiaries. The only Parties to this Agreement are the City and Property Owner and their successors-in-interest. There are no third party beneficiaries and this Agreement is not intended, and shall not be construed to benefit or be enforceable by any other person whatsoever.

6.23 Entire Agreement. This Agreement sets forth and contains the entire understanding and agreement of the Parties and there are no oral or written representations, understandings or ancillary covenants, undertakings or agreements which are not contained or expressly referred to herein (or any such representations, understandings or ancillary covenants, undertakings or agreements are integrated in this Agreement) and no testimony or evidence of any such representations, understandings, or covenants shall be admissible in any proceedings of any kind or nature to interpret or determine the provisions or conditions of this Agreement.

6.24 Legal Advice; Neutral Interpretation; Headings, Table Of Contents, and Index. Each Party acknowledges that it has received independent legal advice from its attorneys with respect to the advisability of executing this Agreement and the meaning of the provisions hereof. The provisions of this Agreement shall be construed as to their fair meaning, and not for or against any Party based upon any attribution to such Party as the source of the language in question. The headings, table of contents, and index used in this Agreement are for the convenience of reference only and shall not be used in construing this Agreement.

6.25 Discretion to Encumber. This Agreement shall not prevent or limit Property Owner in any manner, at its sole discretion, from encumbering the Property or any portion of the Property or any improvement on the Property by any mortgage, deed of trust or other security device securing financing with respect to the Property or its improvements.

6.26 Expeditious Processing. In the event of legal action, Property Owner and the City agree that upon request from Property Owner, procedures and associated fees will be identified which permit the City to cooperate in the expeditious processing of any legal action seeking specific performance, declaratory relief or injunctive relief, to set court dates at the earliest practicable date(s) and to avoid delay in the prosecution/defense of the action, provided such cooperation shall not require any Party to waive any rights.

6.27 Entitlement to Written Notice of Default. The mortgagee of a mortgage or beneficiary of a deed of trust encumbering the Property, or any part thereof, and their successors

and assigns shall, upon written request to the City, be entitled to receive from the City written notification of any default by Property Owner of the performance of Property Owner's obligations under this Agreement which has not been cured within sixty (60) days following the date of default. Property Owner shall reimburse the City for its actual costs, reasonably and necessarily incurred, to prepare this notice of default.

6.28 Counterparts. This Agreement is executed in duplicate originals, each of which is deemed to be an original. This Agreement, not counting the Cover Page, Table of Contents or Index, consists of 23 pages and _____() Exhibits.

IN WITNESS WHEREOF, the Parties hereto have executed this Agreement as of the date first written above.

CITY OF LOS ANGELES, a municipal corporation of the State of California

By:

Antonio Villaraigosa, Mayor

DATE:

MAY 12 2011

APPROVED AS TO FORM: CARMEN TRUTANICH, City Attomey

By:

Laura M. Cadogan, Deputy City Attorney

DATE: 5-4-11



ATTEST:

City Clerk By

Deputy

05-12-2011 DATE: -118827

LOYOLA MARYMOUNT INIVERSITY By:

Name: David W. Burcham Title: President APPROVED AS TO FORM:

Heather A. Crossner of Latham & Watkins LLP By:

By:

Name: Title:

CALIFORNIA ALL-PURPOSE ACKNOWLEDGMENT

State of California	1
County of Los Angeles	}
On $b5[12/201]$ before me,	Sharon Dickinson, Notary Public Here Insert Name and Title of the Officer
personally appeared An-to	Sharon Dickinson, Notary Public Here Insert Name and Title of the Officer enio R. Villaraigosa Name(s) of Signer(s)
	who proved to me on the basis of satisfactor evidence to be the person(s) whose name(s) is/ar subscribed to the within instrument and acknowledge to me that he/spe/tbey executed the same i his/her/tbeir authorized capacity(ies), and that b his/her/tbeir signature(s) on the instrument th person(s), or the entity upon behalf of which th person(s) acted, executed the instrument.
SHARON M. DICKINSON Commission # 1853606 Notary Public - California Los Angeles County My Comm. Expires Jun 11, 2013	I certify under PENALTY OF PERJURY under the laws of the State of California that the foregoing paragraph is true and correct.
	WITNESS my hand and official seal. Signature: <u>Aharon Dickinson</u>
Place Notary Seal and/or Stamp Above	OPTIONAL Signature of Notary Public
Though the information below is not require and could prevent fraudulent ren	ed by law, it may prove valuable to persons relying on the document noval and reattachment of this form to another document.
Description of Attached Document	4.2.2
	t Agreement (City of LA & Loyola Marymount Univ.
Document Date: 05/12/2011	Number of Pages:25
Signer(s) Other Than Named Above:	
Capacity(ies) Claimed by Signer(s)	
Signer's Name:	Signer's Name:
Corporate Officer — Title(s): <u>Mayor</u>	
OF S	IUMBPRINT Individual RIGHT THUMBPRINT IGNER OF SIGNER numb here Partner — I Limited General
Attorney in Fact	Top of thumb here
□ Trustee	
Guardian or Conservator	□ Guardian or Conservator
Other:	□ Other:
Signer Is Representing:	Signer Is Representing:

CALIFORNIA ALL-PURPOSE ACKNOWLEDGMENT

State of California)
County of Los Angeles	
On <u>4 27 11</u> before me,	Sarah D. Hughes
personally appeared David	Here Insert Name and Title of the Officer
	Name(s) of Signer(s)
* 	who proved to me on the basis of satisfactor
	evidence to be the person(s) whose name(s) is/ar subscribed to the within instrument and acknowledge
	to me that he/she/they executed the same i
	his/her/their authorized capacity(ies); and that b
	his/her/their signature(s) on the instrument th
	person(s), or the entity upon behalf of which th person(s) acted, executed the instrument.
	I certify under PENALTY OF PERJURY under th
	laws of the State of California that the foregoin
SARAH D. HUGHES Commission # 1776544	paragraph is true and correct.
Los Angeles County MyComm.Extres/Oct27.2011	WITNESS my hand and official seal.
A contract of the contract of	Q A r I I
	Signature: Salah), Hughe
Place Notary Seal and/or Stamp Above	Signature of Notary Public
Though the information below is not required by lav	v, it may prove valuable to persons relying on the document d reattachment of this form to another document.
Description of Attached Document	
Title or Type of Document:	ent LMU/City of CI
Document Date:	Number of Pages:
Signer(s) Other Than Named Above:	A
Capacity(ies) Claimed by Signer(s)	
Signer's Name: David W. Burcha	M Signer's Name:
Corporate Officer — Title(s):	Corporate Officer — Title(s):
Individual RIGHT THUMBPRIN OF SIGNER	OF SIGNER
Partner — Limited General Top of thumb here Atterney in Fact	
 Attorney in Fact Trustee 	Attorney in Fact Tructop
Guardian or Conservator	□ Trustee □ Guardian or Conservator
Other:	
Signer Is Representing:	Signer Is Representing:
National Notary Association • 9350 De Soto Ave., P.O. Box 2402 • Chatsworth, CA	91313-2402 • www.NationalNotary.org Item #5907 Reorder: Call Toll-Free 1-800-

Exhibit A:

Conditions of Approval

Specific Plan attached.

EGEI \mathbb{N} Ľ 1) FEB 2 5 2011 By

LOYOLA MARYMOUNT UNIVERSITY SPECIFIC PLAN City of Los Angeles

Ordinance No.

Adopted _/_/_

TABLE OF CONTENTS

SECTION 1:	AUTHORITY AND SCOPE	3
SECTION 2:	LOYOLA MARYMOUNT SPECIFIC PLAN OVERVIEW	6
SECTION 3:	LAND USE REGULATIONS	. 12
SECTION 4:	CAMPUS DESIGN REGULATIONS	. 21
SECTION 5:	TRANSPORTATION AND PARKING REGULATIONS	.27
SECTION 6:	SECURITY REGULATIONS	. 32
SECTION 7:	SIGNAGE	. 33
SECTION 8:	IMPLEMENTATION	. 41
SECTION 9:	OTHER SPECIFIC PLAN PROCEDURES	42
SECTION 10:	INTERPRETATION	42
SECTION 11:	SEVERABILITY	. 43

LIST OF EXHIBITS

LMU Specific Plan Area Boundary	44
Planning Areas	45
Height Areas	46
Setbacks	47
Conditions For On-Site Alcohol Consumption Permits	48
Sign Definitions	51
– 6C. Examples of Banner Signs	53
Existing Ambient Noise Levels	56
McConnell Buffer Areas	57
Specific Plan	58
	LMU Specific Plan Area Boundary Planning Areas Height Areas Setbacks Conditions For On-Site Alcohol Consumption Permits Sign Definitions - 6C. Examples of Banner Signs Existing Ambient Noise Levels McConnell Buffer Areas Specific Plan

APPENDIX

APPENDIX A. Mitigation Monitoring and Reporting Program

SECTION 1: AUTHORITY AND SCOPE

Section 1.1 ESTABLISHMENT OF THE LOYOLA MARYMOUNT UNIVERSITY SPECIFIC PLAN

- A. A specific plan is a regulatory land use ordinance, which controls or provides a framework for the systematic implementation of the General Plan of the City of Los Angeles (City). The Loyola Marymount University Specific Plan¹ is hereby established and is applicable to that Specific Plan Area of the City shown within the heavy dashed lines on the Loyola Marymount University Specific Plan Area Boundary figure as shown on Exhibit 1 for the area bounded generally by Lincoln Boulevard to the west, the Westchester bluffs and Playa Vista to the north, McConnell Avenue to the east, and 80th Street, Fordham Road, and 78th Street to the south.
- B. In order to regulate the use of property as provided for in this Specific Plan, the Specific Plan is divided into six Planning Areas designated as follows: Academic/Residential (A/R); Athletic (A); Campus Core/Open Space (CC/OS); Buffer/Open Space 1 (B/OS1); Buffer/Open Space 2 (B/OS2); and Buffer/Open Space 3 (B/OS3). The location and boundaries of these Planning Areas are shown on the Loyola Marymount University Specific Plan Planning Areas figure (Exhibit No. 2).

Section 1.2 RELATIONSHIP TO THE LOS ANGELES GENERAL PLAN

The Los Angeles General Plan is a comprehensive long-range policy document that guides the physical development of the City of Los Angeles. The General Plan includes certain state-mandated elements related to land use, circulation, housing, conservation, open space, noise, and safety. Whereas the General Plan is a broad policy document, this Specific Plan is a policy statement and implementation tool to cover a defined planning area. This Specific Plan must be consistent with the General Plan by furthering the objectives and policies of the General Plan, and not obstruct their attainment, pursuant to California Government Code Section 65454. This Specific Plan is intended to detail and foster the General Plan's development policies, systematically implement the General Plan, and contain defined standards and development criteria that supplement those of the General Plan.

í.

¹ Capitalized terms are defined in Section 2.3 or Exhibit 6 of this Specific Plan.

Section 1.3 RELATIONSHIP TO THE COMMUNITY PLAN

A Community Plan is a focused planning policy document that designates the distribution of types, amounts, and location of land uses for a particular area of the City. The Westchester – Playa del Rey Community Plan is one of 35 Community Plans that comprise the Land Use element of the Los Angeles General Plan. The Westchester – Playa del Rey Community Plan is applicable to the Loyola Marymount University Specific Plan Area. The Westchester – Play del Rey Community Plan is consistent with the objectives and policies of the General Plan; the Specific Plan is a more focused regulatory document that is consistent with both Plans and reflects the unique constraints and opportunities of the Specific Plan Area. The Specific Plan creates a regulatory framework that accounts for the special needs of the Specific Plan Area and the surrounding community, and allows flexibility for adapting to future changes that could occur in higher education.

Section 1.4 RELATIONSHIP TO CEQA

Pursuant to the California Environmental Quality Act (CEQA), the LMU Master Plan Project Environmental Impact Report (City Case No. ENV-2008-1342-EIR) was prepared for the LMU Master Plan. The LMU Master Plan Project Environmental Impact Report (State Clearing House No. 2008051103) identifies potential effects on the environment of the LMU Master Plan as implemented by this Specific Plan and sets forth mitigation measures to lessen those impacts. All Projects approved pursuant to this Specific Plan shall comply with all applicable Project Design Features and Mitigation Measures as set forth in the Mitigation Monitoring and Reporting Program in Appendix A of this plan.

Section 1.5 RELATIONSHIP TO THE LOS ANGELES MUNICIPAL AND ADMINISTRATIVE CODES

- A. The regulations of this Specific Plan are in addition to those set forth in the Los Angeles Municipal Code (Municipal Code), including the planning, zoning and subdivision provisions of the Municipal Code, the Los Angeles Administrative Code (Administrative Code), and any other relevant ordinances; and do not convey any rights not otherwise granted under the provisions and procedures contained in that chapter and other relevant ordinances, except as specifically provided herein.
- B. Wherever provisions of this Specific Plan differ either in being more restrictive or less restrictive from provisions contained in the Municipal Code or the Administrative Code as of the date of adoption of this Specific Plan, or other relevant ordinances, this Specific Plan shall prevail and supersede those other provisions.

Loyola Marymount University Specific Plan

Whenever this Specific Plan is silent, the provisions of the Municipal Code, Administrative Code or other relevant ordinance shall apply.

- C. The regulations contained within this Specific Plan regarding Floor Area, maximum densities, and yards shall supersede any regulations within Chapter 1 of the Municipal Code.
- D. Without limiting the above paragraphs, the regulations and procedures in this Specific Plan shall supersede and serve as a substitute ordinance for the following provisions of the Municipal Code:
 - 1. Site Plan Review. Section 16.05.
 - 2. Major Development Project. Section 12.24.U.14.
 - 3. Transitional Height. Sections 12.21.1.A.10 and 12.24.X.22.
 - 4. Floor Area Limitations. Section 12.21.1.A.1.
 - 5. Residential Open Space Requirements. Section 12.21.G.
 - 6. Specific Plan Project Permit Compliance and Modifications. Sections 11.5.7.C and 11.5.7.D.
 - 7. Parking Requirements. Section 12.21.A.4 through and including Section 12.21.A.6.
 - 8. Reduced or Shared Parking. Sections 12.24.X.19 and 12.24.X.20.
 - 9. Transportation Demand Management. Section 12.26.J.
 - 10. Alcoholic Beverage Sales and Service Regulations. Sections 12.21.A.10 and 12.24.W.1.
 - 11. Fences and Walls in the A and R Zones. Sections 12.22.C.20(f)(2), 12.22.C.20(f)(3), and 12.22.C.20(f)(7).

Section 1.6 RELATIONSHIP TO EXISTING USES

The LMU Specific Plan creates a regulatory framework for long-term development of the Specific Plan Area consistent with the LMU Master Plan. The Specific Plan also recognizes, however, that existing uses within the Specific Plan Area will continue as the development permitted by this Specific Plan is implemented. Existing uses shall be permitted to

Loyola Marymount University Specific Plan
continue within the Specific Plan Area. This section shall not modify the requirement of Section 3.3.F. with regard to the relocation of the recycling and waste management area.

SECTION 2: LOYOLA MARYMOUNT SPECIFIC PLAN OVERVIEW

Section 2.1 PURPOSES AND GOALS

The purpose and goals of this Specific Plan are to:

- A. Provide regulatory controls and incentives for the systematic and incremental execution of that portion of the General Plan which relates to this geographic area and to provide for public needs, convenience, and general welfare as the development of such area necessitates;
- B. Assure orderly development and appropriate capacity of public facilities for the intensity and design of development by establishing general procedures for development within the Specific Plan Area;
- C. Provide continued development of the site as an educational institution, in conformance with the goals and objectives of local and regional plans and policies;
- D. Permit the removal, renovation, replacement, and construction of academic, administrative, athletic, residential and other support facilities to meet the evolving needs of students, faculty, staff, alumni, and visitors;
- E. Contribute to alleviation of the regional housing shortage through the creation of new student on-Campus housing, and permit an increase in the percentage of undergraduates living on-Campus;
- F. Establish development standards that minimize the visual impact of building masses and facilitate the creation of a visually attractive built environment, through the effective use of open space, landscaping, setbacks, and Height limitations;
- G. Continue to improve land use compatibility on Campus and between the Campus and neighboring properties;
- H. Honor Loyola Marymount University's pledge as a signatory of the American College and University President's Climate Commitment and design a sustainable Campus environment with the goal of reducing GHG emissions, supporting recycling and the reduction of solid waste, and increasing energy and water efficiency;

- I. Provide sufficient on Campus parking to serve all the demonstrated need and demand for on Campus parking at Loyola Marymount University without development of excessive or unnecessary parking, and relocate parking in closer proximity to the academic, residential, and recreational core areas of the Campus in order to reduce traffic trips through Campus and improve utilization of parking facilities and pedestrian safety;
- J. Allow Loyola Marymount University to offer Ancillary Uses within designated areas on Campus that are comparable to those provided at other major universities; and
- K. Permit the Campus' development to meet the evolving religious needs and demands of Loyola Marymount University's patrons and alumni.

Section 2.2 SPECIFIC PLAN SETTING AND DEVELOPMENT OVERVIEW

- A. The Specific Plan Area is approximately 142 acres located in the Westchester area in the southwestern portion of the City approximately 15 miles west of downtown and less than one mile north of Los Angeles International Airport. The Specific Plan Area is bounded generally by Lincoln Boulevard to the west, the Westchester bluffs and Playa Vista to the north, McConnell Avenue to the east, 80th Street to the south, and Fordham Road and 78th Street to the southwest.
- B. At the time of approval of this Specific Plan, Campus development in the Specific Plan Area totals approximately 2.5 million square feet of Floor Area, which includes approximately 848,000 square feet of Floor Area of facilities for Residential Use with approximately 3,261 beds, approximately 1.486 million square feet of Floor Area of facilities for Academic Use and Administrative Use, and approximately 166,000 square feet of Floor Area of indoor facilities for Athletic Use. The Specific Plan Area also includes approximately 15.2 acres of outdoor facilities for Athletic Use and approximately 25 acres for Open Space Use.
- C. The Project Development Plan proposed with this Specific Plan includes 457,000 square feet of net new Floor Area of facilities for Academic and Administrative Uses, 428,000 square feet of net new Floor Area of facilities for Residential Uses with approximately 989 new beds, 26,000 square feet of net new Floor Area for indoor facilities for Athletic Uses, a net increase of approximately 4.8 acres of outdoor facilities for Athletic Uses, and a net increase of approximately 5 acres of property for Open

Space Uses. Total Project Development Plan proposed with this Specific Plan is approximately 3.41 million square feet.²

Section 2.3 DEFINITIONS

The following words, whenever used in this Specific Plan, shall be construed as defined below. Words and phrases not defined herein shall be construed as defined in the Los Angeles Municipal Code. Terms shown in bold in this Specific Plan are defined in this Subsection 2.3 or in Exhibit 6.

Academic Use. Academic Uses consisting of the following uses and activities:

- (a) Instruction, including credit and noncredit courses for academic, vocational, recreational, and technical instruction, for remedial and tutorial instruction, and for regular, special, and extension sessions;
- (b) Research and scholarship, including all activities specifically organized to produce research and scholarship outcomes, whether commissioned by an agency external to Loyola Marymount University or separately by an organizational unit within Loyola Marymount University;
- (c) Non-athletic extracurricular activities generally associated with a private university including, but not limited to, clubs and organizations, commencement, job-fairs, forums, conferences, lectures, rallies, concerts, and receptions; and
- (d) Service and support uses associated with such academic and research uses and activities.

Administrative Use. Administrative Uses consisting of uses and activities in support of Loyola Marymount University, such as executive management, fiscal operations, general administrative and logistical services, fund development, information technology services, admissions, student services, public relations, development, and facilities management, maintenance, and services.

Administrative Clearance. A process to determine compliance of a project with all applicable provisions of the Specific Plan, as issued by the Director pursuant to Section 8 of this Specific Plan.

Alcohol Use Approval. A Director's approval or conditional approval of each application for the on-site sale or dispensing for consideration of

² The square footage of the Project Development Plan was derived by reducing the gross square footage analyzed in the LMU Master Plan Project Environmental Impact Report by ten percent.

alcoholic beverages in accordance with Subsection 3.8 of this Specific Plan.

Ancillary Use. Uses consisting of activities customarily incidental to a university including, but not limited to, conferences, seminars, meetings, receptions, filming, student unions, bookstores, cafés, food establishments, convenience stores, gift shops, dry cleaners, and laundries.

Athletic Use. Athletic uses and activities including, but not limited to, intercollegiate sports, intramural sports, club sports, practices, and training clinics and camps for students, faculty, staff, alumni, and guests. Athletic Uses also allow for related Athletic Use facilities such as indoor and outdoor facilities such as sports fields, courts, tracks, gymnasia, pools, natatoriums (indoor pools), arenas, stadiums, athletic offices, fitness centers, locker rooms, and other associated uses. Athletic Use facilities may be located at or above grade, or on rooftops, including on the rooftops of parking structures.

Campus. The Westchester Campus of Loyola Marymount University governed by this Specific Plan.

Director. The Director of the Department of City Planning, or his or her designee.

Floor Area. The area in square feet confined within the interior face of the exterior walls of a building, but not including the area of the following: exterior walls; stairways; shafts; light courts; rooms housing building-operating equipment, machinery, recycling or waste management equipment or machinery; parking areas with associated driveways and ramps; space for the landing and storage of helicopters; basement storage areas; and outdoor eating areas.

FTEs. Full-Time Equivalents, used as a unit of measurement to calculate student enrollment and faculty/staff population for academic and land use planning purposes. One FTE undergraduate student is defined as one undergraduate student taking 12 course units, which represents a full course load. One FTE graduate student is defined as one graduate student taking 9 course units. One FTE staff member is equivalent to one full-time staff member working 40 hours per week, or two part-time staff members working 20 hours per week. One FTE faculty member is equivalent to one full-time faculty member, or three part-time faculty members.

Grade. The lowest point of elevation of the finished surface of the ground, paving or sidewalk within the area between the building and the property

line, or when the property line is more than 5 feet from the building, between the building and a line 5 feet from the building.

Height. For Height Area 1, Height is the vertical distance above mean sea level measured to the highest point of the roof, structure, parapet wall, or skylight, whichever is highest. For Height Areas 2 and 3, Height is the vertical distance above Grade measured to the highest point of the roof, structure, or the parapet wall, whichever is highest.

Height Area. The Specific Plan is divided into three Height Areas as shown on Exhibit 3, Height Areas, designated as follows:

- (e) **Height Area 1**, governing that portion of Campus also known as Hughes Campus;
- (f) **Height Area 2**, governing that portion of Campus also known as Leavey Campus with the exception of Drollinger Plaza and property along 78th Street, and the core of Burns Campus; and
- (g) **Height Area 3**, governing that portion of Campus also known as Burns Campus and Drollinger Plaza and property along 78th Street.
- (h) LMU. Loyola Marymount University.

LMU Master Plan. LMU's 20-year development plan to upgrade, renovate, and replace a substantial portion of the existing academic, administrative, residential and athletic facilities at its Westchester campus.

Open Space Use. Open Space Uses include the following uses and activities:

- (a) Picnicking, children's play, passive activities such as lounging, reading, studying, and similar activities;
- (b) Recreational activities including, but not limited to, viewing nature, walking, jogging, and bicycling;
- (c) Educational and cultural gatherings;
- (d) Special events, which include events not scheduled on a regular weekly, monthly, or annual basis as part of LMU's calendar of events and activities;
- (e) Similar activities and uses, as determined by the Director.

Planning Area: The Specific Plan Area is divided into six Planning Areas designated as follows: Academic/Residential (A/R); Athletic (A); Campus

Core/Open Space (CC/OS); Buffer/Open Space 1 (B/OS1); Buffer/Open Space 2 (B/OS2); and Buffer/Open Space 3 (B/OS3).

Project. The construction, erection, addition to, or structural alteration or modification of any building or structure, or use of building or land, or change of use of a building or land located within the Specific Plan Area, which requires the issuance of a grading permit, foundation permit, building permit, or use of land permit, and which results in a net increase of Floor Area, after the effective date of this Specific Plan.

Project Development Plan. The maximum amount of Floor Area of facilities for permitted uses, which may be developed within the Specific Plan Area totaling 3,411,000 square feet.

Protected Tree. A Protected Tree shall be as defined in Sections 17.02 and 46.01 of the Municipal Code as of the date of adoption of this Specific Plan.

Recycling and Waste Management Uses. Activities associated with the temporary storage, staging, and removal off-Campus of recyclable materials, trash and other waste materials generated by LMU, including the sorting, baling, crushing, and transfer of these materials. Recycling and Waste Management Uses do not include collection of recycling and waste materials from individual users of the materials in receptacles throughout Campus.

Religious Uses. Activities customarily associated with LMU's religious traditions, including religious services such as installations, convocations, weddings and funerals, instruction, and ministry at Sacred Heart Chapel and other religious locations on Campus.

Residential Use. Housing for students, visitors, faculty, the Jesuit community, and staff with amenities and facilities associated with, and located within the same building as such housing, including but not limited to kitchens, common rooms, recreation rooms, fitness areas, lounges, study rooms, laundries, and service facilities. Classrooms or other facilities for Campus-wide Academic Use, which are located within a primarily Residential Use building are not considered Residential Use facilities.

Specific Plan. The Loyola Marymount University Specific Plan, Ordinance No. _____, implementing the LMU Master Plan.

Specific Plan Area. That area within the Specific Plan, shown within the heavy dashed lines on Exhibit 1.

SECTION 3: LAND USE REGULATIONS

Section 3.1 PURPOSE AND INTENT

The land use regulations established by this Specific Plan specify the distribution, location, and extent of the uses of land, including open space, within the Specific Plan Area.

Section 3.2 PROHIBITIONS

- A. Except as provided herein, no grading permit, foundation permit, building permit, use of land permit, or permit for a change of use shall be issued for any Project on any lot located in whole or in part within this Specific Plan Area unless the Director has issued an Administrative Clearance pursuant to Section 8 of this Specific Plan.
- B. No Administrative Clearance shall be approved for a Project that would cause overall development in the Specific Plan Area to exceed 457,000 square feet of net new Floor Area of facilities for Academic and Administrative Uses, 428,000 square feet of net new Floor Area of facilities for Residential Uses with approximately 989 net new beds, and 26,000 square feet of net new Floor Area for indoor facilities for Athletic Uses.
- C. No Administrative Clearance shall be issued unless the Project complies with all applicable provisions of this Specific Plan, including compliance with all applicable Project Design Features and Mitigation Measures as set forth in the Mitigation Monitoring and Reporting Program in Appendix A of this Specific Plan.
- D. Exceptions and Exemptions. The prohibitions specified in this Subsection shall not apply to any of the uses and activities exempted from an Administrative Clearance, as set forth in Section 8 of this Specific Plan.

Section 3.3 LAND USE PLAN AND CAMPUS LIMITATIONS

A. General Description. To allow new development, the Specific Plan is divided into six Planning Areas designated as follows: Academic/Residential (A/R); Athletic (A); Campus Core/Open Space (CC/OS); Buffer/Open Space 1 (B/OS1); Buffer/Open Space 2 (B/OS2); and Buffer/Open Space 3 (B/OS3) as shown on the Planning Area figure (Exhibit 2). The boundary between the Academic/Residential Planning Area and the Athletic Planning Area, depicted on Exhibit 2, may shift by up to 50 feet so long as there are no changes to the boundaries of the Campus Core/Open Space or Buffer/Open Space Planning Areas.

B. **Maximum Permitted Floor Area Development**. The maximum permitted development shall be as set forth in Table No. 1.

Facilities					
Land Use	Existing (sf)	To Remain (sf)	New (sf)	Net New (sf)	Total (sf)
Academic / Administrative	1,486,000	1,022,000	921,000	457,000	1,943,000
Residential	848,000	515,000	761,000	428,000	1,276,000
Athletic Indoors	166,000	95,000	97,000	26,000	192,000
Totał	2,500,000	1,632,000	1,779,000	911,000	3,411,000

Table No. 1 Project Development Plan (square footage of Floor Area)

- C. **Residential Capacity**. The number of individuals permitted to reside on Campus shall be limited to 4,250 students, faculty, staff, visitors, or religious affiliates.
- D. **Enrollment Capacity**. The number of students enrolled at the Campus governed by this Specific Plan shall be limited to 7,800 FTE students.
- E. **Unified Campus**. The development limits and open space requirements in this Specific Plan shall be calculated on a Campus-wide basis. The development limits and open space requirements of this Specific Plan apply regardless of the recordation of any lot or parcels within the Specific Plan Area.
- F. Initial Specific Plan Development. Following the effective date of this Specific Plan, the Campus recycling and waste management area shall be relocated from its current location adjacent to the Facilities Management Area to the lower level of Drollinger Parking Plaza as the first Project implemented under this Specific Plan.

Section 3.4 PERMITTED USES AND FACILITIES

The following uses and facilities shall be permitted within the Specific Plan Area, as permitted by Subsection 3.5:

- A. Academic Uses.
- B. Administrative Uses.
- C. Ancillary Uses.

- D. Athletic Uses.
- E. Child Care Facilities.

F. Civic and community meetings and activities.

- G. Columbarium.
- H. Exercise stations.
- I. Gardens, plazas, fountains, and water features.
- J. Infrastructure.
- K. Open air seating.
- L. Open Space Uses.
- M. Parking lots and structures.
- N. Recycling and Waste Management Uses.
- O. Religious Uses.
- P. Residential Uses.
- Q. Signs, permitted by Section 7 of this Specific Plan.
- R. Storage.

Section 3.5 PERMITTED USES AND FACILITIES BY PLANNING AREA

- A. Planning Areas. The Specific Plan is divided into six Planning Areas designated as follows: Academic/Residential (A/R); Athletic (A); Campus Core/Open Space (CC/OS); Buffer/Open Space 1 (B/OS1); Buffer/Open Space 2 (B/OS2); and Buffer/Open Space 3 (B/OS3) as shown on Exhibit 2, Planning Areas. The following is a list of these Planning Areas with the land uses and facilities permitted within each. Those land uses and facilities not expressly permitted by this Specific Plan are prohibited within the Planning Areas. No structures with Floor Area are permitted in any of the Buffer/Open Space Planning Areas.
- B. Academic/Residential Planning Area. The following land uses and facilities are permitted:
 - 1. Academic Uses.
 - 2. Administrative Uses.

- 3. Ancillary Uses.
- 4. Child Care Facilities.
- 5. Civic and community meetings and activities.
- 6. Columbarium.
- 7. Gardens, plazas, fountains, and water features.
- 8. Infrastructure, including any below-ground infrastructure along with ancillary above-ground equipment.
- 9. Open air seating.
- 10. Open Spaces Uses.
- 11. Parking lots and structures.
- 12. Recycling and Waste Management Uses.
- 13. Religious Uses.
- 14. Residential Uses.
- 15. Signs, permitted by Section 7 of this Specific Plan.
- 16. Storage.
- C. Athletic Planning Area. The following land uses and facilities are permitted:
 - 1. Academic Uses.
 - 2. Administrative Uses.
 - 3. Ancillary Uses.
 - 4. Athletic Uses.
 - 5. Child Care Facilities.
 - 6. Civic and community meetings and activities.
 - 7. Exercise stations.
 - 8. Gardens, plazas, fountains, and water features.

- 9. Infrastructure, including any below-ground infrastructure along with ancillary above-ground equipment.
- 10. Open air seating.
- 11. Open Space Uses.
- 12. Parking lots and structures.
- 13. Recycling and Waste Management Uses.
- 14. Religious Uses.
- 15. Signs, in accordance with Section 7 of this Specific Plan.
- 16. Storage.
- D. **Campus Core/Open Space Planning Area**. The following land uses and facilities are permitted:
 - 1. Academic Uses, not including structures with Floor Area.
 - 2. Athletic Uses, not including structures with Floor Area.
 - 3. Civic and community meetings and activities.
 - 4. Gardens, plazas, fountains, and water features.
 - 5. Infrastructure, including any below-ground infrastructure along with ancillary above-ground equipment.
 - 6. Open air seating.
 - 7. Open Space Uses.
 - 8. Religious Uses.
 - 9. Signs, permitted by Section 7 of this Specific Plan.
- E. **Buffer/Open Space 1 Planning Area.** The following land uses and facilities are permitted:
 - 1. Academic Uses, not including structures with Floor Area.
 - 2. Exercise stations.
 - 3. Gardens, plazas, fountains and water features.

- 4. Infrastructure, including any below-ground infrastructure along with ancillary above-ground equipment.
- 5. Open air seating.
- 6. Open Space Uses.
- 7. Religious Uses.
- 8. Signs, permitted by Section 7 of this Specific Plan.
- F. **Buffer/Open Space 2 Planning Area.** The following land uses and facilities are permitted:
 - 1. Academic Uses, not including structures with Floor Area, except that commencement activities, job fairs, forums, conferences, rallies, concerts, and receptions shall not be permitted.
 - 2. Gardens, plazas, fountains and water features.
 - 3. Infrastructure, including any below-ground infrastructure along with ancillary above-ground equipment.
 - 4. Temporary open air seating.
 - 5. Open Space Uses, except that this Planning Area shall not be used to satisfy the open space requirements of any child care center operating on Campus. However, informal children's play shall be permitted.
 - 6. Signs, permitted by Section 7 of this Specific Plan.
- G. **Buffer/Open Space 3 Planning Area.** The following land uses and facilities are permitted:
 - 1. Academic Uses, not including structures with Floor Area.
 - 2. Infrastructure, including any below-ground infrastructure along with ancillary above-ground equipment;
 - 3. Open Space Uses.
 - 4. Parking lots.
 - 5. Signs, permitted by Section 7 of this Specific Plan
 - 6. Storage, not including structures with Floor Area.

Section 3.6 NOISE REGULATIONS

- A. No source of outdoor amplified sound shall be installed or maintained on Campus within 150 feet of residential areas in the R1 zone.
- B. All outdoor amplified sound within the Athletic Planning Areas shall be turned off by 10:00 p.m., except that outdoor amplified sound within the Athletic Planning Areas used in conjunction with athletic training and competitions may continue after 10:00 p.m. only in the case of overtime or extra innings. Outdoor amplified sound within the Athletic Planning Areas shall have an absolute cut-off time of 10:00 p.m. for athletic training and competitions in which LMU is not a participant (i.e. not a participant in the league, play-offs, or tournament). Outdoor amplified sound shall be prohibited in the Athletic Planning Area in the northeast corner of Campus shown in Exhibit 2. The use of air horns shall be prohibited on Campus.
- C. All speakers for modified and new outdoor amplified sound systems within the Athletic Planning Areas shall be mounted to face spectator areas, be directed away from adjacent residences, and be set to provide that sound levels from the systems do not exceed the ambient noise levels in Exhibit 7 by 5 dB(A).
- D. Organized use of the open space area behind University Hall shall be restricted to small classes or informal university activities of less than 25 people. No amplification shall be permitted behind University Hall. Use of the area shall be restricted to the hours of 8:00 am to 6:00 pm.
- E. Both points of Campus entry, at LMU Drive and at Loyola Boulevard, and internal roads along the eastern border of the Campus, shall be posted with signs advising drivers that they are entering a residential quiet zone.
- F. If a new sports pavilion is constructed near the corner of 80th Street and Fordham Road, the structure shall be designed such that the main entrance of the new sports pavilion faces away from 80th Street and Fordham Road, and shall be designed to meet the following requirements:
 - The pedestrian walkways and main entries providing primary access to the sports pavilion shall be located in the interior portion of Campus, not adjacent to Fordham Road or 80th Street.
 - 2. Outdoor generators, generator trucks, television trucks, staging vehicles servicing events at the sports pavilion, and

any event technology such as outdoor amplified sound systems shall be prohibited on Campus between the sports pavilion and Fordham Road or 80th Street.

- 3. Buses transporting people to events at the sports pavilion shall be prohibited from parking on Campus between the sports pavilion and Fordham Road or 80th Street. LMU shall direct such buses to park in the interior portion of Campus or at a designated off-Campus area, and shall direct buses not to park in the adjacent residential community.
- 4. No access road shall be constructed between the sports pavilion and Fordham Road, or between the sports pavilion and 80th Street, unless required by the Los Angeles Fire Department. If such an access road is required by the Los Angeles Fire Department, it shall only be used for emergency purposes.
- G. New parking structures shall include a half wall on the grade-level parking deck and/or full walls on the sides of the parking structure that face nearby residential receptors.
- H. Construction staging shall be prohibited in any residential areas, or on the Campus within 150 feet of any residential perimeter prior to 7:00 am.
- I. No paging system shall be installed that is audible outside the building in which it is located.
- J. Emergency address broadcasts shall be exempted from the requirements of this Section.

Section 3.7 HISTORIC PRESERVATION REGULATIONS

- A. Renovation and rehabilitation of Sacred Heart Chapel, St. Robert's Hall, and Xavier Hall shall be permitted, provided that such work conforms to the Secretary of the Interior's Standards. Prior to the issuance of a building permit for any exterior, or structural work on these structures, LMU shall obtain written approval of conformance from the Department of City Planning, Office of Historic Resources, which may be provided in the form of Planning's sign-off on the requested building permit.
- B. Prior to the issuance of any permit for the demolition of or exterior or structural modification of the Edward T. Foley Center, LMU shall undertake additional historic resources assessment in coordination with the Department of City Planning, Office of Historic Resources, of the subject building's eligibility for the National Register of

Historic Places, California Register of Historic Resources, and City Historic-Cultural Monument status. If the building is identified as a historical resource, written approval shall be obtained from the Office of Historic Resources that such work conforms to the Secretary of the Interior's Standards for the Treatment of Historic Properties.

C. No demolition permit for the Edward T. Foley Center shall be issued prior to the issuance of a building permit for a replacement Project on the site of the existing Edward T. Foley Center.

Section 3.8 OPERATIONAL REGULATIONS FOR ON-SITE SALES, SERVICE, AND CONSUMPTION OF ALCOHOL

- A. **General Provision**. The sale and service of alcoholic beverages for on-site consumption shall be permitted anywhere on Campus, except that the sale of alcoholic beverages to General Admission seating within sports facilities during sporting events shall be prohibited. Entities that sell and serve alcoholic beverages for onsite consumption shall obtain approvals from other agencies, as required, including licenses or permits from the State Department of Alcoholic Beverage Control (ABC).
- B. Alcohol Use. Alcohol Use Approvals shall be required for permanent establishments on Campus, such as restaurants and dining facilities that operate on an ongoing daily basis, seeking new approvals for the sale or dispensing for consideration of alcohol. No Alcohol Use Approval shall be issued unless the Director determines that the on-site sale or dispensing for consideration of alcohol meets the requirements of Exhibit 5, and the Director may impose conditions of approval as deemed necessary to ensure community welfare. The grant or denial of an alcohol use approval is appealable under the procedures of Los Angeles Municipal Code § 12.24-W.

In addition to sale or dispensing for consideration of alcohol at permanent establishments, the sale and dispensing for consideration of alcohol may be permitted at on-campus events, for which an ABC permit has been obtained. The first 40 such events in a calendar year, starting July 1 of each year, shall not require an Alcohol Use Approval, but shall adhere to the requirements in Exhibit 5. LMU shall notify the Director in writing at least 5 days prior to the day of the event and provide details including the nature and location of the event, hours, and the approximate number of persons expected to attend. For any events occurring after the first 40 on-campus events involving the sale or dispensing for consideration of alcohol, LMU shall be required to obtain an Alcohol

Use Approval, which may be approved or conditionally approved by the Director. LMU shall advise the Director in writing four times a year, on a quarterly basis, of the total number of on-campus events at which alcohol was sold or dispensed for consideration.

C. Modification. If the conditions of the Alcohol Use Approval, the requirements set forth in Exhibit 5 or the requirements of this section have not been complied with, or the Director finds that the subject use adversely affects the welfare of the pertinent community or nearby residentially zoned communities, the City may give notice to the property owner and lessee of the real property affected to appear at a time and place fixed by the City and show cause why the use permitted by this Subsection should not be modified, discontinued or revoked. An order to modify, discontinue or revoke the use may be appealed to the appellate body specified in Los Angeles Municipal Code § 12.24-W.

Section 3.9 CAMPUS OPERATIONAL REGULATIONS

- A. The end time for auditorium/theater events attended by off-Campus patrons shall be 11:00 p.m., Sunday through Thursday and 12:00 a.m., Friday and Saturday evenings. Academic productions, with substantial student involvement, are specifically exempted.
- B. The regular hours of operation for classes in University Hall shall be from 8:00 A.M. until 10:00 P.M., seven days a week.
- C. The maximum occupancy for existing University Hall shall be limited to 2,600 persons.

SECTION 4: CAMPUS DESIGN REGULATIONS

Section 4.1 DESIGN REGULATIONS

- A. Building Height. The Heights of buildings, structures and Signs shall not exceed the Heights set forth in Exhibit 3.
 - 1. The Height of 139 feet above mean sea level in Height District 1 shall be measured to the highest point of any skylights.
- B. Setbacks. The only required yards or setbacks of buildings and structures shall be as set forth in Exhibit 4.

- 1. Storage and storage containers shall be prohibited within the first twenty feet of land adjacent to the property line along the McConnell Avenue Campus boundary.
- C. Minimum Open Space Requirements. At least 25 acres of open space shall be retained in the Open Space Planning Areas.
- D. Outdoor Athletic Area Requirements. At least 13 acres of outdoor Athletic Use fields and other Athletic Use facilities (such as a pool and/or diving well) shall be provided in the Athletic Planning Area. Athletic fields may be grass or artificial.
 - 1. Athletic fields and other outdoor athletic facilities shall be designed such that any outdoor spectator seating shall be located at least 120 feet from any off-Campus residentially-zoned property.
- E. Regulation of Protected Trees shall be governed by Section 17.05.R, Section 17.06.C., and Section 46.00, et seq. (Chapter IV, Article 6) of the Municipal Code as of the date of adoption of this Specific Plan.
- F. Lighting Regulations: The requirements of the Municipal Code shall apply with the following additional requirements:
 - 1. Pole mounted lights shall be designed to minimize potential off-site light spillage.
 - 2. Light sources shall be shielded so as to reduce visibility from off-site locations.
 - 3. No floodlighting shall be located so as to shine directly onto any adjacent residential property.
 - 4. Installation of low level security and exit lighting shall be allowed.
 - 5. Lighting fixtures and visors shall be adjusted upon installation to reduce spillover onto adjacent residential properties, while still maintaining adequate lighting to allow safe use of outdoor athletic facilities. Additionally, vegetation and other screening or filtering devices shall be maintained or supplemented at the edges of lit fields or at the campus perimeter at all times, such that light spillover shall not be permitted at any time to exceed an intensity of 2 foot-candles on residential property lines located along W. 78th Street, Fordham Road, W. 80th Street, and McConnell Avenue.

- 6. All outdoor lighting shall be directed downward to illuminate the intended surface (i.e., playing fields, pedestrian pathways and other high-traffic areas such as building entrances and plazas in the campus interior).
- 7. All new outdoor lighting shall be equipped with louvers, shields, hoods, or other screening devices.
- 8. The use of field lighting within all outdoor athletic facilities shall be limited to only those hours during which the facilities are being utilized, which shall not surpass 10:00 p.m., except in the case of overtime or extra innings.
- G. Landscaped Buffers: All landscaped buffer areas described in this subsection shall be maintained in good condition, subject to local and state regulations regarding water use, drought-tolerant vegetation, and other landscaping requirements.
 - 1. Landscaped Buffer along Altavan Ave. and 78th Street. The established buffers shall be maintained as indicated by the existing fence lines in these locations.
 - 2. Landscaped Buffer along 80th Street. The established buffer of 20 feet shall be maintained.
 - 3. Landscaped Buffer along Fordham Road. LMU shall maintain the established 15 foot landscaped buffer along Fordham Road and the Campus edge, and shall provide enhanced landscaping prior to issuance of permits for new construction on the Athletic Planning Area located adjacent to Fordham Road and 80th Street. Prior to implementation of a landscaping plan, the landscaping plans for enhancement shall be provided to the Council Office representing the Campus' District 11 for a 30 day period to review with adjacent neighbors.
 - 4. Landscaped Buffer along McConnell Avenue: Condition for First Two Projects (Recycling Relocation and Life Sciences Building).
 - a) Existing Buffers to Remain Areas A and D. The existing landscaped buffers adjacent to Areas A and D shown in Exhibit 8 shall be maintained at the current width. Prior to the addition of seating or lighting at Page Baseball Stadium or Smith Softball Field, the landscaped buffer adjacent to Area A shall be enhanced with additional vegetation. Prior to implementation of a landscaping plan for this area, the landscaping plan shall be provided to the Council Office

representing the Campus' District for a 30 day period to review with adjacent neighbors.

- b) Area B. Upon relocation of the recycling and waste management area to Drollinger Parking Plaza, pursuant to Section 3.3.F of this Specific Plan, an interim condition shall be created to accommodate the need to replace the 83 parking spaces in Drollinger Parking Plaza to be displaced by this relocation. These spaces shall be relocated to the area adjacent to Area B. These parking spaces shall be located a minimum of 12 feet from the adjacent property line, and shall be limited to use by LMU employees and contractors. During this interim condition, a 12 foot landscaped buffer shall be enhanced with thick vegetation. Prior to implementation of a landscaping plan for this area, the landscaping plan for enhancement shall be provided to the Council Office representing the Campus' District for a 30 day period to review with adjacent neighbors.
- c) Area C. The houses located in Area C shown in Exhibit 8 are approximately 10 feet below the grade of the adjacent edge of Campus. A five foot landscaped buffer adjacent to Area C shall be maintained and tall trees and/or hedge-type landscaping shall be added on the LMU side in connection with the relocation of the recycling and waste management area to Drollinger Parking Plaza pursuant to section 3.3 F of this Specific Plan. Prior to implementation of a landscaping plan for this area, the landscaping plan shall be provided to the Council Office representing the Campus' District for a 30 day period to review with adjacent neighbors.
- 5. Landscaped Buffer along McConnell Avenue: Condition at Full Buildout. Please refer to Section 4.1.I.2 below.
- The enhanced landscaped buffers along Fordham Road and McConnell Avenue described in Subsections 4.1.H.3 and 4.1.H.4 shall be consistent with the following guidelines:
 - a. Groupings of plant materials will be provided within the landscaped buffer areas to substantially screen views into Campus from adjacent residences. Such groupings will include vegetation of varied heights and thicknesses, and will be arranged in layers to add density, in order to minimize visibility into Campus and to provide a visual buffer.

1

- b. Included in Exhibit 9 is a palette of plants chosen for their density, expected size, relative short time to maturity, and drought-tolerance which may be included in the landscaped buffers. Plants with similar properties may also be planted in the landscaped buffer areas.
- c. The existing mature trees within the 12 feet adjacent to Area B shall be preserved during relocation of the recycling and waste management area to Drollinger Parking Plaza, and the relocation of the Drollinger Parking Plaza parking spaces to the area adjacent to Area B. The existing mature trees adjacent to Area D located south of Rosecrans Lane shall be preserved as part of the Master Plan.
- d. Any existing significant mature trees located within the landscaped buffer areas that dies or is damaged due to accident, disease, weather, or other cause will be replaced by a tree or other landscaped element that is anticipated to provide equivalent screening upon maturity.
- 7. For the purposes of determining the locations of existing fences and the widths of existing buffers described in Sections 4.1.H and 4.1.I, reference shall be made to the property survey conducted by Mollenhauer Group dated 2010, which is on file with the City of Los Angeles.
- H. Permitted Uses Within Building Setback Adjacent to McConnell Avenue.
 - Buildings shall be prohibited within the 40 foot setback area adjacent to the McConnell Avenue boundary of Campus. Parking within the 40 foot setback area adjacent to the McConnell Avenue boundary shall be permitted except as provided in Subsection 4.1.I.2 below. Parking within the triangular parking area adjacent to Area D, shown in Exhibit 8, shall be permitted.
 - 2. As the Master Plan is developed, the existing landscaped buffers adjacent to Area A and Area D (north of Rosecrans Lane) shall be maintained at the current width and the triangular parking area adjacent to Area D, shown in Exhibit 8, shall be permitted to be maintained and improved and further developed pursuant to the Specific Plan, provided that the minimum landscaped buffer shown in Exhibit 8 is not reduced.

As each of the areas adjacent to the McConnell Avenue boundary parallel to Areas B, C, and D (south of

Rosecrans Lane) are developed pursuant to the Master Plan, the uses within the 40 foot setback shall be limited to landscaped buffers, a fire lane expected to be 20 to 28 feet in width (as determined by the Los Angeles Fire Department), and a sidewalk on the LMU side of the fire lane: parking shall no longer be permitted. The width of each of these components shall be determined prior to obtaining a building permit for the first building parallel to this area, starting with the determination of the Fire Department as to the required width of the fire lane. LMU shall use good faith efforts to request the Fire Department to grant approval of a fire lane 20 feet in width, recognizing that the Fire Department's determination will govern the width of the fire lane to be required under its rules. In addition, prior to implementation of a landscaping plan for this area, a landscaping plan shall be provided to the Council Office representing the Campus' District for a 30day period to review with adjacent neighbors. As part of this landscaping plan, the existing mature trees adjacent to Area D located south of Rosecrans Lane shall be preserved.

- 3. The fire lane to be constructed adjacent to the McConnell Avenue boundary shall be used exclusively for fire truck access and as a university service road, which includes maintenance activities and emergencies.
- 4. Storage shall be prohibited within 20 feet adjacent to the McConnell Avenue boundary, until full buildout when storage shall be prohibited within 40 feet adjacent to McConnell Avenue. In no event shall the landscaped buffer adjacent to the McConnell Avenue boundary be less than 10 feet upon full buildout, and such minimum shall be extended where necessary adjacent to Area D located south of Rosecrans Lane to preserve the existing mature trees.
- I. Screening/Walls.
 - 1. Any structures on the roofs of new buildings, such as air conditioning units and other equipment, shall be fully screened from view by any nearby single-family residential property, with the exception of communications equipment requiring an outside signal, or equipment requiring direct sunlight (i.e. solar panels.).

- 2. All above-grade parking levels shall be fully screened from view by any nearby single-family residential property.
- J. Balconies. New residential buildings constructed adjacent to the eastern boundary of campus shall not include balconies above the first floor within 100 feet of any adjacent single-family use.
- K. Recycling and Waste Management Uses. Recycling and Waste Management Uses shall be located at least 40 feet from the nearest property line of any off-Campus residentially zoned property if they completely are enclosed within a building or structure, or 150 feet from any such property if not fully enclosed.
- L. Residential Uses. Dedication of land and fees associated with LAMC sections 12.33, 17.07, 17.12 or 17.58 is not required in connection with the issuance of building permits for student-housing on Campus.

SECTION 5: TRANSPORTATION AND PARKING REGULATIONS

Section 5.1 VEHICULAR AND PEDESTRIAN CIRCULATION PLAN/ACCESS.

- A. LMU Drive shall be the primary vehicular route for visitors, deliveries, and truck traffic to Campus. Loyola Boulevard shall be a gate-controlled access route.
 - Vehicular use of the Loyola Boulevard campus entrance shall be restricted to a limited number of vehicles, and LMU shall utilize keycards, color coded parking stickers, or a similar system to separately identify the vehicles authorized to enter and exit the campus via the Loyola Boulevard gate. All other vehicles shall be required to use the LMU Drive entrance from Lincoln Boulevard, except for emergency and maintenance vehicles as necessary.
 - 2. LMU shall maintain the existing signs posted at Loyola Boulevard directing visitors to use the LMU Drive entrance to Campus, and shall cooperate with the Departments of City Planning and Transportation to provide additional signage on Manchester Avenue, Lincoln Boulevard, and/or other streets in the area to direct traffic destined for the university, should those City Departments determine such to be necessary.
- B. All commercial vehicles in excess of 6,000 pounds gross weight shall use LMU Drive. Other driveways to the Campus may be

permitted, but shall be gated and locked and used only for public utility, maintenance, and emergency purposes.

- C. All deliveries shall be through the main gate at LMU Drive and Lincoln Boulevard. All vendor contracts shall include the requirement to limit trucks' (delivery, trash collection, etc.) ingress and egress access to only LMU Drive during the hours of 7:00 a.m. to 4:00 p.m. Monday through Friday and 9:00 a.m. to 2:00 p.m. on Saturday. Deliveries shall be restricted to off-peak hours wherever feasible. No trash, recycling, or delivery trucks will be allowed on Sundays. However, exceptions shall be allowed for events not scheduled on a regular weekly, monthly, or annual basis as part of LMU's calendar of events and activities.
- D. A fence shall be maintained along the southern and eastern boundaries of Height Area 1, and along the western, southern, and eastern boundaries of Height Area 3, shown on Exhibit 3. Such fences may include controlled access gates, however these gates shall remain locked except for emergencies or necessary maintenance, and as expressly allowed herein.
 - 1. The fence along Altavan Avenue shall be maintained at least six feet in height and the fences along 78th Street, Fordham Road, and 80th Street up to Loyola Boulevard shall be maintained at least eight feet in height measured from the lowest point of such fence. The fence along 80th Street up to Loyola Boulevard shall be increased to the height specified no later than the issuance of a Certificate of Occupancy for a sports pavilion constructed near the corner of Fordham Road and 80th Street. The fences along 78th Street and Fordham Road shall be increased to the heights specified within one year of the date that LMU begins charging for parking on Campus.
 - 2. Vehicular and pedestrian access to the Campus shall be prohibited from Altavan Avenue, 78th Street, or Fordham Road, except for emergencies or necessary maintenance.
 - 3. The pedestrian gate located along the McConnell Avenue property line shall remain permanently closed and locked, except that emergency access shall be permitted.
 - 4. No fence is required along the Lincoln Boulevard frontage of the Campus.
- E. LMU shall maintain a bike path connecting LMU Drive to Loyola Boulevard.

- F. Site Access and Internal Circulation. LMU shall submit a driveway access and internal circulation plan to the Los Angeles Department of Transportation (LADOT) WLA/Coastal Development Review Section for review and approval prior to the construction of any Project pursuant to this Specific Plan, and prior to any change to the driveway access and internal circulation scheme.
- G. Transportation Demand Management Program. LMU shall implement and monitor a Transportation Demand Management (TDM) program in accordance with the guidelines set forth in the Transportation Impact Analysis prepared in December 2009 for the Proposed Project. The final, detailed TDM plan shall be presented to the Los Angeles Department of Transportation for approval. If at any time it is determined that required trip reductions from the program are not being met, then LMU will be required to provide the action(s) necessary to bring the program into compliance, including but not limited to such measures as a reduction in the proposed Campus FTE growth level and implementing a variety of alternative TDM measures that prove to be more effective in reducing trips to Campus.
 - 1. LMU shall submit a plan detailing the actions to be taken to demonstrate that the goals of the TDM effort will be met to LADOT for approval prior to the campus population exceeding 8,500 FTE students, faculty and staff. Final approval of this plan shall be required prior to increasing the total campus population to the level of 9,000 FTE students, faculty and staff.
 - 2. LMU shall submit an annual report to LADOT with the campus population for the current school year to ensure these thresholds are not exceeded without submitting the required TDM Plan first.

Section 5.2 PARKING REGULATIONS

- A. A minimum of 4,035 parking spaces shall be provided on Campus provided that there is no increase beyond 8,360 FTE students, faculty, and staff on Campus.
- B. As Campus FTE levels increase, there shall be at least the following minimum number of parking spaces on Campus at the following FTE levels:

Table 2 FTE – Parking Space Requirement

FTE Students, Faculty, Staff	Minimum Number of Parking Spaces on Campus		
8,360	4,035		
8,422	4,071		
8,484	4,107		
8,546	4,143		
8,608	4,179		
8,670	4,215		
8,732	4,251		
8,794	4,287		
8,856	4,323		
8,918	4,359		
8,980	4,395		
9,042	4,431		
9,104	4,467		
9,166	4,503		
9,228	4,539		
9,290	4,575		
9,352	4,611		
9,414	4,647		
9,476	4,683		
9,538	4,719		
9,600	4,742		

- C. Parking spaces may include any standard, compact, or disabled parking space including tandem, stacked, or other methods to optimize the number of parking spaces.
- D. LMU shall provide an annual report to the Department of City Planning, due in October of each year, that certifies the number of FTE students, faculty, and staff enrolled or working on Campus for the current academic year, and that includes parking plans showing that the Campus provides the number of parking spaces required by Section 5.2 of this Specific Plan.
- E. In addition LMU shall conduct an annual parking survey during October of each year. The survey shall be designed to determine peak parking demand and shall be conducted between the hours of 6:00 am and 10:00 pm for a number of days directed by LADOT which shall not be less than two days and which days shall not include Fridays, Saturdays, Sundays, or holidays.
- F. If LMU does not provide this annual report and survey by October 31, or if LMU cannot show that the Campus provides for the

number of parking spaces required by Section 5.2.B of this Specific Plan, the Department of City Planning shall not issue an Administrative Clearance for any building permit on Campus which will result in increased parking demand. An Administrative Clearance for any building permit which will result in increased parking demand shall not be issued until LMU submits documentation to the Department of City Planning providing the FTE certification and parking information required by this Subsection 5.2.D, which may be submitted at any time during the year. LMU shall have one year to provide any additional parking determined necessary by the parking survey conducted pursuant to Section 5.2.E. after which year a building permit which will result in increased parking demand shall not be issued until documentation of such additional parking is provided to the Department of City Planning, which may be submitted at any time during the year. Upon request by LADOT, LMU shall provide additional information or analysis with regard to parking demand on Campus. Prior to issuance of a building permit for any work that requires additional parking, the Department of Building and Safety shall obtain a written clearance from the Department of City Planning to verify the required and provided parking spaces.

- G. LMU shall provide approximately 190 net new parking spaces on campus in connection with construction of a new Life Sciences Building.
- Event Parking Management Program. LMU will maintain an Event H. Parking Management Program to accommodate occasional university functions expected to bring non-campus registered vehicles and over 100 non-campus affiliated visitors onto Campus. The Event Parking Management Program will provide for a temporary increase in traffic management and parking personnel to accommodate the additional vehicles and visitors on the Campus. LMU shall direct visitors leaving events by car to exit the Campus via LMU Drive. The Event Parking Management Plan shall be provided to the Council Office representing the Campus' District for a 30 day review period prior to review by LADOT. Upon the effective date of the Specific Plan, LMU shall provide LADOT with the proposed program. LADOT shall have 30 days to review and approve or disapprove the program or any updated program. The program will be approved if LADOT has not disapproved the program within 30 days.
- I. General Parking Management Program. LMU shall maintain a General Parking Management Program to manage day-to-day parking and discourage people from driving their cars to Campus. Such measures may include, but are not limited to, installing

parking lot/garage equipment to control and regulate parking for visitors, faculty, staff, resident students, and commuter students; providing visitor parking at clearly marked convenient locations; collecting and maintaining vehicle information for all campus residents parking on Campus, and all students living off-Campus; and charging for parking on Campus to provide a further disincentive for people to drive their cars to Campus. Upon the effective date of the Specific Plan, LMU shall provide LADOT with the proposed program. LADOT shall have 30 days to review and approve or disapprove the program or any updated program. The program will be approved if LADOT has not disapproved the program within 30 days.

- J. Prior to the construction of a new sports pavilion, or a new conference center on the southwest side of University Hall, LMU shall submit an updated parking study to LADOT. LMU shall provide any additional parking required by LADOT, if LADOT determines that such additional parking is needed. Prior to the construction of a new sports pavilion, or a new conference center on the southwest side of University Hall, the Event Parking Management Plan shall be updated as to such facility, and the updated Plan shall be provided to the Council Office representing the Campus' District for a 30 day review period prior to review by LADOT.
- K. LMU shall cooperate with the community, the Department of Transportation, and other City agencies as necessary, to implement a permit parking program or other program which restricts parking on neighboring residential streets, if such restrictions are requested by the community.
- L. No construction parking shall be permitted on any residential streets.

SECTION 6: SECURITY REGULATIONS

Section 6.1 GENERAL PROVISIONS

- A. LMU shall implement a security program which includes, at a minimum, the following measures:
 - 1. Maintaining operation of the Department of Public Safety, or a department of similar responsibilities;
 - 2. Maintaining a security booth located on LMU Drive, off of Lincoln Boulevard;
 - 3. Operating a network of security cameras on the Campus;

- Securing all student residence halls and apartments by an authorized card-key and/or hard-key entry, or similar security device;
- 5. Lighting subterranean parking levels 24 hours a day;
- 6. Regularly patrolling Campus, particularly parking areas during evening hours and areas around sports facilities before, during, and after sports events, and the areas surrounding University Hall;
- 7. Maintaining a hot line to LMU's Department of Public Safety available 24 hours a day, 7 days a week, for community calls about serious student behavior issues in the community. This number shall be advertised on LMU's website. A public safety officer shall be available 24 hours a day, 7 days a week, to answer such phone calls. LMU shall make a public safety officer available Thursdays, Fridays, and Saturdays from 6:00 pm to 2:00 am to address complaints about student disturbances off-Campus;
- LMU shall maintain a Community Relations Policy that addresses student behavior in the community, which shall be updated annually;
- LMU's Department of Public Safety shall respond to oncampus incidents regarding excessive noise and student violations shall be sanctioned as provided in LMU's Student Conduct Code; and
- 10. LMU shall maintain an information and referral service to assist students with off-Campus housing. This service shall also advise all prospective residents of off-Campus housing regarding the university's Community Relations Policy and rules on student behavior in the community.

SECTION 7: SIGNAGE

Section 7.1 GENERAL PROVISIONS

A. Purpose

1. Purpose. The signage program in this Specific Plan is intended to create clear, attractive signage that enhance the Specific Plan area while complementing and protecting the character of the surrounding residential areas by limiting visual clutter through regulation of the location,

height, types, and illumination of Signs allowed within the Specific Plan Area.

- 2. Purpose. The signage program in this Specific Plan is also intended to provide LMU the opportunity to acknowledge those individuals, families, trusts, or other entities (commercial or non-commercial), which underwrite or directly support LMU academic and research programs and/or athletic activities, organizations, facilities, or intramural, club, or varsity athletic teams.
- 3. The signage program in this Specific Plan is not intended to supersede the requirements of the City's Building Code regulations for signs or the Fire Department's regulations for signs. All signs must comply with the applicable Building Code and Fire Code requirements.

B. Prohibited Signs.

- 1. Supergraphic Signs shall be prohibited on Campus.
- 2. Off-Site Signs shall be prohibited on Campus.
- 3. Flashing, moving, and animated Signs shall be prohibited on Campus, except as permitted below in connection with Animated Scoreboard Signs and Electronic Message Signs.
- Signs shall be prohibited in the Buffer/Open Space Planning Areas on Campus, except for Campus Identification Signs at entry ways at Loyola Boulevard and LMU Drive and Informational Signs.
- 5. All Signs not expressly permitted by this Specific Plan or the Municipal Code shall be prohibited. The Department of Building and Safety shall not issue a permit for a Sign, a Sign structure, or Sign illumination, unless the Sign complies with the requirements of this Specific Plan or the Municipal Code.

C. Permitted Signs.

- 1. Animated Scoreboard Signs.
 - a) Animated Scoreboard Signs shall only be permitted within the Athletic Planning Areas and shall be located a minimum of 40 feet from Campus perimeter property lines.

- b) Animated Scoreboard Signs shall be oriented to face spectator areas and shall be predominantly viewable from on Campus locations. An Animated Scoreboard Sign shall not be prohibited if it is merely incidentally viewable from the R1 residential properties within 1,000 feet of the Animated Scoreboard Sign. Landscaping, however, shall be used as feasible to screen any incidental views of Animated Scoreboard Signs from the R1 residential properties within 1,000 feet of the Animated Scoreboard Sign.
- c) Animated Scoreboard Signs shall be limited to 750 square feet in size and shall be limited to one per athletic field or facility.
- d) Animated Scoreboard Signs may include University Donor Signs provided that static University Donor Signs on Animated Scoreboard Signs shall not exceed 40% of the Animated Scoreboard Sign area.
- e) Animated Scoreboard Signs shall be arranged and illuminated in a manner that shall not produce a light intensity of greater than two foot candles above ambient lighting as measured at the property line of the nearest residentially zoned property outside of the perimeter of the Campus.
- f) The illumination of all Animated Scoreboard Signs shall be turned off by 10:00 p.m., except in the case of overtime or extra inning events.
- g) Prior to the installation of a new Animated Scoreboard Sign, LMU shall submit to the Department of Planning a landscape plan for the Campus perimeter from Loyola Boulevard to Regis Street. The landscape plan shall provide for the installation of additional trees and shrubs to further reduce the visibility of the Animated Scoreboard Sign from the R1 residential properties from the area along 80th Street from Loyola Boulevard to Regis Street. The landscaping shall be installed prior to the installation of the Animated Scoreboard Sign.
- Building / Place Identification Signs. Signs identifying buildings, facilities, structures and other places on Campus, which may include University Donor Signs, shall be permitted on Campus. Signs identifying businesses on Campus, such as food and coffee

establishments, bookstores, gift and sundries stores, shall be permitted.

- a) Building/Place Identification Signs shall be limited to a maximum 36" high letters or a Monument Sign not to exceed eight feet above grade.
- b) Building / Place Identification Signs may include University Donor Signs, however if a Building/Place Identification Sign is more than merely incidentally viewable from the R1 residential properties within 500 feet of the Building/Place Identification Sign, it shall not display any University Donor logos.
- c) Building/Place Identification Signs may include Banner Signs that are permanent subject to Building and Safety and Fire Code compliance. Typical Building/Place Identification Banner Signs are set forth in Exhibit 6A. Such Banner Signs shall not be subject to the Temporary Sign provisions of this Section. Individual Building/Place Identification Banner Signs attached to building walls shall be limited to a Sign area of no more than 900 square feet.
- 3. Campus Identification Signs. Campus Identification Signs shall be permitted on Campus including at the entry locations at LMU Drive and Loyola Boulevard.
 - a) Campus Identification Signs may include University Donor Signs, however if a Campus Identification Sign is more than merely incidentally viewable from the R1 residential properties within 500 feet from the Campus Identification Sign, it shall not display University Donor logos.
 - b) Sign letters shall be limited to 36" in height, except that Campus Identification Signs located at the LMU Drive and Loyola Boulevard Campus entries shall have Sign letters limited to 72" in height.
 - c) The number of Campus Identification Signs located at the LMU Drive and Loyola Boulevard Campus entries shall be limited to five. These five signs shall not exceed a total area of 2,000 square feet in the aggregate.
 - d) Campus Identification Signs may include Banner Signs that are permanent subject to Building and Safety and Fire Code compliance. Typical Campus Identification Banner Signs are set forth in Exhibit 6B. Such Banner Signs shall not be subject to the Temporary Sign provisions of this

Section. Individual Campus Identification Banner Signs attached to building walls shall be limited to a Sign area of no more than 900 square feet.

- 4. The LMU Bluff Sign shall be permitted on Campus.
- 5. Temporary Signs. Temporary Signs which are not permanently affixed to the ground or a building, including temporary Banner Signs used to publicize Campus events, shall be permitted on Campus.
 - a) Each Temporary Sign that is more than merely incidentally viewable from the R1 residential properties within 500 feet from the Temporary Sign shall be permitted for no longer than 30 days. All other Temporary Signs shall be permitted for 60 days.
 - b) Individual Temporary Signs shall not exceed 300 square feet in sign area.
 - c) No more than three Temporary Signs may be placed on any building face at any given time.
 - d) Only a total of 1,000 square feet of Temporary Signs may be more than merely incidentally viewable from R1 residential properties within 500 feet of the Temporary Sign.
 - e) Temporary Signs may include Building/Place Identification Signs, Campus Identification Signs, Information Signs, and University Donor Signs, provided if the Temporary Sign is more than merely incidentally viewable from R1 residential properties within 500 feet of the Temporary Sign, the Sign may not include any University Donor logos (but may include sponsors for charitable events [e.g., AIDS Walk sponsors; Cancer 10k run sponsors]).
- 6. Construction Signs. Signs used to screen construction areas on Campus may be maintained on Campus for the length of the construction project. Construction Signs may include LMU logos and University Donors provided that if the Construction Sign is more than merely incidentally viewable from the R1 residential properties within 500 feet of the Construction Sign it shall not include a University Donor logo.
- Athletic Field Sponsorship Signs. Signs used to acknowledge University Donors and LMU teams shall be permitted on athletic fields. Individual Athletic Field Sponsorship Signs shall be limited to 100 square feet. Athletic Field Sponsorship Signs that are more

than merely incidentally viewable from the R1 residential properties within 500 feet of the Sign shall not extend beyond 5 feet above grade and may not include University Donor logos. A typical Athletic Field Sponsorship Sign is set forth in Exhibit 6C.

- 8. Information Signs. Information Signs, such as directional and wayfinding signs, shall be permitted on Campus. There shall be no limit on the number of Information Signs on Campus. Individual Information Signs shall not exceed 25 square feet in Sign area if the Informational Sign is more than merely incidentally viewable from the R1 residential properties within 500 feet of the Informational Sign. If not more than merely incidentally viewable from the R1 residential properties within 500 feet of the Informational Sign. If not more than merely incidentally viewable from the R1 residential properties within 500 feet of the Informational Sign, individual Information Signs shall be limited to 100 square feet in Sign area.
- 9. Electronic Message Sign. Electronic Message Signs shall be permitted on Campus provided that such Electronic Message Sign shall not be located within 40 feet of the Campus boundary.
 - a) Electronic Message Signs shall be limited to 200 square feet in size.
 - b) There shall be no more than one Electronic Message Sign per building face.
 - No more than 500 square feet of Electronic Message Signs may be visible from the R1 residential properties within 1,000 feet of the Electronic Message Signs.
 - d) There shall be no more than ten Electronic Messages Signs on Campus which are greater than 100 square feet in size.
 - e) Any Electronic Message Sign that is incidentally visible from the R1 residential properties shall be screened by landscaping, a wall or other structure or architectural feature to the extent reasonably feasible to minimize spillover light into the surrounding residential neighborhood.
 - f) Electronic Message Signs shall be arranged and illuminated in a manner that shall not produce a light intensity of greater than two foot candles above ambient lighting as measured at the property line of the nearest residentially zoned property.
 - g) The illumination of any Electronic Message Sign that is visible from any off-Campus residentially zoned property shall be turned off by 10:00 p.m.

- h) An Electronic Message Sign may include University Donor Signs provided that it complies with the regulations for University Donor Signs.
- 10. University Donor Signs. Signs recognizing University Donors, including plaques and building signs, shall be permitted on Campus, pursuant to the regulations set forth above. University Donor Signs may not include University Donor logos if the University Donor Sign is more than merely incidentally viewable from the R1 residential properties within 500 feet of the Sign (unless a greater distance is provided for in this Section).

D. General Provisions.

- 1. Definitions. Sign definitions for this Specific Plan are as set forth in Exhibit 6.
- 2. Existing Signs. Repairs, modifications, maintenance and replacements of existing signs shall be permitted pursuant to the provisions of the Municipal Code and this Specific Plan. Signs and/or Sign support structures that exist on the effective date of this Specific Plan shall be permitted to continue to exist and shall not be required to comply with this Specific Plan, but shall comply with the Municipal Code related to the construction, installation, and maintenance of Sign structures. Existing Signs that do not conform to the provisions of this Specific Plan may be replaced only as permitted pursuant to the nonconforming use provisions in Section 91.6216 of the Los Angeles Municipal Code or by a replacement sign that conforms to all provisions of this Specific Plan.
- 3. Los Angeles Municipal Code regulations regarding Sign area, number of Signs, illumination, and other applicable limitations shall apply to all Signs erected on Campus, except where such regulations and limitations are expressly superseded by a different limitation imposed by the provisions of this Specific Plan.
- 4. Height. (i) All new Signs on Campus shall be regulated by the heights established by Subsection 4.1 of this Specific Plan as shown on Exhibit 3; (ii) If attached to a building, Signs shall not extend above the building's parapet and/or cornice.
- 5. Area.
 - a) The area of all Building/Place Identification Signs on Campus that are attached to a building wall shall not exceed 2 square feet per linear foot of public street frontage adjacent to the Campus and roadways internal to the Campus.

- b) The combined area of all Building/Place Identification Signs, Campus Identification Signs, and Electronic Message Signs on Campus that are attached to a building wall shall not exceed 4 square feet per linear foot of public street frontage adjacent to the Campus and roadways internal to the Campus.
- c) The aggregate area of Building/Place Identification Signs, Campus Identifications Signs and Electronic Message Signs that are attached to any given building facade shall not exceed 20% of the building facade.
- 6. For purposes of the Sign regulations in this Specific Plan, premises shall include all properties within the boundaries of Campus.
- 7. Indoor Signs. There shall be no limitations on signs within buildings or structures.
- 8. Signs may be illuminated, provided that they shall not produce a light intensity of greater than two foot candles above ambient lighting as measured at the property line of the nearest R1 residentially zoned property outside of the perimeter of the Campus.
- 9. Limitation on Signs Facing R1 Residential Properties. Signs shall be prohibited from facing those R1 residential properties within 100 feet of the Sign, if the Sign is located more than 25 feet above grade.

E. Sign Approval Procedure

- 1. Planning Department Sign-Off. The Department of Building and Safety shall issue no permit for any Signs until the Director of Planning has certified in writing on the permit application that the proposed Sign is in conformance with all applicable provisions of this Specific Plan.
- 2. For Signs requiring Planning Department sign-off, LMU shall submit three copies of the Sign drawn to scale, indicating the Sign area, height, placement, lettering styles, materials, colors, and lighting methods for the proposed Sign(s), and also a map or other suitable illustration showing graphically the proposed Sign location, and any other materials as necessary, to enable the Director to determine conformance with the regulations of this Specific Plan. In addition, LMU may provide a Campus signage program for review and approval by the Planning Department. The Campus signage program shall provide substantially the same information as provided for in this subsection.

SECTION 8: IMPLEMENTATION

8.1 ADMINISTRATIVE CLEARANCE

- A. Administrative Clearance. Except for the exceptions set forth in Subsection 8.1.C below, and the exempt uses set forth in Subsection 8.1.D below,-no grading permit, foundation permit, building permit, use of land permit, or permit for a change of use shall be issued for a Project located within the Specific Plan Area unless an Administrative Clearance has been issued pursuant to the procedures set forth in this Section. The provisions of this Section shall replace the Project Permit Compliance provisions of Section 11.5.7.C of the Municipal Code.
 - 1. Director's Authority. The Director shall have the authority to review each Project for compliance with all applicable provisions of this Specific Plan, and if in compliance, to grant an Administrative Clearance.
 - 2. Procedures. The Director shall establish the appropriate procedures, forms and fees required for the administrative clearance application. LMU shall circulate copies of its Administrative Clearance application to the Council Office representing the Campus's District and the Westchester/Playa Neighborhood Council when it submits the application to the Director
 - 3. Ministerial Review. The Administrative Clearance shall be a ministerial review of the applicable provisions of this Specific Plan and determination of whether a Project complies with the applicable provisions of the Specific Plan, including the Mitigation Monitoring and Reporting Program in Appendix A.
- B. Modification of Administrative Clearance. Once an Administrative Clearance has been issued, any subsequent proposed material modification to the Project shall require a review by the Director, who shall grant approval of the modification if he or she finds the modification complies with the Specific Plan.
 - 1. Procedures. To modify an approved Project, LMU shall file an application pursuant to the application procedure set forth in Subsection 8.1.A above. The application shall include a description of the proposed modification.
- C. Exceptions. Subsections 8.1.A and 8.1.B shall not apply to:
- Demolition, interior remodeling of a building, normal building repair and maintenance, the change of use of a building or land, or the relocation of existing uses unless the interior remodeling, change of use, or relocation: (a) increases the footprint of a building by more than 10%; or (b) increases the Floor Area of a building by more than 10% and exceeds 50,000 additional square feet of Floor Area.
- 2. Exterior remodeling of a building, unless the aggregate value of the work, in any one 24-month period, is greater than 50 percent of the building or structure's replacement value before the alterations or additions, as determined by the Department of Building and Safety;
- 3. Any demolition or construction for which a permit is required in order to comply with an order issued by the Department of Building and Safety to repair, replace, or demolish an unsafe or substandard condition.
- D. Exemptions.
 - 1. Temporary Structures. Temporary structures of one year duration or less as determined by the Director shall be exempt from the requirements of this Specific Plan.

SECTION 9: OTHER SPECIFIC PLAN PROCEDURES

SECTION 9.1 ISSUANCE OF PERMITS

For all Projects within the Specific Plan for which an Administrative Clearance is required, the Department of Building and Safety shall not issue a grading, building or change of use permit unless an Administrative Clearance has been obtained pursuant to the applicable procedures. The Administrative Clearance shall be a sign off by the Department of City Planning on a Clearance Summary Worksheet for any relevant application for a building permit from the Department of Building and Safety, and shall be appealable through the related building permit via the procedures in Los Angeles Municipal Code §12.26K.

SECTION 10: INTERPRETATION

Whenever any ambiguity or uncertainty related to the application of this Specific Plan exists so that it is difficult to determine the precise application of these provisions, the Director shall, upon application by the property owner, issue binding interpretations of the Specific Plan requirements consistent with the purpose and intent of this Specific Plan. Ambiguity between the Specific Plan and Municipal Code shall be

Loyola Marymount University Specific Plan

interpreted in favor of and consistent with the goals and purposes of this Specific Plan. Even if the Municipal Code is more recent or more restrictive, this Specific Plan controls.

SECTION 11: SEVERABILITY

If any provision of this Specific Plan or the application thereof to any person or circumstance is held to be unconstitutional or otherwise invalid by any court of competent jurisdiction, the invalidity shall not affect other Specific Plan provisions, clauses, or applications, which can be implemented without the invalid provision, clause, or application, and to this end the provisions and clauses of this Specific Plan are declared to be severable.

Exhibit B:

Mitigation Monitoring Program

Attached.

Appendix A

LOYOLA MARYMOUNT UNIVERSITY MASTER PLAN PROJECT MITIGATION MONITORING AND REPORTING PROGRAM

A. INTRODUCTION

Section 21081.6 of the Public Resources Code requires a Lead Agency to adopt a "reporting or monitoring program for changes to the project or conditions of project approval, adopted in order to mitigate or avoid significant effects on the environment." In addition, Section 15097(a) of the *California Environmental Quality Act* (CEQA) *Guidelines* requires that:

[I]n order to ensure that the mitigation measures and project revisions identified in the EIR or negative declaration are implemented, the public agency shall adopt a program for monitoring or reporting on the revisions which it has required in the project and measures it has imposed to mitigate or avoid significant environmental effects. A public agency may delegate reporting or monitoring responsibilities to another public agency or to a private entity which accepts the delegation; however, until mitigation measures have been completed the lead agency remains responsible for ensuring that implementation of the mitigation measures occurs in accordance with the program.

The City of Los Angeles Department of City Planning has been designated as the Lead Agency for the Proposed Loyola Marymount University (LMU) Master Plan Project.

A Draft EIR was prepared to address the potential environmental impacts of the Proposed Project. Where appropriate, the Draft EIR identified project design features or recommended mitigation measures to avoid or to mitigate potential impacts to the environment to a level that would be less than significant. This Mitigation Monitoring and Reporting Program (MMRP) is designed to monitor implementation of the project design features and mitigation measures required for the Proposed Project.

The project design features and mitigation measures identified in the Draft EIR are categorized by environmental impact section. Following each project design features and mitigation measure is identification of the following:

- Enforcement Agency: The agency with the power to enforce the project design feature and mitigation measure.
- Monitoring Agency: The agency to which reports involving feasibility, compliance, implementation and development are made.
- Monitoring Phase: The phase of the project during which the project design feature and mitigation measure shall be monitored, such as: Pre-Construction, including the design phase; Construction; Prior to Issuance of a Building Permit; Prior to Issuance of a Certificate of Occupancy; Occupancy (post-construction).

- Monitoring Frequency: The frequency with which the project design feature and mitigation measure is to be implemented (i.e., once at a specific point during Pre-Construction, Construction, etc.) or ongoing throughout a phase of the Project.
- Action Indicating Compliance with Mitigation Measure(s): The performance standard indicating that compliance with the project design feature and mitigation measure has been satisfactorily achieved.

The MMRP for the LMU Master Plan Project will be in place throughout all phases of the Proposed Project. The Project applicant will be responsible for implementing all project design features and mitigation measures unless otherwise noted. The applicant shall also be obligated to provide certification, as identified below, to the appropriate monitoring agency and the appropriate enforcement agency that compliance with the required mitigation measure has been implemented. The City's existing planning, engineering, review, and inspection processes will be used as the basic foundation for the MMRP procedures and will also serve to provide the documentation for the reporting program.

The substance and timing of each certification report that is submitted to Department of City Planning shall be at the discretion of City Planning. Generally, each report will be submitted to Department of City Planning in a timely manner following completion/implementation of the applicable mitigation measure and shall include sufficient information to reasonably determine whether the intent of the measure has been satisfied. The Department of City Planning, in conjunction with the Project applicant, shall assure that Project construction occurs in accordance with the MMRP. The South Coast Air Quality Management District (SCAQMD) shall be responsible for the implementation of corrective actions relative to violations of SCAQMD rules associated with mitigation. Departments listed below are all departments of the City of Los Angeles, unless otherwise noted.

B. PROJECT DESIGN FEATURES AND MITIGATION MEASURES FOR ENVIRONMENTAL IMPACTS

1. Aesthetics and Views

Project Design Features

PDF-AES-1 The Proposed Project shall site buildings in compliance with the requirements of the proposed LMU Specific Plan, to retain the distinctive axial and cross-axial alignment of the Alumni Mall and Sunken Garden.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building during Administrative Clearance

City of Los Angeles ENV-2008-1342-EIR Action Indicating Compliance with Mitigation Measure(s): Administrative Clearance approval

PDF-AES-2 The Proposed Project shall comply with the height and setback restrictions established by the proposed LMU Specific Plan.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building during Administrative ClearanceAction Indicating Compliance with Mitigation Measure(s):Administrative Clearanceapproval

PDF-AES-3 The LMU campus shall provide, at a minimum, the acreage of open space and outdoor athletic facilities required by the proposed LMU Specific Plan.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building during Administrative ClearanceAction Indicating Compliance with Mitigation Measure(s):Administrative Clearanceapproval

Mitigation Measures

MM-AES-1 The Project applicant shall implement screening measures, which may include, but are not limited to, temporary visual barriers such as fencing around construction areas in order to limit views of the construction site(s).

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction In directing Compliance with Mitigation Measure(c):Field inspection circle

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

2. Shade and Shadow

Project Design Features

PDF-SH-1 The Proposed Project shall comply with the height and setback restrictions established by the proposed LMU Specific Plan.

Enforcement Agency:	City of Los Angeles, Department of City Planning
Monitoring Agency:	City of Los Angeles, Department of Building and Safety
Monitoring Phase:	Pre-Construction

Monitoring Frequency: Once for each building during Administrative Clearance Action Indicating Compliance with Mitigation Measure(s): Administrative Clearance approval

Mitigation Measures

No mitigation measures required.

3. Light and Glare

Project Design Features

No Project Design Features proposed.

Mitigation Measures

MM-LIGHT-1 The use of nighttime lighting during Project construction shall be limited to only those features on the construction site requiring illumination.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

MM-LIGHT-2 All security lights shall be properly shielded and projected downwards during construction such that light is directed only onto the work site.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Construction

Monitoring Frequency: Periodic field inspections

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

MM-LIGHT-3 Lighting fixtures and visors shall be adjusted upon installation to reduce spillover onto adjacent residential properties, while still maintaining adequate lighting to allow safe use of outdoor athletic facilities. Additionally, vegetation and other screening or filtering devices shall be maintained or supplemented at the edges of lit fields or at the campus perimeter at all times, such that light spillover shall not be permitted at any time to exceed an intensity of 2 foot-candles on residential property lines located along W. 78th Street, Fordham Road, W. 80th Street, and McConnell Avenue.

> City of Los Angeles, Department of Building and Safety **Enforcement Agency:** City of Los Angeles, Department of Building and Safety Monitoring Agency: **Monitoring Phase:** Operation Monitoring Frequency: Field inspection during operation Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

MM-LIGHT-4 All outdoor lighting shall be directed downward to illuminate the intended surface (i.e., playing fields, pedestrian pathways and other high-traffic areas such as building entrances and plazas in the campus interior).

Enforcement Agency:	City of Los Angeles, Department of City Planning									
Monitoring Agency:	City of Los Angeles, Department of Building and Safety									
Monitoring Phase:	Prior to issuance of building permit for new buildings,									
	Operation									
Monitoring Frequency	: Once prior to issuance of building permits for new buildings,									

Periodic field inspections during operation

Action Indicating Compliance with Mitigation Measure(s): Issuance of building permits for new buildings; field inspection sign-off

MM-LIGHT-5 All new outdoor lighting shall be equipped with louvers, shields, hoods, or other screening devices.

> Enforcement Agency: City of Los Angeles, Department of City Planning Monitoring Agency: City of Los Angeles, Department of Building and Safety Prior to issuance of building permit for new buildings or **Monitoring Phase:** electrical permit for athletic field lighting Monitoring Frequency: Once prior to issuance of permits for new buildings or athletic field lighting Action Indicating Compliance with Mitigation Measure(s): Issuance of building permits for new buildings; field inspection sign-off for athletic fields

MM-LIGHT-6 The use of field lighting within all outdoor athletic facilities shall be limited to only those hours during which the facilities are being utilized, which shall not surpass 10:00 PM except in the case of overtime or extra innings.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:OperationMonitoring Frequency:Periodic field inspection

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

MM-LIGHT-7 The Applicant shall use exterior building materials and facades which eliminate or minimize highly reflective materials. At the time of plan check review for specific development projects, building materials shall be reviewed to assure that they do not exceed the reflectivity of standard building materials. If the Applicant should desire to use more reflective materials in locations isolated from major thoroughfares, adequate analysis must be presented to the Department of Building and Safety to determine that the building, due to location, would not cause glare impacts on motorists or nearby population.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once during building plan checkAction Indicating Compliance with Mitigation Measure(s):Plan check approval

4. Air Quality

Project Design Features

See Project Design Features for Transportation, Energy, and Water Supply.

Mitigation Measures

MM-AQ-1 General contractors shall implement a fugitive dust control program pursuant to the provisions of SCAQMD Rule 403.

Enforcement Agency:South Coast Air Quality Management DistrictMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction Indicating Compliance with Mitigation Measure(s):Quarterly compliancecertification report submitted by Project contractor

All construction equipment shall be properly tuned and maintained in accordance with MM-AO-2 manufacturer's specifications.

> City of Los Angeles, Department of Building and Safety **Enforcement Agency:** City of Los Angeles, Department of Building and Safety Monitoring Agency: Monitoring Phase: Construction Monitoring Frequency: Periodic field inspections

Action Indicating Compliance with Mitigation Measure(s): Quarterly compliance certification report submitted by Project contractor

MM-AO-3

General contractors shall maintain and operate construction equipment so as to minimize exhaust emissions. During construction, trucks and vehicles in loading and unloading queues shall turn their engines off when not in use to reduce vehicle emissions. Construction emissions shall be phased and scheduled to avoid emissions peaks and discontinued during second-stage smog alerts.

Enforcement Agency: City of Los Angeles, Department of Building and Safety City of Los Angeles, Department of Building and Safety Monitoring Agency: **Monitoring Phase:** Construction

Monitoring Frequency: Periodic field inspections

Action Indicating Compliance with Mitigation Measure(s): compliance Quarterly certification report submitted by project contractor

MM-AQ-4 Electricity rather than temporary diesel- or gasoline-powered generators shall be used at least half of the time.

City of Los Angeles, Department of Building and Safety **Enforcement Agency:** Monitoring Agency: City of Los Angeles, Department of Building and Safety **Monitoring Phase:** Construction Monitoring Frequency: Throughout grading and excavation, construction during field inspection

Action Indicating Compliance with Mitigation Measure(s): Quarterly compliance certification report submitted by project contractor

MM-AQ-5 All construction vehicles shall be prohibited from idling in excess of 5 minutes, both on and off site.

> Enforcement Agency: City of Los Angeles, Department of Building and Safety Monitoring Agency: City of Los Angeles, Department of Building and Safety **Monitoring Phase:** Construction

Monitoring Frequency: Periodic field inspection

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-offs and quarterly compliance certification report submitted by project contractor

MM-AQ-6 The Applicant shall utilize coatings and solvents that are consistent with applicable SCAQMD rules and regulations.

Enforcement Agency:South Coast Air Quality Management DistrictMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-offs
and quarterly compliance certification report submitted by
project contractor

MM-AQ-7 The Applicant shall schedule routine deliveries during off-peak traffic periods to encourage the reduction of trips during the most congested periods.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Grading/Construction

Monitoring Frequency: Throughout grading and construction

Action Indicating Compliance with Mitigation Measure(s): Quarterly compliance certification report submitted by project contractor

MM-AQ-8

- LMU shall require all on-site construction equipment to meet EPA Tier 2 or higher emissions standards according to the following :
 - April 2010 through December 31, 2011: All offroad diesel-powered construction equipment greater than 50 horsepower (hp) shall meet Tier 2 offroad emissions standards. In addition, all construction equipment shall be outfitted with the BACT devices certified by CARB. Any emissions control device used by the contractor shall achieve emissions reductions that are no less than what could be achieved by a Level 2 or Level 3 diesel emissions control strategy for a similarly sized engine as defined by CARB regulations.
 - January 1, 2012 through December 31, 2014: All offroad diesel-powered construction equipment greater than 50 horsepower (hp) shall meet Tier 3 offroad emissions standards. In addition, all construction equipment shall be outfitted with the BACT devices certified by CARB. Any emissions control device used by the contractor shall achieve emissions reductions that are no less than what could be achieved by a Level 3 diesel emissions control strategy for a similarly sized engine as defined by CARB regulations.
 - Post-January 1, 2015: All offroad diesel-powered construction equipment greater than 50 horsepower (hp) shall meet Tier 4 offroad emissions standards. In addition, all construction equipment shall be outfitted with the BACT devices certified by CARB. Any emissions control device used by the contractor shall achieve emissions reductions that are no less than what could be achieved by a Level 3 diesel emissions control strategy for a similarly sized engine as defined by CARB regulations.

A copy of each unit's certified tier specification, BACT determination, and CARB or SCAQMD operating permit shall be provided at the time of mobilization of each applicable unit of equipment.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Construction

Monitoring Frequency: Periodic field inspections

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off and quarterly compliance certification report submitted by project contractor

MM-AQ-9

For equipment not covered by MM-AQ-8 above, the Project Applicant shall evaluate the potential for reducing exhaust emissions from on-road and off-road construction equipment, and implement such measures. Control technologies to be considered may include particulate traps and filters, selective catalytic reduction, oxidation catalysts, air enhancement technologies, and the use of alternatively (non-diesel) fueled engines. Considerations will include commercial availability of appropriate California Air Resources Board verified technologies.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction Indicating Compliance with Mitigation Measure(s):Field-inspection sign-offs
and quarterly compliance certification report submitted by

project contractor

MM-AQ-10 The Applicant shall install shaker plates at construction site exits, to minimize dirt track out and dust generation.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction Indicating Compliance with Mitigation Measure(s):Field-inspection sign-off

MM-AQ-11 The Applicant shall operate street sweepers that comply with SCAQMD Rules 1186 and 1186.1 on roads adjacent to the construction site in a nearly continuous manner so as to minimize dust emissions. Paved parking and staging areas shall be swept daily.

Enforcement Agency:South Coast Air Quality Management DistrictMonitoring Agency:Los Angeles City, Department of Building and Safety

Monitoring Phase: Construction Monitoring Frequency: Periodic field inspections Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-offs

MM-AQ-12

An information sign shall be posted at the entrance to each construction site that identifies the permitted construction hours and provides a telephone number to call and receive information about the construction project or to report complaints regarding excessive fugitive dust generation. Any reasonable complaints shall be rectified within 24 hours of their receipt.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field-inspections

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

MM-AQ-13 LMU shall require the contractor to limit construction activity over unpaved surfaces to

five acres of disturbance per day or less.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field-inspections

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

MM-AQ-14 LMU shall require the contractor to provide temporary traffic controls such as a flag person during all phases of construction to maintain smooth traffic flow.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field-inspectionsAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

MM-AQ-15 LMU shall require the contractor to replace ground cover in disturbed areas as quickly as possible as permitted by the sequence of the Master Plan's project schedule.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field-inspections

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

5. Global Climate Change

Project Design Features

See Project Design Features for Section IV.K, Transportation, Section IV.L.4, Energy, and Section IV.L.1, Water Supply.

Mitigation Measures

No mitigation measures required.

6. Biological Resources

Project Design Features

PDF-BIO-1

Development of the Proposed Project shall comply with the land use regulations for open space areas established by the proposed LMU Specific Plan.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building during Administrative ClearanceAction Indicating Compliance with Mitigation Measure(s):Administrative Clearanceapproval

Mitigation Measures

MM-BIO-1

Prior to any earthmoving activities during the breeding and nesting season from March 1 through August 31, the Applicant shall have a survey conducted by a qualified biologist to determine if active nests for breeding birds are present within the area of potential influence of the species. This area of influences shall include the nest site as well as an appropriate buffer determined by the biologist based on field observations and the biology of the species. This survey shall be conducted within three (3) days before the clearing/grubbing. If nesting birds protected under the Migratory Bird Treaty Act or California Fish and Game Code are found, the breeding/nesting area(s) shall be protected according to the biologist's recommendation that include, but are not limited to, suitable buffer area around the nest, which shall not be disturbed until the young have fledged.

Enforcement Agency: California Department of Fish & Game Monitoring Agency: City of Los Angeles, Department of City Planning

MMP-11

Monitoring Phase: Construction

Monitoring Frequency: Once prior to each earthmoving activity conducted during the breeding and nesting seasons

Action Indicating Compliance with Mitigation Measure(s): Compliance report from a qualified biologist

MM-BIO-2

Prior to any removal of trees during the months of October through February, the Applicant shall have conducted by a qualified biologist a survey to determine if monarch butterfly clusters are present within the trees to be effected by the removal. Removal of trees occupied by monarch butterfly clusters during the months of October through February shall be prohibited unless it is determined by the City that such removal is necessary by reason of good forestry practice, disease of the tree, or safety considerations. Any such determinations shall be accompanied by a written evaluation of the impacts of the proposed action on the monarch butterfly by a qualified expert.

Major construction activity within 100 feet of any trees occupied by monarch butterfly clusters shall be prohibited between October and February while the monarch butterflies are present unless a qualified expert is present and determines that such construction activities will not disturb the monarch butterfly cluster.

Enforcement Agency:California Department of Fish & GameMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:ConstructionMonitoring Frequency: Once prior to removal of trees between
October and February

Action Indicating Compliance with Mitigation Measure(s): Compliance report from a qualified biologist

MM-BIO-3

For each protected tree intended for removal in implementation of the Project, replacement trees shall be planted in accordance with Section 46.01 of the Los Angeles Municipal Code.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-constructionMonitoring Frequency:Once during building plan checkAction Indicating Compliance with Mitigation Measure(s):Building plan checkapproval

7. Paleontological Resources

Project Design Features

PDF-PALEO-1 Development of the Proposed Project shall comply with the land use regulations for open space areas established by the proposed LMU Specific Plan.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building during Administrative ClearanceAction Indicating Compliance with Mitigation Measure(s):Administrative Clearanceapproval

Mitigation Measures

MM-PALEO-1 Retention of Paleontologist. Prior to the initiation of construction-related earthmoving activities and excavation at depths of 5 feet below the surface of campus, the services of a qualified paleontological consulting firm approved by the City and the Natural History Museum of Los Angeles County Vertebrate Paleontology Section shall be retained and consulted. Using field observations, bore logs, geologic reports, and construction plans, the paleontologist shall determine when and where any monitoring of earthmoving activities will be required.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-constructionMonitoring Frequency:Once prior to ground disturbance for each buildingAction Indicating Compliance with Mitigation Measure(s):Compliance report by a
qualified paleontologist

MM-PALEO-2 Preconstruction Coordination and Environmental Awareness Training. If monitoring is required, the paleontologist or another mitigation program staff member shall coordinate with appropriate construction contractor personnel to provide information regarding applicable requirements concerning the protection of paleontological resources. Contractor personnel, particularly heavy-equipment operators, shall also be briefed on procedures to be followed in the event that fossil remains and a currently unrecorded fossil site are encountered by earthmoving activities, particularly if a paleontological construction monitor is not on site. The briefing shall be presented to new contractor personnel as necessary. Names and telephone numbers of the monitor and other appropriate mitigation program personnel shall be provided to appropriate contractor personnel.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once prior to commencement of construction and subsequently
for new contractor personnel as neededAction Indicating Compliance with Mitigation Measure(s):Compliance certification
report from Applicant or Representative with evidence of
training provided by paleontologist or another mitigation
program staff member to appropriate construction contractor
personnel

MM-PALEO-3 Paleontological Monitoring and Fossil Specimen and Sample Recovery. When required, monitoring shall consist of visually inspecting debris piles and freshly exposed strata to allow for the discovery and recovery of larger fossil remains, and periodically dry test screening rock, sediment, and debris to allow for the discovery and recovery of smaller fossil remains. As soon as practicable, the monitor shall recover all larger vertebrate fossil remains, a representative sample of invertebrate or plant fossil specimens, or any fossiliferous rock or sediment sample that can be recovered easily. If recovery of a large or unusually productive fossil occurrence is warranted, earthmoving activities shall be diverted temporarily around the fossil site and a recovery crew shall be mobilized as necessary to remove the occurrence as quickly as possible. If not on site when a fossil occurrence is uncovered by such activities, the activities shall be diverted temporarily around the fossil site and the monitor called to the site to evaluate and, if warranted, recover the occurrence. If the paleontologist or monitor determines that the fossil site is too unproductive or the fossil remains not worthy of recovery by the monitor, no further action will be taken to preserve the fossil site or remains, and earthmoving activities shall be allowed to proceed through the site immediately. The location and proper geologic context of any recovered fossil occurrence or rock or sediment sample shall be documented.

Any recovered rock or sediment sample from the Palos Verdes Sand or non-marine terrace cover shall be processed to allow for the recovery of smaller fossil remains that normally are too small to be observed by the monitor. Pursuant to Society of Vertebrate Paleontology (1995) standard measures, no more than 6,000 pounds (12,000 pounds total) of the rock or sediment shall be processed from either the Palos Verdes Sand or terrace cover.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Construction

Monitoring Frequency: Periodically as needed

Action Indicating Compliance with Mitigation Measure(s): Compliance report by a qualified paleontologist, per MM-PALEO-5

MM-PALEO-4 Final Laboratory Tasks. All fossil specimens recovered from the Project area as a result of mitigation, including those recovered as the result of processing rock or sediment samples, will be treated (i.e., prepared, identified, curated, catalogued) in accordance with designated museum repository requirements. Rock or sediment samples will be submitted to commercial laboratories for microfossil, pollen, radiometric dating, or other analysis, as appropriate.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:ConstructionMonitoring Frequency:Periodically as neededAction Indicating Compliance with Mitigation Measure(s):Compliance report by a

qualified paleontologist, per MM-PALEO-5

MM-PALEO-5 Reporting. The monitor shall maintain daily monitoring logs that include the particular tasks accomplished, the earthmoving activity monitored, the location where monitoring was conducted, the rock unit(s) encountered, the fossil specimens recovered, and associated specimen data and corresponding geologic and geographic site data. A final technical report of results and findings shall be prepared by the paleontologist in accordance with any City requirement and archived at the museum repository.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:ConstructionMonitoring Frequency:Periodically as needed

Action Indicating Compliance with Mitigation Measure(s): Final technical report of

results and findings by monitor/paleontologist

8. Archaeological Resources

Project Design Features

PDF-ARCH-1 The Proposed Project shall be consistent with the open space land use regulations established by the proposed LMU Specific Plan.

Enforcement Agency: City of Los Angeles, Department of City Planning

MMP-15

 Monitoring Agency:
 City of Los Angeles, Department of City Planning

 Monitoring Phase:
 Pre-Construction

 Monitoring Frequency:
 Once for each building during Administrative Clearance

 Action Indicating Compliance with Mitigation Measure(s):
 Administrative Clearance

 approval
 Prevolution

Mitigation Measures

MM-ARCH-1: Archaeological resources shall be avoided, or unavoidable disturbance shall be mitigated through data recovery, documentation, analysis, and curation. Archaeological treatment plans shall be developed and implemented, as applicable. All materials and records resulting from implementation of the archaeological treatment plans shall be curated in accordance with 36 Code of Federal Regulations, Part 79 (Curation of Federally Owned and Administered Archaeological Collections).

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:ConstructionMonitoring Frequency:Throughout grading and excavation for each buildingAction Indicating Compliance with Mitigation Measure(s):Compliance report by
archaeological monitor, per MM-ARCH-12

MM-ARCH-2: Prior to starting ground-disturbing activities such as construction work on campus, LMU shall retain a Project archaeologist who meets the Secretary of the Interior's guidelines and is listed in the Register of Professional Archaeologists. In addition, a Native American member of the Gabrieleno/Tongva Tribal community shall be retained under contract as a monitor.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-construction

Monitoring Frequency: Once prior to construction for each building

Action Indicating Compliance with Mitigation Measure(s): Retention of archaeologist and Native American monitor

MM-ARCH-3: Before beginning the planned ground-disturbing activities (such as material grading and excavation activities), LMU shall consult with the archaeologist to determine if any potential exists as a result of the planned ground-disturbing activities for disturbance or damage to archaeological resources. The Proposed Project archaeologist shall conduct a preliminary archaeological evaluation (which may include subsurface evaluation) to determine if there are archaeological resources present. If none are determined to be present within the area of planned ground-disturbing activity, then the archaeologist

shall determine there is no potential for disturbance or damage to archaeological resources and the area may be cleared for construction work without the need for further archaeological work.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-construction

Monitoring Frequency: Once prior to ground disturbance for each building

Action Indicating Compliance with Mitigation Measure(s): Memorandum from archaeologist documenting determination whether monitoring is required or not

MM-ARCH-4: If the archaeologist determines there is potential for damage to archaeological resources due to planned ground-disturbing activities, all ground-disturbing activities shall be monitored by the Proposed Project archaeologist and a Native American member of the Gabrieleno/Tongva Tribal community and mitigation for any potential adverse effects to archaeological resources from construction, as identified in mitigation measures MM-ARCH-5 through MM-ARCH-12, shall be conducted.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Construction

Monitoring Frequency: Throughout grading and excavation

Action Indicating Compliance with Mitigation Measure(s): Compliance report by archaeological monitor, per MM-ARCH-12

MM-ARCH-5: If based on a preliminary archaeological evaluation the archaeologist determines there are no archaeological resources present, but archaeological resources are encountered, work shall halt and LMU shall consult again with the archaeologist to determine if any potential exists as a result of the planned ground-disturbing activities for disturbance or damage to archaeological resources (see mitigation measure MM-ARCH-3).

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:ConstructionMonitoring Frequency:Throughout grading and excavationAction Indicating Compliance with Mitigation Measure(s):Compliance report by
archaeological monitor, per MM-ARCH-12

MM-ARCH-6: If archaeological discoveries are identified during monitoring of ground-disturbing activity, the archaeologist may order the temporary diversion of work outside a 200-foot

radius around the discovery until the archaeologist has evaluated the nature and significance of the find.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:ConstructionMonitoring Frequency:Throughout grading and excavation

Action Indicating Compliance with Mitigation Measure(s): Compliance report by archaeological monitor, per MM-ARCH-12

MM-ARCH-7: If potential human remains are encountered during ground-disturbing activities, all work shall halt, and the Los Angeles County Coroner's Office shall be notified, as prescribed in Public Resources Code Section 5097.98 and Health and Safety Code Section 7050.5, and as required by the *State CEQA Guidelines* (Section 15126.4(b)(3) of the California Code of Regulations). If the Coroner determines that the remains are of Native American origin, the Coroner shall proceed as directed in Section 15064.5(e) of the State CEQA Guidelines. LMU shall follow all guidelines outlined in Public Resources Code Section 5097.94(k).

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City Planning

Monitoring Phase: Construction

Monitoring Frequency: Throughout grading and excavation

Action Indicating Compliance with Mitigation Measure(s): Compliance report by archaeological monitor, per MM-ARCH-12

MM-ARCH-8: If significant archaeological resources are encountered, a data recovery plan to mitigate potential adverse effects of construction to a less than significant level shall be developed and implemented. This data recovery plan shall include methods for hand-excavation, analysis, and report writing and also shall provide procedures for the curation of any collected material at a facility meeting professional standards.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Construction

Monitoring Frequency: Throughout grading and excavation

Action Indicating Compliance with Mitigation Measure(s): Compliance report by archaeological monitor, per MM-ARCH-12

MM-ARCH-9: After the archaeologist determines that potential impacts to archaeological resources have been mitigated, where necessary, work may resume in the area where the archaeological resources were encountered.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Construction

Monitoring Frequency: Throughout grading and excavation

Action Indicating Compliance with Mitigation Measure(s): Compliance report by archaeological monitor, per MM-ARCH-12

MM-ARCH-10: Any artifacts uncovered shall be recorded and removed for storage at a location to be determined by the archaeologist.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:ConstructionMonitoring Frequency:Throughout grading and excavation

Action Indicating Compliance with Mitigation Measure(s): Compliance report by archaeological monitor, per MM-ARCH-12

MM-ARCH-11: If archaeological resources are encountered outside of presently recorded site boundaries

of CA-LAN-61, CA-LAN-212, and CA-LAN-1018, the site shall be recorded in accordance with requirements of the State Office of Historic Preservation (i.e., using Department of Parks and Recreation [DPR] 523 forms) and evaluated.

Enforcement Agency: City of Los Angeles, Department of City Planning

Monitoring Agency: City of Los Angeles, Department of City Planning

Monitoring Phase: Construction

Monitoring Frequency: Throughout grading and excavation

Action Indicating Compliance with Mitigation Measure(s): Compliance report by archaeological monitor, per MM-ARCH-12

MM-ARCH-12: Draft reports on archaeological findings shall be prepared by the Proposed Project archaeologist for submission to the City of Los Angeles for review. Final versions of these reports shall be submitted to the City of Los Angeles, LMU, and the South Central Coastal Information Center at California State University, Fullerton. The report shall outline the data recovery plan in place for mitigation and shall describe the history of the Proposed Project area, research questions, the field and laboratory methods and results, and how these findings coincide with both the project research questions and the broader context of archaeology in the region. Collected material and project paperwork shall be curated at a facility meeting professional standards.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Post-construction

Monitoring Frequency: Once at conclusion of grading and excavation

Action Indicating Compliance with Mitigation Measure(s): Submittal of reports on archaeological findings by archaeologist to City of Los Angeles, Department of City Planning, LMU, and the South Central Coastal Information Center at California State University, Fullerton

9. Historical Resources

Project Design Features

PDF-HIST-1 Xavier Hall, St. Robert's Hall, Sacred Heart Chapel and the bluff-face letter "L" shall be retained under the Proposed Project. Any renovations to these historic resources shall be made in compliance with the Secretary of the Interior's Standards.

Enforcement Agency: City of Los Angeles, Department of City Planning (Office of Historic Resources)

Monitoring Agency: City of Los Angeles, Department of City Planning (Office of Historic Resources)

Monitoring Phase: Pre-Construction

- Monitoring Frequency: Once during building plan check for renovation of historic resource buildings
- Action Indicating Compliance with Mitigation Measure(s): Building plan check approval

Mitigation Measures

MM-HIST-1 LMU shall prepare documentation of Xavier Hall, St. Robert's Hall, and Sacred Heart Chapel prior to issuance of a construction permit for any work on those buildings. This documentation shall include:

- A brief written construction history in narrative format for each building.
- A site plan showing the location of each building. This site plan shall include a photo key.
- A sketch floor plan for each building.
- Field photographs (35mm) based on Historic American Buildings Survey guidelines. Views shall include contextual views, all exterior elevations, detailed views of significant exterior architectural features, and interior views of significant historical architectural features or spaces (if any).
- Available historic photographs and historic plans.

 Enforcement Agency:
 City of Los Angeles, Department of City Planning (Office of Historic Resources)

 Monitoring Agency:
 City of Los Angeles, Department of City Planning (Office of Historic Resources)

 Monitoring Phase:
 Pre-Construction

Monitoring Frequency: Once during building plan check for renovation of historic resource buildings

Action Indicating Compliance with Mitigation Measure(s): Building plan check approval

- MM-HIST-2 Renovation and rehabilitation of Xavier Hall, St. Robert's Hall, and Sacred Heart Chapel shall conform to the Secretary of the Interior's Standards
 - Enforcement Agency: City of Los Angeles, Department of City Planning (Office of Historic Resources)
 - Monitoring Agency: City of Los Angeles, Department of City Planning (Office of Historic Resources)
 - Monitoring Phase: Pre-Construction
 - Monitoring Frequency: Once during building plan check for renovation of historic resource buildings

Action Indicating Compliance with Mitigation Measure(s): Building plan check approval

MM-HIST-3 Prior to issuance of a permit for earth excavation or earth moving activities that could impact Xavier Hall, St. Robert's Hall, or Sacred Heart Chapel, LMU shall create a shoring plan to ensure the protection of Xavier Hall, St. Robert's Hall, and Sacred Heart Chapel.

Enforcement Agency:City of Los Angeles, Department of City Planning (Office of
Historic Resources), Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once for each building during building plan check

Action Indicating Compliance with Mitigation Measure(s): Building plan check approval

MM-HIST-4 Prior to the issuance of any permit for the demolition or exterior or structural modification of the Edward T. Foley Center, LMU shall undertake additional historic resources assessment in coordination with the Department of City Planning, Office of Historic Resources, of the subject building's eligibility for the National Register of Historic Places, California Register of Historic Resources, and City Historic-Cultural Monument status. If the building is identified as a historical resource, written approval shall be obtained from the Office of Historic Resources that such work conforms to the Secretary of the Interior's Standards for the Treatment of Historic Properties.

Enforcement Agency:City of Los Angeles, Department of City Planning (Office of
Historic Resources), Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-constructionMonitoring Frequency:Once during building plan check

Action Indicating Compliance with Mitigation Measure(s): Building plan check approval

10. Geology

Project Design Features

No Project Design Features are proposed.

Mitigation Measures

Geological Hazards

MM-GEO-1

Project-level (i.e., building-specific) geotechnical investigations shall be required prior to finalizing grading and construction plans for individual Proposed Project buildings and campus improvements.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building prior to issuance of building permits

Action Indicating Compliance with Mitigation Measure(s): Submittal of geotechnical reports for individual Proposed Project buildings and campus improvements.

MM-GEO-2 Individual buildings and improvements shall be designed and constructed in accordance with the requirements outlined in the most current edition of the California Building Code and the Los Angeles Uniform Building Code, as well as all applicable provisions of Chapter IX, Division 70 of the Los Angeles Municipal Code, which addresses grading, excavation, and fill, Department of the State Architect requirements, and federal building code requirements.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building prior to issuance of building permitsAction Indicating Compliance with Mitigation Measure(s):Building plan checkapproval

MM-GEO-3

Prior to issuance of a grading permit for an individual building or improvement, expansion testing shall be performed in accordance with UBC Standard 29-2 and ASTM Standard D4829 to determine the expansion potential of any import soils. Any required import fill and at least the upper 2 feet of fill beneath floor slabs and beneath other concrete slabs and walks shall consist of relatively non-expansive soils with an Expansion Index of less than 35.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once for each building prior to issuance of grading permit Action Indicating Compliance with Mitigation Measure(s): Issuance of grading permit MM-GEO-4 Prior to issuance of a grading permit for an individual building or improvement, corrosion testing shall be performed and proper corrosion protection shall be implemented where required in accordance with the Los Angeles Uniform Building Code, including all applicable provisions of Chapter IX, Division 70 of the Los Angeles Municipal Code, which addresses grading, excavations and fills.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of grading permitAction Indicating Compliance with Mitigation Measure(s):Issuance of grading permit

MM-GEO-5 Slope stability evaluations shall be performed prior to issuance of a grading permit for buildings and improvements adjacent to bluff slopes. Slope stability evaluations shall be performed along critical cross sections of the slope adjacent to each area of potential development during the design-level geotechnical studies. The design minimum factors of safety under static and pseudostatic loading conditions shall be taken as 1.5 and 1.1, respectively, following accepted geotechnical practices and agency guidelines.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once for each building prior to issuance of grading permit Action Indicating Compliance with Mitigation Measure(s): Issuance of grading permit

Sedimentation and Erosion

MM-GEO-6

Project-level hydrology plans shall be required prior to finalizing grading and construction plans for individual Proposed Project buildings and campus improvements. Hydrology plans shall be designed in conformance with current local, state, and federal regulatory requirements.

Enforcement Agency:City of Los Angeles, Department of Public WorksMonitoring Agency:City of Los Angeles, Department of Public WorksMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of grading permitAction Indicating Compliance with Mitigation Measure(s):Issuance of grading permit

MM-GEO-7 Prior to the start of soil-disturbing activities at the site, a Notice of Intent and Storm Water Pollution Prevention Plan shall be prepared in accordance with, and in order to partially fulfill, the California State Water Resources Control Board Order No. 99-08-DWQ, National Pollution Discharge Elimination System General Permit No. CAS000002

City of Los Angeles ENV-2008-1342-EIR MMP-24

(General Construction Permit) and Chapter 6 Article 4.4, Stormwater and Urban Runoff Pollution Control from the Los Angeles Municipal Code. The Storm Water Pollution Prevention Plan shall meet the applicable provisions of Sections 301 and 402 of the California Water Act and Chapter 6 Article 4.4, Stormwater and Urban Runoff Pollution Control from the Los Angeles Municipal Code, by requiring controls of pollutant discharges that utilize best available technology economically achievable and best conventional pollutant control technology to reduce pollutants.

Enforcement Agency:City of Los Angeles, Department of Public WorksMonitoring Agency:City of Los Angeles, Department of Public WorksMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of a grading permitAction Indicating Compliance with Mitigation Measure(s):Issuance of grading permit

MM-GEO-8 General contractors shall implement a fugitive dust control program pursuant to the provisions of SCAQMD Rule 403.

Enforcement Agency:South Coast Air Quality Management DistrictMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

11. Hazards

Project Design Features

PDF-HAZ-1 The transport, storage, use, and disposal of hazardous materials shall be overseen by the LMU Environmental Health and Safety Department in compliance with federal, state, and local regulations. In the unlikely event of a real or potential release, the LMU Environmental Health and Safety Department's emergency procedure for hazardous materials spills and releases shall be employed.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:OperationMonitoring Frequency:Periodic field inspection

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

Mitigation Measures

MM-HAZ-1 Prior to issuance of a demolition permit for a building, LMU shall submit verification to the City of Los Angeles Department of Building and Safety that an asbestos survey of the building has been conducted. If asbestos is found, such asbestos shall be removed prior to demolition in accordance with SCAQMD Rule 1403 and any other applicable regulations.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once for each building prior to issuance of a demolition permit Action Indicating Compliance with Mitigation Measure(s): Issuance of a demolition permit

MM-HAZ-2 Prior to issuance of a demolition permit for a building, LMU shall submit verification to the City of Los Angeles Department of Building and Safety that a lead-based paint survey of the building has been conducted. If lead based paint is found, LMU shall follow all procedural requirements and regulations for proper removal and disposal of the lead based paint prior to demolition.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once for each building prior to issuance of a demolition permit Action Indicating Compliance with Mitigation Measure(s): Issuance of a demolition permit

MM-HAZ-3 Fluorescent light ballast and other product labels shall be inspected prior to demolition. If the labels do not include the statement, "No PCBs," the product(s) shall be properly removed by a licensed PCB removal contractor and disposed of as PCB-containing waste prior to demolition.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once for each building prior to issuance of a demolition permit Action Indicating Compliance with Mitigation Measure(s): Issuance of a demolition permit MM-HAZ-4 All personnel potentially exposed to asbestos- or lead-containing materials shall be trained and protected in accordance with California Division of Occupational Safety and Health regulations.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once for each building prior to issuance of a demolition permit Action Indicating Compliance with Mitigation Measure(s): Report from Owner/Contractor documenting personnel training completion prior to issuance of a demolition permit

MM-HAZ-5 During subsurface excavation activities, including borings, trenching, and grading, California Division of Occupational Safety and Health worker safety measures shall be implemented as required to preclude any exposure to unsafe levels of soil gases, including but not limited to methane.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

MM-HAZ-6 Construction of new buildings and paved areas within the portions of campus located in a Methane Zone and Methane Buffer Zone as designated by the Los Angeles Department of Building and Safety shall comply with the City's Methane Seepage Regulations and the specifications of the Los Angeles Department of Building and Safety.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building during building plan checkAction Indicating Compliance with Mitigation Measure(s):Building plan check

12. Surface Water Hydrology and Water Quality

Project Design Features

PDF-HWQ-1 The runoff from 16 acres at the southwest corner of Burns Campus shall be diverted to the McConnell Avenue storm drain by rerouting these storm drains to the southeast.

Enforcement Agency:City of Los Angeles, Department of Public WorksMonitoring Agency:City of Los Angeles, Department of Public Works

City of Los Angeles ENV-2008-1342-EIR Monitoring Phase: Pre-construction

Monitoring Frequency: Once during building plan check for each new building to be located on the southwest corner of Burns Campus

Action Indicating Compliance with Mitigation Measure(s): Building plan check approval

PDF-HWQ-2 During Proposed Project construction of components on Burns Campus, the 24-inch LMU-owned storm drain leaving the Sunken Garden and the 12-inch LMU-owned storm drain immediately to the east of the Sunken Garden, and the 8-inch and 12-inch storm drains at the southwest corner of Burns Campus, shall be upsized to alleviate flooding in the Sunken Garden area and to accommodate additional runoff volume.

Enforcement Agency:City of Los Angeles, Department of Public WorksMonitoring Agency:City of Los Angeles, Department of Public WorksMonitoring Phase:Pre-Construction

Monitoring Frequency: Reviewed during building plan check for each new building to be connected to the upsized storm drains

Action Indicating Compliance with Mitigation Measure(s): Building plan check approval

Mitigation Measures

MM-HWQ-1 Prior to the start of soil-disturbing activities for individual projects on campus, a Notice of Intent (NOI) and Storm Water Pollution Prevention Plan shall be prepared in accordance with, and in order to partially fulfill, the California SWRCB Order No. 99-08-DWQ, National Pollutant Discharge Elimination System General Permit No. CAS000002 (General Construction Permit). The Storm Water Pollution Prevention Plan shall meet the applicable provisions of Sections 301 and 402 of the Clean Water Act and Chapter 6 Article 4.4, Storm Water and Urban Runoff Pollution Control from the City of Los Angeles Municipal Code by requiring controls of pollutant discharges that utilize best available technology economically achievable and best conventional pollutant control technology to reduce the rate and quantity of stormwater runoff. Examples of best available technology economically achievable and best conventional pollutant control technology that may be implemented during site grading and construction could include straw hay bales, straw bale inlet filters, filter barrier infiltration pits, stormwater cisterns, and silt fences.

Enforcement Agency:City of Los Angeles, Department of Public WorksMonitoring Agency:City of Los Angeles, Department of Public WorksMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of a grading permit

Action Indicating Compliance with Mitigation Measure(s): Issuance of a grading permit

MM-HWQ-2 LMU shall prepare and implement for individual projects on campus a Standard Urban Storm Water Mitigation Plan in accordance with the requirements of Chapter 6 Article 4.4, Storm Water and Urban Runoff Pollution Control, from the City of Los Angeles Municipal Code, to ensure that stormwater runoff water quality is managed through implementation of appropriate and applicable Best Management Practices. Prior to issuance of any grading or building permits for individual projects on campus, the City of Los Angeles Department of Public Works must approve the Standard Urban Storm Water Mitigation Plan.

Enforcement Agency:City of Los Angeles, Department of Public WorksMonitoring Agency:City of Los Angeles, Department of Public WorksMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of a grading permitAction Indicating Compliance with Mitigation Measure(s):Issuance of a grading permit

MM-HWQ-3 During the construction of individual projects, project contractors shall properly store hazardous materials to prevent contact with precipitation or runoff.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionsAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

MM-HWQ-4 During Proposed Project construction and subsequent operation, project contractors and LMU, respectively, shall develop and maintain effective monitoring and cleanup programs for spills and leaks of hazardous materials.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Construction and OperationMonitoring Frequency:Periodic field inspectionAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

MM-HWQ-5 During Proposed Project construction and subsequent operation, project contractors and LMU, respectively, shall place equipment to be repaired or maintained in covered areas on a pad of absorbent material to contain leaks, spills, or small discharge.

Enforcement Agency: City of Los Angeles, Department of Building and Safety

City of Los Angeles ENV-2008-1342-EIR MMP-29

Monitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Construction and OperationMonitoring Frequency:Periodic field inspectionAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

MM-HWQ-6 During Proposed Project construction and subsequent operation, project contractors and LMU, respectively, shall provide periodic and consistent removal of landscape and construction debris.

Enforcement Agency:City of Los Angeles, Department of Public Works (Bureau of
Sanitation)Monitoring Agency:City of Los Angeles, Department of Public Works (Bureau of
Sanitation)Monitoring Phase:Construction and OperationMonitoring Frequency:Periodic field inspectionAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

MM-HWQ-7 During Proposed Project construction and subsequent operation, project contractors and LMU, respectively, shall sweep parking lots at regular, frequent intervals to remove debris and shall also remove any significant chemical residue on the Project site through appropriate methods.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Construction and OperationMonitoring Frequency:Periodic field inspectionAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

MM-HWQ-8 LMU shall prepare and implement for individual projects on campus a Wet Weather Erosion Control Plan during between October 1 and April 15 in accordance with the requirements of Section 7002 of the Los Angeles Building Code.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-Construction, ConstructionMonitoring Frequency:Once prior to issuance of grading permit, Periodic field
inspection

Action Indicating Compliance with Mitigation Measure(s): Issuance of grading permit, Field inspection sign-off

13. Land Use and Planning

Project Design Features

PDF-LU-1	The Proposed Project	shall	comply	with	the	land	use	regulations	established	by	the
	proposed LMU Specifi	c Plan	•								

Enforcement Agency: City of Los Angeles, Department of City Planning
 Monitoring Agency: City of Los Angeles, Department of City Planning and Department of Building and Safety
 Monitoring Phase: Pre-construction, Operation
 Monitoring Frequency: Once during Substantial Compliance Review for new buildings; periodic during operation
 Action Indicating Compliance with Mitigation Measure(s): Administrative Clearance

approval; field inspection sign-off

Mitigation Measures

No mitigation measures required.

14. Noise

Project Design Features

PDF-NOISE-1 During construction activities, the operation of vibratory rollers and sonic pile drivers shall occur at a minimum distance of 50 feet from the campus boundary, and shall occur at a minimum distance of 84 feet from Xavier Hall, St. Robert's Hall, and Sacred Heart Chapel.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

PDF-NOISE-2 LMU shall provide construction worker parking at an off-site location in the campus vicinity. A shuttle service shall transport workers to and from campus in the morning and afternoon. No worker parking shall be permitted on residential streets.

Enforcement Agency:City of Los Angeles, Department of TransportationMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Periodic field inspectionAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

PDF-NOISE-3 All speakers for modified and new outdoor audio systems shall be mounted to face spectator areas, be directed away from adjacent residences, and be set to provide that sound levels from the systems do not exceed the off-campus ambient noise levels listed in Exhibit 7 of the LMU Specific Plan by 5 dB(A).

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-Construction, OperationMonitoring Frequency:Once prior to issuance of building (electrical) permit for new
equipment, Periodic field inspectionAction IndicatingCompliance with Mitigation Measure(s):Issuance of building
(electrical) permit for new equipment; field inspection sign-off

PDF-NOISE-4 New parking structures shall include a half-wall on the grade-level parking deck and/or full walls on the sides of the parking structure that face nearby residential receptors.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-Construction

Monitoring Frequency: Once for each parking structures during Administrative Clearance

Action Indicating Compliance with Mitigation Measure(s): Administrative Clearance approval

PDF-NOISE-5 LMU's Department of Public Safety shall continue to respond to on-campus incidents regarding excessive noise and student violations shall be sanctioned as provided in LMU's Student Conduct Code.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Operation

Monitoring Frequency: Periodically during operation

Action Indicating Compliance with Mitigation Measure(s): Compliance certification report by Applicant

PDF-NOISE-6 No source of outdoor amplified sound shall be installed or maintained on the LMU Campus within 150 feet of residential areas in the R1 zone zone. Outdoor amplified sound shall be prohibited in the Athletic Planning Area in the northeast corner of Campus. Emergency address systems shall be exempted from this requirement.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City Planning and
Department of Building and SafetyMonitoring Phase:Pre-construction, Operation

Monitoring Frequency: Once prior to issuance of building (electrical) permit for new equipment; periodically during operation

Action Indicating Compliance with Mitigation Measure(s): Issuance of building (electrical) permit for new equipment; field inspection sign-off

Mitigation Measures

- MM-NOISE-1 All construction activity shall be conducted in accordance with Section 112.05 of the Los Angeles Municipal Code Noise Ordinance, which states that all technically feasible measures shall be implemented to reduce noise levels of construction equipment operating within 500 feet of residential areas in cases where noise levels exceed 75 dB(A) at 50 feet from the noise source. The Project applicant shall therefore require in contract specifications that the following construction best management practices (BMPs) be implemented by contractors to reduce construction noise levels:
 - Two weeks prior to the commencement of construction of new buildings or any
 project requiring an Administrative Clearance under the Specific Plan, notification
 must be provided to surrounding land uses within 500 feet of a Project site disclosing
 the construction schedule, including the various types of activities that would be
 occurring throughout the duration of the construction period;
 - Ensure that construction equipment is properly muffled according to industry standards and in good working condition;
 - Place noise-generating construction equipment and locate construction staging areas away from sensitive uses;
 - Schedule high noise- and vibration-producing activities between the hours of 8:00 AM and 5:00 PM to minimize disruption to sensitive uses;
 - Implement noise attenuation measures, which may include, but are not limited to, temporary noise barriers such as curtains around construction areas or noise blankets around stationary construction noise sources in order to limit construction noise generation from exceeding existing ambient exterior noise levels by 5 dB(A) at a noise sensitive use;
 - Use electric air compressors and similar power tools rather than diesel equipment, for construction equipment that is available and economically feasible;
 - Construction-related equipment, including heavy-duty equipment, motor vehicles, and portable equipment, shall be turned off when not in use for more than 20 minutes; and
 - Construction hours, allowable workdays, and the phone number of the job superintendent shall be posted clearly at all construction entrances to allow for surrounding owners and residents to contact the job superintendent. If the City of
Los Angeles or the construction relations liaison receives a complaint, the liaison shall investigate, take appropriate corrective action, and report the action taken to the reporting party. Contract specifications shall be included in the Project construction documents, which shall be reviewed by City of Los Angeles representatives prior to issuance of a grading permit.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-Construction, ConstructionMonitoring Frequency:Prior to issuance of building permit; periodic field-inspectionAction Indicating Compliance with Mitigation Measure(s):Submittal of constructioncontracts with BMPs included; field inspection sign-off

MM-NOISE-2 The Project applicant shall require by contract specifications that heavily loaded trucks used during construction be routed away from residential streets. Contract specifications shall be included in the Project construction documents, which shall be reviewed by the City of Los Angeles representatives prior to issuance of a grading permit.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once prior to issuance of a grading permit

Action Indicating Compliance with Mitigation Measure(s): Issuance of a grading permit

MM-NOISE-3 The use of existing or improved audio systems associated with Sullivan Field, Page Stadium, Smith Field, and Burns Recreation Center Pool shall be limited to only those hours during which the facilities are being utilized, which shall not surpass 10:00 PM, except in the case of overtime or extra innings.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SaftyMonitoring Phase:OperationMonitoring Frequency:Periodically during operationAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

15. Police Protection

Project Design Features

PDF-POL-1 The LMU Department of Public Safety shall continue to provide first-response police protection services for the LMU campus and implement such security measures as maintaining a security booth located off Lincoln Boulevard, operating a network of security cameras on the campus, and securing all student resident halls by authorized key card entry.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Operation

Monitoring Frequency: Ongoing throughout operation

Action Indicating Compliance with Mitigation Measure(s): Compliance certification report by Applicant

Mitigation Measures

MM-POL-1 During Project construction, construction sites shall be secured with fencing and locked entrances. Construction equipment, tools and materials shall be secured by locking or placing them within sheds and/or other inaccessible areas while not in use. Construction sites shall be lighted as necessary at night to deter theft and vandalism.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Construction

Monitoring Frequency: Ongoing throughout construction

Action Indicating Compliance with Mitigation Measure(s): Field inspection sign-off

MM-POL-2 Prior to the issuance of a building permit, the Los Angeles Police Department, Pacific Division, Crime Prevention Unit, shall have the opportunity to comment regarding security and crime prevention features.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of building permitAction Indicating Compliance with Mitigation Measure(s):Los Angeles PoliceDepartment correspondence to Department of City Planning
documenting review.

MM-POL-3 Upon completion of each building or facility, a diagram of each completed building or facility shall be provided to the Los Angeles Police Department Pacific Division Commanding Officer. The diagram shall include access routes, building locations, and any additional information that would facilitate police response.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Police DepartmentMonitoring Phase:Post-constructionMonitoring Frequency:Once at completion of building/facility construction

MMP-35

Action Indicating Compliance with Mitigation Measure(s): Submittal of building or facility diagram to Police Department.

16. Fire Protection and Emergency Medical Services

Project Design Features

PDF-FIRE-1 LMU's Department of Public Safety shall continue to act as a first responder in emergencies and implement LMU's emergency procedures.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:OperationMonitoring Frequency: AnnuallyAction Indicating Compliance with Mitigation Measure(s):Compliance certificationreport by Applicant

Mitigation Measures

MM-FIRE-1 Prior to the issuance of any building permit, a plot plan shall be submitted to the Fire Department for approval.

Enforcement Agency:City of Los Angeles, Fire DepartmentMonitoring Agency:City of Los Angeles, Department Building and SafetyMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of building permitAction Indicating Compliance with Mitigation Measure(s):Fire Department approval

MM-FIRE-2 Prior to the issuance of any building permit, definitive plot plan and specifications including fire prevention features for the Project shall be submitted to and approved by the Fire Department.

Enforcement Agency:City of Los Angeles, Fire DepartmentMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of building permitAction Indicating Compliance with Mitigation Measure(s):Fire Department approval

MM-FIRE-3 Adequate off-site public and on-site private fire hydrants shall be required. The exact number and location of the hydrants shall be determined after the Fire Department reviews the plot plan. LMU shall be required to pay for any hydrant installations required by the Fire Department.

Enforcement Agency: City of Los Angeles, Fire Department

MMP-36

Monitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once for each building prior to issuance of building permit Action Indicating Compliance with Mitigation Measure(s): Fire Department approval

MM-FIRE-4 Adequate vehicular access ways around all multi-story buildings shall be required by the Fire Department where buildings exceed two stories in height.

Enforcement Agency:City of Los Angeles, Fire DepartmentMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of building permit

Action Indicating Compliance with Mitigation Measure(s): Fire Department approval

MM-FIRE-5 Where fire apparatus will be driven onto the road level surface of a subterranean parking structure, the structural foundation of the subterranean parking structures shall be engineered to withstand a bearing pressure of 8,600 pounds per square foot.

Enforcement Agency:City of Los Angeles, Fire DepartmentMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-Construction

Monitoring Frequency: Once for each building prior to issuance of building permit Action Indicating Compliance with Mitigation Measure(s): Fire Department approval

MM-FIRE-6 LMU shall covenant that all streets on campus shall be open to free travel of emergency vehicles.

Enforcement Agency:City of Los Angeles, Department of TransportationMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-constructionMonitoring Frequency:Once prior to issuance of building permit for first building
developed pursuant to LMU Specific Plan

Action Indicating Compliance with Mitigation Measure(s): Applicant submittal of recorded covenant

MM-FIRE-7 LMU shall work with the Los Angeles Department of Water and Power to construct or otherwise suitably guarantee the construction of water system improvements as needed to meet the on-site fire flow requirements set forth by LAFD. In the event off-site improvements are necessary, LMU shall contribute its fair share to water system improvements to meet the on-site fire flow requirements set forth by the Fire Department.

Enforcement Agency: Los Angeles Department of Water and Power; City of Los Angeles, Fire Department

Monitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once after submittal of plans for water system improvements to

meet the on-site fire flow improvements for each building prior to issuance of building permit

Action Indicating Compliance with Mitigation Measure(s): Department of Water and Power and Fire Department approval

17. Recreation and Parks

Project Design Features

PDF-REC-1 The LMU Campus shall provide, at a minimum, the acreage of open space and outdoor athletic facilities required by the proposed LMU Specific Plan.

Enforcement Agency:City of Los Angeles, Department of City PlanningMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:Pre-ConstructionMonitoring Frequency:Once for each building during Administrative ClearanceAction Indicating Compliance with Mitigation Measure(s):Administrative Clearanceapproval

Mitigation Measures

No mitigation measures required.

18. Transportation

Project Design Features

No Project Design Features are proposed.

Mitigation Measures

Construction Mitigation Measures

- MM-TRAF-1 Prior to the issuance of demolition permits, a Construction Traffic Management Plan shall be prepared and submitted to the City of Los Angeles Department of Transportation for review and approval. The Construction Traffic Management Plan shall include information such as haul routes and staging plans. The Construction Traffic Management Plan shall include the following elements:
 - Provisions to configure construction parking to minimize traffic interference and avoid parking on residential streets.

- Consolidating and coordinating haul trucks, deliveries, and pick-ups to reduce the potential for trucks waiting to load or unload for protracted periods of time.
- Construction equipment traffic from the contractors shall be controlled by flagman in order to minimize circulation conflicts and obstruction of through-traffic lanes.
- Maintaining access to residences and businesses.
- Provision of safety precautions for pedestrians and bicyclists through such measures as alternate routing, and protection barriers.

Enforcement Agency:City of Los Angeles, Department of TransportationMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-constructionMonitoring Frequency:Once for each building prior to issuance of demolition permits

Action Indicating Compliance with Mitigation Measure(s): Approval by Department of Transportation; issuance of demolition permit

MM-TRAF-2 Construction worker parking shall be provided at an off-site location in the campus vicinity, where sufficient parking for the expected number of workers can be accommodated. A shuttle service shall be provided to transport construction workers to and from campus in the morning and afternoon. No construction worker parking shall be permitted on residential streets.

Enforcement Agency:City of Los Angeles, Department of TransportationMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:ConstructionMonitoring Frequency:Ongoing throughout constructionAction Indicating Compliance with Mitigation Measure(s):Field inspection sign-off

- MM-TRAF-3 The Proposed Project applicant shall implement at least one of the following Transportation Demand Management (TDM) strategies to mitigate Proposed Projectrelated intersection impacts during Project buildout.
 - Increase number of residential beds to 4,250 from the 3,261 beds currently available (an increase of 989 beds). This mitigation measure would allow the campus to grow to 9,545 FTE students, faculty, and staff before a significant traffic-related significant impact occurs.
 - Implement a 5 percent faculty/staff TDM program as discussed in the *Transportation Impacts Analysis* prepared in December 2009 for the Proposed Project and increase the number of residential beds to at least 4,197 from the 3,261 beds currently available (an increase of 936 beds). This mitigation measure would allow the campus to grow to the full 9,600 FTE students, faculty, and staff buildout without significant trafficrelated impacts.

• Implement a 10 percent faculty/staff TDM program as discussed in the *Transportation Impacts Analysis* prepared in December 2009 for the Proposed Project and increase the number of residential beds to at least 4,032 from the 3,261 beds currently available (an increase of 771 beds). This mitigation measure would allow the campus to grow to the full 9,600 FTE students, faculty, and staff buildout without significant trafficrelated impacts.

Enforcement Agency:City of Los Angeles, Department of TransportationMonitoring Agency:City of Los Angeles, Department of City PlanningMonitoring Phase:OperationMonitoring Frequency:Ongoing until Project buildout

Action Indicating Compliance with Mitigation Measure(s): Compliance certification report by Applicant approved by Department of Transportation

MM-TRAF-4 The Transportation Demand Management (TDM) program implementation and monitoring shall be conducted in accordance with the guidelines set forth in the *Transportation Impact Analysis* prepared in December 2009 for the Proposed Project. The final, detailed TDM plan shall be presented to the Los Angeles Department of Transportation for approval.

Enforcement Agency:City of Los Angeles, Department of TransportationMonitoring Agency:City of Los Angeles, Department of TransportationMonitoring Phase:Operation

Monitoring Frequency: Ongoing until Project buildout

Action Indicating Compliance with Mitigation Measure(s): Submittal of a final TDM Plan to the Department of Transportation

MM-TRAF-5 The Proposed Project shall add parking in phases with increases in the campus full-time equivalent population, in conformance with the requirements of the proposed Specific Plan. The location of such additional parking shall be approved by the Department of Transportation.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:OperationMonitoring Frequency:Annually until all 4,742 parking spaces required are provided

on campus

Action Indicating Compliance with Mitigation Measure(s): Annual submittal of reports to the Department of Building and Safety, due by October 1 of each year, that certifies the number of full-time equivalent students, faculty, and staff enrolled or working on Campus for the current academic year, and that includes parking plans showing that the Campus provides the number of parking spaces required by the Specific Plan

MM-TRAF-6

Prior to pulling building permits for the new sports pavilion (i.e., the replacement facility for Gerston Pavilion), or the construction of more than 1000 additional seats (individually or cumulatively) at Page Stadium, Smith Softball Field, and/or Sullivan Field, the Applicant shall obtain approval from the Department of Transporation of a parking plan for the new sports pavilion or the new sports facility seating at Page Stadium, Smith Softball Field, and/or Sullivan Field demonstrating sufficient parking availability for such new sports pavilion or new sports facility seating at Page Stadium, Smith Softball Field, and/or Sullivan Field. Parking for the new sports pavilion or new sports facility seating at Page Stadium, Smith Softball Field, and/or Sullivan Field may be met by demonstrating sufficent capacity through a shared use study of the then-existing parking demand, construction of new parking spaces, or a special event parking plan which may include valet/stacked parking and/or shuttle services from University Hall to other locations on campus, or shuttle services to and from off-site parking locations.

Enforcement Agency:City of Los Angeles, Department of TransportationMonitoring Agency:City of Los Angeles, Department of TransportationMonitoring Phase:Pre-constructionMonitoring Frequency:Once during building plan check for new sports pavilion or the
construction of more than 1000 additional seats (individually or
cumulatively) at Page Stadium, Smith Softball Field, and/or
Sullivan FieldAction IndicatingCompliance with Mitigation Measure(s):Department of
Transportation review and determination on additional parking

space requirements, if any; building plan check approval

MM-TRAF-7

LMU will maintain an Event Parking Management Program to accommodate occasional university functions expected to bring a large number of non-registered vehicles onto campus. The Event Parking Management Program will provide for a temporary increase in traffic management and parking personnel to accommodate the additional vehicles on the campus. LMU shall direct visitors leaving events by car to exit the campus via LMU Drive.

Enforcement Agency:City of Los Angeles, Department of TransportationMonitoring Agency:City of Los Angeles, Department of TransportationMonitoring Phase:Operation

Monitoring Frequency: Annually

Action Indicating Compliance with Mitigation Measure(s): Compliance certification report by Applicant

19. Water Supply

Project Design Features

- PDF-WATER-1 The Proposed Project would include the following water conservation features in new development approved as part of the Proposed Project through Proposed Project buildout, unless alternative or equivalent measures are substituted with City approval. Proposed Project buildout means the addition of 508,000 net new gross square feet of academic/administrative facilities, 476,000 net new gross square feet of residential facilities, and 28,000 net new gross square feet of athletic indoor facilities on campus:
 - Bathroom faucets 1.5 gallons per minute (private), 0.5 gallon per minute (public)
 - Self-closing faucets in public restrooms
 - Kitchen faucets 1.5 gallons per minute
 - Pre-rinse kitchen spray head
 - Showerheads: no more than 1 showerhead per stall
 - Low-flow showerheads 2.0 gallons per minute
 - High efficiency clothes washers water savings factor of 5.0 or less (residential); water savings factor of 7.5 or less (residential)
 - High efficiency toilets 1.28 gallons per flush or less, or dual flush
 - High efficiency/ultra low flow urinals 0.125 to 0.5 gallon per flush
 - Energy Star dishwashers
 - Domestic water heating system located in close proximity to point(s) of use
 - Tankless and on-demand water heaters
 - Cooling tower conductivity controllers or cooling tower pH conductivity controllers
 - (Cooling towers to operate at minimum of 5.5 cycles of concentration)
 - Water-saving pool filter
 - Rotating sprinkler nozzles 0.5 gallon per minute
 - Micro-spray nozzles
 - Drip/subsurface irrigation (micro-irrigation)and bubbler irrigation

- Weather based irrigation controller
- Hydro-zoning plantings (grouping similar water needs plants together)
- Zoned irrigation
- Drought-tolerant plants: 75 percent of new landscape plantings
- Artificial turf (cost permitting)
- Landscaping contouring to minimize precipitation runoff
- Infiltration planters (i.e., notched curb to allow runoff to flow into planted areas)
- Stormwater capture and infiltration of on campus sump
- Campus-wide reclaimed water irrigation (by Project buildout)
- Cooling towers using 100 percent reclaimed water use, as permitted by law (by Project buildout)
- New buildings designed to meet the U.S. Green Building Council's Leadership in Energy and Environmental Design[®] (LEED[®]) Certified level (or higher), or an equivalent criteria.

Enforcement Agency: City of Los Angeles, Department of Building and Safety
Monitoring Agency: City of Los Angeles, Department of Building and Safety
Monitoring Phase: Pre-Construction, Operation
Monitoring Frequency: Periodically during plan check for each building and operation as needed
Action Indicating Compliance with Mitigation Measure(s):Issuance of building permit;

field inspection sign-off

Mitigation Measures

No mitigation measures required.

20. Wastewater

Project Design Features

See Section IV L 1, Water Supply.

Mitigation Measures

MM-WW-1 If future capacity studies and calculations during the course of Project improvements determine that any sewer pipe sections on campus would operate at flow depths greater

MMP-43

than 75 percent of pipe depth, such pipes shall be upsized as necessary at the expense of LMU.

Enforcement Agency: City of Los Angeles, Department of Public Works (Bureau of Sanitation)

Monitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-Construction

Monitoring Frequency: As needed during plan check for new buildings

Action Indicating Compliance with Mitigation Measure(s): Sign-off by Department of Public Works (Bureau of Sanitation) on future capacity studies

21. Solid Waste

Project Design Features

PDF-SW-1 LMU shall continue to achieve a campuswide waste diversion rate of at least 58.6 percent through recycling activities.

Enforcement Agency: City of Los Angeles, Department of Public Works (Bureau of Sanitation)

Monitoring Agency: City of Los Angeles, Department of City Planning

Monitoring Phase: Operation

Monitoring Frequency: Annually until Project buildout

Action Indicating Compliance with Mitigation Measure(s): Submittal of compliance report by Applicant to the Department of Public Works

Mitigation Measures

No mitigation measures required

22. Energy (Electricity and Natural Gas)

Project Design Features

PDF-ENG-1 LMU shall implement the following energy conservation measures as part of the Proposed Project:

- Central Plant motors shall include variable frequency drivers to adjust electrical motor speed based on demand;
- Major building renovations and additions shall be integrated into the Campus Energy Management System, which is a set of computer-aided tools used to monitor, control, and optimize the performance of building HVAC and lighting systems;

- Future cooling loads shall be met using thermal energy storage, or an additional energy efficient chiller, or other comparable storage technologies;
- New and replacement buildings with flat roofs shall use white reflective material or comparable heat rejecting material on the building roofs;
- New appliances shall meet or exceed the minimum efficiency levels mandated in the California Code of Regulations;
- All irrigation shall use reclaimed water by Project buildout;
- All irrigation shall use automatic irrigation timers and at least 50 percent of the campus's non-turf areas shall include drought-tolerant or native plantings;
- All new and renovated buildings shall incorporate water conservation measures such as ultra-low-flush water closets and urinals, low-flow shower heads, and low-flow faucet aerators;
- All new construction shall be designed to the 2008 LEED Certified criteria (or an equivalent criteria) or better;
- Buildings shall be well sealed to prevent outside air from infiltrating and increasing interior space-conditioning loads;
- Buildings shall incorporate thermal insulation in walls and ceilings;
- Window systems shall be designed to reduce thermal gain and loss, thus, reducing cooling loads during warm weather and heating loads during cool weather; and
- High-intensity-discharge (HID) lamps, light-emitting diode (LED), or other energy efficient lighting shall be installed for all outdoor lighting to reduce electricity consumption.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Construction, OperationMonitoring Frequency:Periodically during plan check for each building and operation
as needed

Action Indicating Compliance with Mitigation Measure(s): Issuance of building permit; field inspection sign-off

Mitigation Measures

Electricity

MM-ENG-1 Prior to submittal of final site plans associated with specific Project facilities for approval by the City's Building and Safety Department, LMU shall consult with LADWP to determine the appropriate specifications for additional transmission or distribution facilities supplying electricity to the Project site. Upon finalization of these specifications, LMU shall fund its fair share of the cost of on campus or off campus infrastructure installation, as applicable.

Enforcement Agency:Los Angeles Department of Water and Power; City of Los
Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and Safety

Monitoring Phase: Pre-construction

Monitoring Frequency: Once prior to submittal to Department of Building and Safety of final site plans for each Project facility

- Action Indicating Compliance with Mitigation Measure(s): Documentation of consultation with LADWP; payment of fair share of costs, if applicable
- MM-ENG-2 Prior to submittal of final site plans for specific Project buildings or facilities to the City's Building and Safety Department demonstrating compliance with the State's Energy Conservation Standards, LMU shall consult with LADWP's Energy Solutions Group regarding the incorporation of possible energy efficiency measures into Project design.

Enforcement Agency:Los Angeles Department of Water and Power; City of Los
Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-constructionMonitoring Frequency:Once prior to submittal of final site plans for each Project facility
to Department of Building and Safety

Action Indicating Compliance with Mitigation Measure(s): Documentation of consultation with LADWP's Energy Solutions Group

Natural Gas

MM-ENG-3 Prior to submittal of final site plans for specific Project buildings or facilities to the City's Building and Safety Department demonstrating compliance with the State's Energy Conservation Standards, LMU shall incorporate the appropriate specifications for necessary modifications to the natural gas conveyance system to the Project site as required by the Gas Company. Upon finalizing these specifications, LMU shall fund its fair share of the cost of on campus or off campus infrastructure installation, as applicable.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once prior to submittal of final site plans for specific Project buildings or facilities to Department of Building and Safety

Action Indicating Compliance with Mitigation Measure(s): Documentation of consultation with the Southern California Gas Company; payment of fair share costs, if applicable

MM-ENG-4

Prior to submittal of final site plans for specific Project buildings or facilities to the City's Building and Safety Department demonstrating compliance with the State's Energy Conservation Standards, LMU shall consult with the Gas Company regarding the incorporation of feasible energy conservation measures into Project design.

Enforcement Agency:City of Los Angeles, Department of Building and SafetyMonitoring Agency:City of Los Angeles, Department of Building and SafetyMonitoring Phase:Pre-construction

Monitoring Frequency: Once prior to submittal of final site plans for specific Project buildings or facilities to Department of Building and Safety

Action Indicating Compliance with Mitigation Measure(s): Documentation of consultation with the Southern California Gas Company

Exhibit C:

Project Description

The Project, also called the Loyola Marymount University Proposed Master Plan Project, refers to the proposed plans to improve the academic, administrative, residential, athletic, and recreational facilities on LMU's Westchester campus, while maintaining the existing approved enrollment cap of 7,800 full-time enrolled students. Of the campus's existing approximately 1,651,000 square feet of academic and administrative facilities, LMU would remove approximately 515,000 square feet of obsolete facilities, and build approximately 1,023,000 square feet of new facilities, for a net increase of approximately 508,000 square feet of academic and administrative facilities. Of the campus's existing approximately 942,000 square feet of existing residential uses, LMU would remove approximately 370,000 square feet, and add approximately 846,000 square feet, for a net increase of approximately 476,000 square feet. Of the campus's existing approximately 185,000 square feet of existing indoor athletic uses, LMU would remove approximately 80,000 square feet, and add approximately 108,000 square feet, for a net increase of approximately 28,000 square feet. LMU would also add approximately 4.8 acres of net new athletic outdoor facilities, and add approximately 5 aces of net new landscaped open space. LMU would also add parking on campus, in surface, structure, or underground parking areas, in phases with the addition of student enrollment and building construction. A total of approximately 4,742 parking spaces would be provided at full buildout.

Exhibit D:

Project Approvals

- 1. Certification of Final Environmental Impact Report, State Clearinghouse No. 2008051103.
- 2. Establishment of the Loyola Marymount University Specific Plan through approval of Ordinance No. <u>181605</u>.
- 3. General Plan Amendment changing the Westchester-Playa del Rey Community Plan designation of the Property from Low Residential Density to High Medium Residential Density. Council File No. <u>10-1809</u>
- 4. Zone Change from [Q]R4-1 to R4-1 through approval of Ordinance No. 181603.

Exhibit E:

Legal Description of Property

Attached.



CIVIL ENGINEERING SURVEYING+MAPPING LAND DEVELOPMENT

EXHIBIT "A"

LOYOLA MARYMOUNT CAMPUS

LEGAL DESCRIPTION

Parcel 1

Parcel A of Parcel Map L.A. No. 5409, in the City and County of Los Angeles, State of California, as per map filed in Book 171 Pages 71 through 76 inclusive, of Parcel Maps, in the Office of the County Recorder of said County.

Except therefrom that portion of said Parcel A described as follows:

Beginning at the most southerly corner of said Parcel A; thence, along the generally southerly boundary of said land North 55°04'49" East 94.96 feet; thence, North 34°55'55" West 215.30 feet to an angle point in said boundary and the True Point of Beginning; thence,

1st - Continuing along said boundary North 62°28'06" East 121.27 feet to a non-tangent curve concave to the east having a radius of 1235.00 feet through which a radial line bears South 62°28'06" West; thence,

2nd – Northerly along said curve through a central angle of 0°27'50" an arc length of 10.00 feet; thence,

3rd- South 62°28'06" West 122.61 feet to a line which bears North 34°55'55" West and passes through the True Point of Beginning; thence,

4th – Along said line South 34°55'55" East 10.08 feet to the True Point of Beginning.

Parcel 2

Parcel B of Parcel Map L.A. No. 5409, in the City and County of Los Angeles, State of California, as per map filed in Book 171 Pages 71 through 76 inclusive, of Parcel Maps, in the Office of the County Recorder of said County.

Parcel 3

That portion of Lot 38 of Section 26, Township 2 South, Range 15 West, San Bernardino Meridian in the Rancho Sausal Redondo as shown on map filed in Book 1 Pages 507 and 508 of Patents, bounded as follows:

Bounded on the east by the easterly line of said Section 26, said easterly line also being the westerly lines of Tract No. 15167, recorded in Book 331 Pages 3 through 5 inclusive, of Maps and Tract No. 14759, recorded in Book 315 Pages 32 through 34 inclusive, of Maps, both in the Office of the County Recorder of said County.

Bounded on the south by the north line of 80th Street, 73 feet wide, as described in Resolution recorded in Book D-2878 Page 572, Official Records of said County.

Bounded on the west by the east line of Fordham Road, 60 feet wide, as shown on Tract No. 9430, recorded in Book 151 Pages 67 through 69 inclusive, of Maps in the Office of the County Recorder of said County and the northerly prolongation of said east line as shown on Parcel Map L.A. No. 5409, filed in Book 171 Pages 71 through 76 inclusive, of Parcel Maps, in the Office of the County Recorder of said County.

Except therefrom that portion of land lying within the boundaries of the land described in Resolution recorded in Book D-4068 Page 933, Official Records of said County.

Bounded on the north by the southerly line of a 30 foot wide easement for sewer purposes described in deed to the City of Los Angeles recorded May 14, 1924 as Instrument No. 1275 in Book 3080 Page 289, Official Records of said County.



EXHIBIT "A" – CONT. LOYOLA MARYMOUNT CAMPUS LEGAL DESCRIPTION

Parcel 4

Being a portion of the north half of Section 26, Township 2 South, Range 15 West, San Bernardino Meridian in the Rancho Sausal Redondo as shown on map filed in Book 1 Pages 507 and 508 of Patents and being a portion of Fordham Road, 60.00 feet wide, as shown on Tract No. 9430 recorded in Book 151 Pages 67 through 69 inclusive, of Maps, described as follows:

Beginning at a point on the north line of 78th Street, 60.00 feet wide, as shown on Parcel Map L.A. No. 5409, in the City and County of Los Angeles, State of California, as per map filed in Book 171 Pages 71 through 76 inclusive, of Parcel Maps, said point being the east terminus of a line that bears North 89°52'00" East 30.61 feet; thence,

- 1st- North 89°52'00" East 12.99 feet along the east produced line to the beginning of a tangent curve concave to the southwest having a radius of 57.00 feet; thence,
- 2nd Easterly and southerly 89.55 feet along said curve through a central angle of 90°00'40" to a point on the east line of Fordham Road, 60.00 feet wide, as shown on said Tract 9430; thence,
- 3rd North 00°07'20" West 1589.29 feet along said east line of Fordham Road, 60.00 feet wide, as described in Book 14039 Page 96, Official Records of said County, to the southerly line of a 30 foot wide easement for sewer purposes described in deed to the City of Los Angeles recorded May 14, 1924 as Instrument No. 1275 in Book 3080 Page 289, Official Records of said County. Said line being a curve concave to the South having a radius of 185.00 feet through which a radial line bears North 18°30'31" West; thence,
- 4th Southwesterly 69.06 feet along said curve through a central angle of 21°23'18" to the west line of said Fordham Road, said line also being the east line of Parcel C of said Parcel Map L.A. No. 5409; thence,
- 5th South 00°07'20" East 1488.92 feet along said west line of Fordham Road to the beginning of a tangent curve concave to the northwest having a radius of 10.00 feet; thence,
- 6th- Southerly and westerly 15.71 feet along said curve through a central angle of 89 59'20" to the Point of Beginning.

The attached exhibit is hereby made a part for reference purposes only.

William A. Nailling, PLS No. 8476





Exhibit F:

Meeting Room Costs Associated With Section 3.1.3.3

This list includes the typical, non-rental costs associated with use of Property Owner's facilities, comparable to the rates allocated to internal users for these facilities.

This is not an exclusive list and other costs may also apply. A schedule of costs shall be included in the annual report required as part of Section 4.2 of this Agreement, and rates may be adjusted periodically by Property Owner. Prior to the first annual report required as part of Section 4.2 of this Agreement, a list of rates shall be provided upon request to the organizer of the event.

Event Setup:

Normal university business hours are Monday –Friday, 8:00 am to 5:00 pm. If the setup for an event or the event itself occurs outside of normal business hours, Property Owner will charge to cover labor costs. If Property Owner is able to complete setups during the business day (Monday – Friday, 8:00 am to 5:00 pm) then there will be no charge for setup. Property Owner will only charge for the portion of setup that takes place outside of business hours. Property Owner shall make good faith efforts to complete setups during business hours when feasible, for instance when there are no intervening uses of a facility between the close of business hours and the time of the meeting which requires setup.

Audio-Visual (AV) Attendant:

If the event organizer requests an AV technician to be present or on-call for the event, there will be an hourly charge regardless of business hours.

Cleaning:

If the event requires or results in the need to schedule additional custodial or grounds staff, or carpet cleaning, there will be a charge. Events with guest counts of 100 or greater require a minimum of two custodial staff.

Insurance:

All organizations must provide a Certificate of Insurance naming Loyola Marymount University as additional insured on both Public and General Liability and Property Damage coverage. If the organization does not have insurance, Property Owner can assist with obtaining an insurance policy for the event.

Parking:

Events during normal business hours with a guest count of 50 or greater will require additional parking resources, with specified charges. This may include, but is not limited to, shuttles, valet parking personnel, traffic control personnel, or the rental of an off-site parking facility.

Exhibit G:

Permit Parking Area

